

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk Summary of thesis "Statistical Analysis of Congestion in Road Traffic at an Intersection", by D. H. Reid

The problem considered is that of delays caused to vehicles on a major two-lane road at its intersection with a minor road. There are no traffic lights or policemen at the intersection and delays are assumed to occur purely as a result of right-turning traffic (where left-hand-drive is the convention observed). A survey of literature relevant to this problem is made in Chapter 1.

In Chapters 2, 3, 4 mathematical models of increasing complexity are proposed as idealisations of the actual situation. These models are constructed with the aims of simplicity and realism in view, so that a valid mathematical analysis of the steady-state behaviour of each may be made. The models are based on such simplifications as the replacement of vehicles by geometrical points, the generation of vehicle arrivals by a Poisson process, and rules for the interaction of vehicles at the intersection which, although probably unrealistic in model I, may provide a reasonable picture of actual behaviour in model III. The distributions o delay to vehicles is determined under the assumptions of each of models I - III, and programmes were constructed to calculate numerical values for certain aspects of these distributions.

Chapter 5 is concerned with the assessment of these models, particular model III, against observations. The observations are described and their limitations discussed. The parameters defining model III are estimated, and the goodness-of-fit of model III to the data is considered. A comparison of model III with model I is attempted. The tentative conclus ProQuest Number: 10647040

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10647040

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

is reached that on the basis of the data available model III may provide an adequately realistic picture of vehicle behaviour.

In a final chapter some practical applications of the work are briefly considered, and further related problems are described. STATISTICAL ANALYSIS OF CONGESTION IN ROAD TRAFFIC AT AN INTERSECTION.

BY

DAVID H. REID

Thesis presented to the University of Glasgew for the degree of Ph.D., March, 1968. Acknowledgement.

I should like to express my grateful approciation of the assistance and encouragement given to me by Dr. A.J. Howie.

(1)

| | | Contents. | Page |
|----|--------|---|------|
| 1. | Introd | uction and Description of Problem. | 1 |
| | 1.1 | Congestion and mathematical models. | 1 |
| | 1.2 | Situations in road traffic which generate congestion. | 3 |
| | 1.2.1 | Delays to vehicles on the open road. | 4 |
| | 1.2.2 | Delays to vehicles at intersections. | 4 |
| | 1.2.3 | The main problem of the thesis. | 6 |
| | 1.3 | Previous relevant work. | 6 |
| | 1.3.1 | Previous discussions of delays to m-vehicles at a priority intersection. | 7 |
| | 1.3.2 | Previous discussions of delays to M-vehicles at a priority intersection. | 16 |
| | 1.3.3 | The relevance of these models to the present problem, | 19 |
| 2. | Constr | enction and Analysis of Model I. | 22 |
| | 2.1 | Derivation of the model. | 22 |
| | 2.2 | Description of model I. | 24 |
| | 2.3 | Steady-state behaviour of model I. | 26 |
| | 2.3.1 | Selection of regeneration points. | 26 |
| | 2.3.2 | Notation for steady-state analysis. | 28 |
| | 2.3.3 | Transition probabilities. | 30 |
| | 2.3.4 | Evaluation of state probabilities. | 32 |
| | 2.3.5 | The mean delay. | 39 |

| | 2.4 | Remarks on model I. | A2 |
|----------------|----------|---|----|
| | 2.4.1 | Validity of the initial assumptions. | 43 |
| | 2.4.2 | Predicted behaviour of model I. | 44 |
| | | | |
| 3 | Constr | uction and Analysis of Model II. | 46 |
| | 3.1 | Derivation of the model. | 46 |
| | 3.2 | Description of model II. | 49 |
| | 3.3 | Steady-state behaviour of model II. | 51 |
| | 3.3.1 | Selection of regeneration points: notation. | 51 |
| | 3.3.2 | Transition probabilities. | 53 |
| | 3.3.3 | Determination of state probabilities. | 55 |
| | 3.3.4 | The distribution of delay. | 59 |
| | 3.3.5 | Numerical results. | 63 |
| | 3.3.6 | Proof of result used in section 3.3.3. | 66 |
| | 3.4 | Romanks on model II. | 67 |
| A | Constr | uction and Analysis of Model TIT | 69 |
| and the second | COMAG GA | ACTION AND INTELISTO OF INGEL AND | 37 |
| | 4.1 | Derivation of the model. | 69 |
| | 4.2 | Description of model III. | 70 |
| | 4.3 | Steady-state behaviour of model III. | 71 |
| | 4.3.1 | Selection of regeneration points: notation. | 71 |
| | 4.3.2 | Transition probabilities. | 73 |
| | 4.3.3 | Determination of state probabilities. | 75 |
| | 4.3.4 | The distribution of delay. | 82 |

(11)

| | 4.3.5 | The probability that a vehicle in lane 1 is not delayed. | 83 |
|----|--------|---|------|
| | 4.3.6 | Numerical results. | 84 |
| | 4.4 | Remarks on model III | 89 |
| | 4.5 | 'Impatient' queueing type I vehicles: medel III(a) | . 90 |
| 5. | Statis | tical Analysis of Observations. | 92 |
| | 5.1 | Introduction. | 92 |
| | 5.2 | The observations. | 92 |
| | 5.3 | Estimation problems. | 93 |
| | 5.3.1 | Estimation of p_i, λ_i . | 93 |
| | 5.3.2 | Estimation of other parameters in model III. | 94 |
| | 5.3.3 | Improvements on the crude estimates of the gap-acceptance parameters. | 97 |
| | 5.4 | The assessment of goodness-of-fit. | 107 |
| | 5.4.1 | The distribution of arrivals. | 107 |
| | 5.4.2 | Realism of predictions from model III. | 109 |
| | 5.4.3 | A comparison of model III with model I. | 117 |
| | 5.4.4 | The correlation coefficient of successive delays to vehicles in lane 2 of model III. | 118 |
| 6, | Conclu | usion. | 127 |
| | 6.1 | Discussion of chapters 2-5: problems cutstanding and conclusions. | 127 |
| | 6.2 | Practical applications of model III. | 129 |
| | 6.3 | Further problems. | 1 30 |

(iii)

(1v)

- Appendix I : Transition probabilities.
- Appendix II : Practical difficulties.
- Appendix III : Observations.
- Appendix IV : References.

Chapter 1.

Introduction and Description of Problem.

1.1. Congestion and mathematical models.

Situations are often encountered in which some type of agency provides service for individuals or groups of a population, and in which the demand for service at times exceeds that which the agency is able to supply. Congestion, in a broad sense, refers to the non-availability of service to the customer at the instant at which it is first required. Examples are the demands by drivers of cars for entry to a parking space, or by telephone subscribers for the routeing of calls through an exchange. In each of these cases it is possible that if the service is not available as soon as it is required, the immediate demand disappears.

The kind of situation of interest here, by contrast, has the property that a customer whose demand for service is not immediately met will wait until the service becomes available. In many cases the customers wait in an orderly manner, forming a queue or queues, and receiving service in the order of their positions in the queue. For customers delayed in this way, the interval of waiting for service often represents an economic loss, and so the minimisation of the total cost of the system may be a problem of some importance. In service- or congestion- systems of this kind, it would appear that efficiency might be improved by the careful weighing of the cost of providing better service facilities against the savings accruing to customers, as a result of reduced delays. In order to assess various possible designs for a typical congestion-system a detailed analysis is required, which might well initially be concerned with an existing system.

The first steps in the analysis of the system, as of any other physical system, should be to observe and to describe it as precisely as possible. The extent to which this can be achieved will vary with different types of system. In some cases, for example, it may be possible to describe the service-mechanism in detail. In others it may be necessary to use inferential methods to discover the nature of the service from observations on the input and output processes. For many systems a purely deterministic description will not be satisfactory and the language and ideas of the theory of stochastic processes will be appropriate.

The description of the system is then used to construct a conceptual, usually mathematical, model of the real situation. In the first stages of the analysis the model is chosen to be as simple as possible while remaining consistent with the salient features of the description. At a later stage it may be possible to modify the original model, in the light of the analysis, to

correspond more closely with the real situation. The description of many models is in terms of certain parameters, to which values must be allocated at the discretion of the analyst, and in this way a considerable degree of flexibility may be built into a model of a particular type.

The inclusion of parameters in a model gives rise to problems of statistical inference. There are naturally problems of estimation, and an assessment of "goodness-of-fit" is desirable in order to decide whether the model really reflects the features of the situation which are of interest, or whether a further model should be considered.

Ideally a model may give guidance on aspects of a congestionsystem other than the purely economic questions. A satisfactory model may provide information on such problems as the general behaviour of the system, its efficiency and capacity, and the effects of minor modifications on the system.

1.2. Situations in road traffic which generate congestion.

It is apparent that there are many situations in road traffic which result in congestion. Vehicles may be delayed by other moving or stationary vehicles. Delays to pedestrians are usually caused by moving vehicles.

20

1.2.1. Delays to vehicles on the open road.

Travel on the open road is often impeded by the presence of slower-moving vehicles in the same carriageway. Mathematical models for this situation have been proposed by Miller (1962), Tanner (1961), and Newell (1955), among others, and have been examined to determine the listributions of delays caused by differences in speeds of vehicles. These authors give more attention to the development of the models than to the systematic assessment of their adequacy and realism.

1.2.2. Delays to vehicles at intersections.

Vehicles may be delayed at intersections by other vehicles, or by pedestrians, policemen, or traffic lights. Several authors have constructed mathematical models to describe the behaviour of vehicles at T-junctions and other types of intersection. Usually the effects of pedestrians on vehicle flow are ignored, although consideration has been given to the delay caused to pedestrians wishing to cross a road. (Tanner, 1953 and Mayne, 1954). Here pedestrians will be assumed to have no appreciable effect on the flow of traffic, and we shall be concerned with intersections which are not controlled by policemen or traffic-lights.

Such uncontrolled intersections are often of what we may call the 'priority' type, in which there are 'halt' or 'stop' signs requiring vehicles in certain lanes at the intersection to give

way to those in other lanes.

It is convenient to distinguish between the terms "T-junction" and "intersection". The former will be used to refer to the junction of a minor road with one side of a major road, the latter to the junction of two minor roads with the opposite sides of a major road at a given position along its length.

For brevity we shall sometimes refer to

'vehicles on the major road' as 'M-vehicles'.

'vehicles on the minor road' as 'm-vehicles'.

and for instance, 'the driver of a vehicle on the major read' as an 'M-driver'.

T-junction





1.2.3. The main problem of the thesis.

The central problem of this thesis is to investigate the delay caused to M-vehicles at a priority intersection by right-turning vehicles^{II} in the absence of controlling factors, such as traffic lights or policemen, and interfering factors, such as pedestrians. It is assumed that the major road has effectively a single lane in each direction, so that it is not possible for vehicles going straight on or turning left at the intersection to pass to the left of a vehicle awaiting an opportunity to complete its right turn.

The form of the thesis is as follows. In chapters 2, 3 and 4, mathematical models of increasing complexity are proposed for the problem. In chapter 5 the attempt is made to relate these models to data from an actual intersection, and to make an assessment of their realism.

1.3. Previous relevant work.

Very little appears to have been published on this problem, the relevant papers being summarised in section 1.3.2. There has been a fair amount of work on delays in simpler but similar situations, and many of the ideas and methods developed in these discussions are useful. A brief account of this work is given below.

"Left-hand drive is the convention observed.

1.3.1. Previous discussions of delays to m-vehicles at a priority intersection.

Many authors have considered the problem of the delay caused to vehicles entering a major road from a particular minor road at a priority intersection. The major road is almost always regarded as containing a single traffic stream, and this assumption is made throughout this section. Further assumptions often made are that the traffic in the major road is unaffected by the presence of vehicles in the minor road, so that for instance an M-driver would not give way to a waiting vehicle, and that the flow of traffic on the major road is not disrupted by the untimely emergence of vehicles from the minor road. Stochastic equilibrium of the system is assumed in most discussions.

Tanner (1962) proposes a model for this situation for which he derives a formula for the mean delay to m-vehicles. The timeintervals between successive arrivals of vehicles in both M- and m-streams follow an exponential distribution, and subject to the following restrictions, vehicles pass through the intersection instantaneously. An M- (m-) vehicle may not pass through the intersection within a fixed time β_1 (β_2) of the previous M- (m-) vehicle, but must, where appropriate, delay its passage to conform with this condition. An m-vehicle may not enter the intersection within a fixed time a (> β_1) of the previous M-vehicle. There is no restriction on how closely an M-vehicle may follow an m-vehicle.

The purpose of the parameter β_1 in this model is to allow traffic on the major road, as viewed by m-drivers, to be nonrandom in that it would conform to the output of a single-server queue with Poisson arrivals, in which the queue discipline is first-comefirst-served, and the service time is constant. In this way the model simulates groups of vehicles at minimum separation followed by larger gaps. The result is that the exit from the minor road is 'blocked' for intervals distributed with the 'Borel-Tanner' distribution, and is free for intervals having an exponential distribution. Tanner's analysis is, however, valid for a general distribution of duration of block.

In order to extend Tanner's model and analysis to the situation in which m-vehicles seek to cross or merge with either of two streams of traffic on the major road, the distribution of blocks in the double M-stream must be specified. This problem does not appear to have been solved. For a two-stream extension of the Borel-Tanner distribution one requires the busy-period distribution of two queues in parallel, each of the above type, where the system is understood to be busy if at least one server is occupied.

Weiss and Maradudin (1962) propose a model which is based on the idea of a 'gap acceptance function', previously proposed by Cohen, Dearnaley and Hansel (1955). This function is defined as follows.

We suppose that at the instant when an m-vehicle arrives at the intersection, there is no queue in the minor road, and that the nearest approaching vehicle on the major road requires an interval of length t to reach the intersection. The gap acceptance function a(t) gives the probability that the driver of the m-vehicle accepts the gap and joins the M-stream.

We observe that Tanner's model uses a gap acceptance function which has the particular form of a step-function, because the model is essentially unaltered if a block is presumed to start an interval $a^{\circ}(< a)$ ahead of the arrival of the first of a group of N-vehicles, and to finish an interval $a - a^{\circ}$ after the last vehicle of the group. Weiss and Maradudin claim greater realism in their model on the grounds that an m-driver cannot infallibly distinguish between gaps greater than and less than a° .

Weiss and Maradudin exploit the gap acceptance function and the methods of Renewal Theory to derive solutions of a number of problems connected with the delay to m-vehicles at a priority intersection. They assume that the gaps between successive vehicles in the (single) M-stream are independently and identically distributed, and that M-vehicles cross the intersection instantaneously, so that the set of arrival instants forms a reneval process. As they are principally concerned with a single m-vehicle the distribution of arrival instants in the m-stream is not specified.

They determine the distribution of delay to a single m-vehicle which arrives at a random instant in time to find no other waiting m-vehicles. The probability that the m-vehicle moves off after the nth subsequent M-vehicle is evaluated.

Several modifications are now made to the original model. By varying the form of the gap acceptance function used by an m-vehicle for successive delaying M-vehicles, they construct a model which includes an 'impatient driver' characteristic. The distribution of delay to a randomly arriving, non-queueing m-vehicle is found for this model. In another version of the model the distribution of successive time gaps between M-vehicles (in the original model) is made dependent on real time. From this the equations governing the distribution of delay to an m-vehicle, which arrived at real time T_0 , say, to find no other queueing vehicle, are formulated in the nonstationary case. These equations are not solved.

Two alternative specifications of N-traffic are also considered. In the first of these, an attempt is made to simulate an n-lane major road by finding the distribution of time to the first arrival in an n-stream major road, when the distribution of arrivals in each stream is independent of arrivals in other streams, and forms a renewal process in time. In the other, the distribution of a gap in a single N-stream depends on the types of the vehicles which define the gap, each vehicle now being considered as belonging to one of R types.

Weiss and Maradudin then apply the simplest of their models to the case of a stream of m-vehicles delayed by the M-stream. They are unable to analyse the resulting model. They do succeed in generalizing their simple model in a more limited way by deriving the distribution of delay experienced by each of a pair of m-vehicles which arrive simultaneously at the intersection.

Contemporary with the paper of Weiss and Maradudia is one by Cohen and Stam (1963). These authors analyse a model which makes some use of the gap acceptance function and also of what may be called a 'follow-on probability'. The latter is defined as follows. Suppose that an m-vehicle arrives at the intersection after waiting in a queue for a non-zero interval. Then, without reference to vehicles approaching on the major road, but en a purely random basis, the driver of this vehicle decides instantaneously whether or not to follow the preceding wehicle from the minor road. That is, with probability k, the 'follow-on probability', he follows the preceding vehicle, and with probability 1 - k he does not follow, but awaits the arrival of the next vehicle on the major road. Once he has made a decision to move, a driver completes his movement through the intersection instantaneously, and so the next driver in the queue (if any) is immediately in a position to make an independent decision in accordance with the

same follow-on probability. It is possible in this way for a queue instantaneously to decrease in size by several vehicles.

Vehicles on the minor road which arrive at the intersection without queueing, or which have delayed to await the passage of one or more M-vehicles, use a gap acceptance function : all other m-vehicles use a follow-on probability. It will be observed that no driver uses the follow-on probability more than once. It is assumed that the arrival instants of M-vehicles form a Remewal process, and those of m-vehicles form a Poisson process.

The follow-on probability is intended to reflect to some extent the behaviour of drivers in the real situation, insofar as they find it easier to merge with the M-traffic when they are moving than when they are starting from rest. However, this rule has a further advantage. Cohen and Stam regard the total delay to a particular m-vehicle, which joins the end of a queue of n m-vehicles, as the sum of the 'completion times' of these n + 1vehicles (partial completion time of the first), where 'completion time' refers to the interval elapsing from the arrival of a vehicle at the head of the queue until its departure. They show that, for the model with follow-on probabilities, the completion times of the n + 1 queueing vehicles are independent. In this way they reduce the problem of delay to one involving the sum of independently distributed random variables. Although they evaluate only the

mean delay, it would presumably be possible by this approach to determine higher moments of the delay distribution.

In a later section of the same paper the model proposed by Weiss and Maradudin is considered, in which the essential change is that the follow-on probabilities are replaced by gap acceptance functions; in this case the probability that of m (> n) queueing vehicles the first n only accept a gap of t is $a^{n}(t)(1 - a(t))$ instead of $a(t)(1 - k)k^{n-1}$. It seems that the property of independence of the completion times of the vehicles in a queue is not possessed by this model. The determination of the delay distribution by the previous method would not be possible, although results concerning the mean delay would be unaffected.

Numerical results for both of the models proposed in the paper by Cohen and Stam are provided in a subsequent paper by Cohen and de Lange (1965).

Hawkes (1966) considers two congestion situations, the first of which concerns the delay to vehicles turning left or right from a minor road onto a major road which has two-way traffie, consisting of a single stream in each direction. Traffic on the major road is Poisson in each lane, and the arrivals of m-wehicles form a Poisson process. Vehicles on the minor road turn either left or right, and do so independently of previous vehicles. The completion of a left turn requires an interval of at least T_1 (a constant) between successive nearaide vehicles on the major road, and drivers of right turning vehicles must await a gap of at least T_2 (a constant $> T_1$) in the combined flow of nearside and farside streams. It is assumed that vehicles depart instantaneously, and that there is a constant 'move-up time' a, which is the minimum separation with which successive m-vehicles may depart. Left- and right-turning vehicles form a single queue in the minor read.

Hawkes uses an embedded Markov Chain to find the distribution of delay to m-vehicles under the assumption that $a \leq T_1 \leq T_2 \leq 2a$. These restrictions are necessary to ensure the validity of the Markov formulation employed. If, for instance, $T_2 > 2a$, then the gap with which an m-vehicle arriving at the head of a queue is confronted may depend, not only on the previous m-vehicle, but on several previous to it.

Like Tanner, Hawkes makes use of a step-function as a gap acceptance function. The cost of generalising the analysis of Tanner to the two-lane major road is, in this case, that the analysis is valid only for random arrivals in the M-streams, in contrast to the analysis of Tanner. Hawkes states that the basic restrictions on the parameters of his model may not be satisfied in some actual situations, where it appears that T_2 has a value of about 2a. Although it would appear that this model would be most applicable at intersections where the behaviour of traffic, at least

insofar as gap acceptance is concerned, is fairly uniform, as with the other models considered here any final assessment of the validity of the model would require detailed comparison of its calculated behaviour with actual observations.

The second model proposed in this paper by Eawkes is for delays to M-vehicles at a priority intersection and is more appropriately considered in the next section, 1.3.2.

In an earlier paper (1965) Haskes discusses a model for the queueing by m-vehicles caused by a series of "blocks' and "gaps" in the single stream major road. Blocks have arbitrary known distribution and gaps are distributed exponentially. Vehicles in the minor road arrive in bunches at random instants of time. The number of m-vehicles in a bunch is distributed independently of the numbers in previous bunches with known distribution. m-vehicles may depart, singly, only during a gap, and after a departure a fixed time a, the 'move-up time' must elapse before another queueing vehicle may depart, should the exit be unblocked at this time. Hovements of m-vehicles through the intersection are completed instantaneously. The distribution of delay to m-vehicles is determined.

Oliver and Bisbee (1962) assume that the distribution of gaps is general, that an m-vehicle requires a time interval T to move out, and that no two m-vehicles may depart in the same gap. For this model they determine the distribution of queue-size at the instants when an m-webicle departs.

The model proposed by Weiss (1965) for delays to m-vehicles at an intersection makes use of a modification of the gap acceptance function described above. If is supposed that a driver at the head of a queue in the minor road requires a time interval of length T to turn into the major road. T is sampled from a distribution with density f(T). If the gap G presented to the driver is greater than T, the driver completes his turn, blocking the intersection for exactly time T. At the end of this time, the next driver in the quoue (if any) bases his decision in a similar way on the gap G - T with which he is presented.

In this model both m- and M-vehicles arrive at random in time, and Weiss evaluates in the case of stationary behaviour the probability generating function for queue length at the instants at which M-vehicles cross the intersection.

1.3.2. Previous discussions of delays to M-vehicles at a priority intersection.

In this section we are concerned with the delay caused to vehicles on the major road by vehicles on this road which turn into the minor road. As the delays to other M-vehicles caused by those M-vehicles which turn left or go straight on are relatively

insignificant, we classify M-vehicles into two groups: type I vehicles turn left or go straight through the intersection; type II vehicles turn right. We are now interested in the delays to M-vehicles arising from the movements of type II vehicles.

The second problem discussed by Hawkes (1966) is concerned with the delay to M-vehicles caused by type II vehicles at a T-junction, which we may regard as an intersection at which only one of the two M-streams contains type II vehicles. Queues may form behind a type II vehicle which does not cross immediately: it is assumed to block its lane completely. The model used by Hawkes is similar to that of the earlier section of the paper, which is discussed above in section 1.5.1.

Another model, originally discussed by Newell (1959) and subsequently modified by Haight (1963, §6.3), is as follows. The model is for a priority intersection, but time is regarded as divided into small intervals ('slots'), in each of which the system remains in a certain state, and transitions from one state to another are made discontinuously between one slot and the next. In each lane of the major road at a particular slot one of the following describes the situation at the intersection.

- (a) A vehicle is ready to go straight through the intersection or to turn left.
- (b) A vehicle is ready to turn right.
- (c) No vehicle is present.

Natural rules are given which between each pair of slots allow the departure of one vehicle or both if at least one vehicle is present at the intersection. These rules are such that the departure of a straight or left-bound vehicle from one lane will prevent the departure of an opposing right-bound vehicle between any pair of slots.

Haight and Nevell analyse this system by defining the "states" at each slot as follows.

State 1 : no vehicle is delayed.

State 2 : a northbound vehicle is delayed.

State 3 : a southbound vehicle is delayed.

In order to describe the arrivals of vehicles in the system, it is convenient to make use of the positions occupied by vehicles immediately prior to their entry to the intersection. These positions are either capty or occupied. If empty, such a position may be occupied at each successive slot with constant probability, entirely independently of the other aspects of the system. Vehicles in these positions are of types I or II independently of the remainder of the system with probabilities constant for each lane.

A matrix of transition probabilities is defined on the states and state probabilities are obtained for the case of stochastic equilibrium. These may be used to find expressions for mean delays and to deduce conditions for stability of flow.

This model seems to be only an approximation to the actual situation, even if time is supposed to proceed by discrete steps. for the following reason. In an actual situation, the probability that the position adjacent to the intersection is filled appears to depend on the extent to which a previous right-turning vehicle in this position has been delayed. It is possible that a queue would form during any prolonged arrest of this vehicle. Thus the model applies exactly to the situation where a vehicle, which arrives in a given lane only to find that lane blocked by a delayed rightturning vehicle, merely disappears, and thus no quoues form and the only delays in the system are caused by vehicles actually at the intersection, rather than waiting in a queue. While it is appreciated that all mathematical models are idealisations to varying degrees, it seems that an essential feature of the actual situation. nemely the possibility of the formation of queues, has been neglected in this model.

1.3.3. The relevance of these models to the present problem.

(1) <u>Discrete time process</u>. Although some authors have treated the arrival process as operating in discrete time, it is apparent that the characteristics of such a process must be based on an examination of the actual continuous process, with the result that most authors

prefer to try to construct a continuous time model.

(ii) <u>Poisson Arrival Process in time.</u> As is well known, this process has several properties leading to mathematical simplification, and it has often been used in this context. The fact that the probability density for the inter-arrival distribution, $\lambda e^{-\lambda t}$, has its greatest value at the origin would lead to difficulties if vehicles of finite size were considered. Thus for this arrival distribution it is customary to replace vehicles by geometrical points in the model, and to attempt to compensate this further idealisation by the construction of rules for the behaviour of vehicles which simulate finiteness of size.

(iii) <u>Poisson arrival process with bunched arrivals</u>. In that it is more flexible this process would presumably be an improvement on the Poisson process. However the difficulties arising from the latter are considerable and there is not much prospect of progress with bunched arrivals at present.

(iv) <u>Reneval Process.</u> Similar remarks to those of (iii) are applicable.

(b) Mechanian of Congestion. In this lies the characteristic feature of the problem, and in the models which follow the attempt is made to achieve greater realism by modification of this mechanism. Some of the devices used by previous authors are applicable, notably those of Weiss (1965). (c) <u>Methods of Analysis.</u> Except for a small section of the paper by Weiss and Maradudin (1962), all of the above work concerns the case of stochastic equilibrium, and the present analysis is confined to this case. The analysis uses the Markov Chain associated with an embedded sequence of regeneration points. (D. G. Kendall (1953)).

Chapter 2

Construction and Analysis of Model I.

2.1 Derivation of the model.

In attempting to construct a model for the type of intersection under investigation, we may take advantage of two characteristics of actual intersections which may lead to simplifications in any models to be considered.

The first such characteristic is that it is usually the case that little change in the delay to M-vehicles would result from the diversion of each left turning vehicle to a straight course. As in section 1.3.2, vehicles are therefore classified into only two types, and we recall that

type I vehicles go straight on or turn left, type II

vehicles turn right.

We would expect that in some sense type I vehicles would have priority over type II vehicles in the model.

The other simplification is a consequence of the fact that queues are not simultaneously present in both lanes at the intersection. At any instant, either vehicles are passing freely through the intersection without delay, or there is a queue present in one lane. The simplest method of incorporating this feature in a model is to stipulate that a vehicle at the intersection can be delayed by an approaching vehicle in the opposing lane only when the types of the two confronting vehicles are different, and that then at most one of the vehicles is delayed. Thus, in the model, a vehicle ready to cross the intersection (i.e. unimpeded by vehicles in the same lane) should be delayed only in the following circumstances, and not necessarily even then. These circumstances are that the vehicle about to cross should be of type II, and that the next vehicle to arrive in the opposite lane should be of type I.

In such a situation it might be conjectured that the driver of the stationary type II vehicle would base his turning decision on several factors, as well as the type of nearest wehicle in the opposite lane. The most obvicus factor would be some measure of distance to the opposing type I vehicle, but distances to subsequent vehicles in the opposite lane might also be considered. It is a matter of observation that it is a comparatively infrequent occurrence for any consideration at all to be given by such a driver to vehicles other than the nearest in the opposing lane, and it is reasonable to suppose that his measure of the distance of this vehicle is equivalent to an assessment of the time due to elapse until its arrival at the intersection. In this first exploratory model the situation is deliberately simplified by the omission from the model of this feature of dependence of the decision on the size of the available gap.

2.2. Description of model I.

(i) Layout.

The model consists of the intersection of a two-lane major road with a minor road. Vehicles on the minor road have no effect on traffic on the major road, as there are halt signs on the minor road and the intersection is not controlled by traffic lights or by a policeman. Thus the minor road may be regarded as serving only as a means of departure of M-vehicles from the system.

The lanes of the major road are labelled 1 and 2 respectively. (ii) Vehicles.

For the purposes of the following analysis m-vehicles are ignored.

Vehicles are represented by geometrical points. The following restrictions are imposed on the movements of these points.

a) They may not pass through each other.

b) They may not traverse the geometrical bounds of the model.

As in section 2.1, vehicles are classified into types I and II.

(iii) Vehicle-arrivals.

The arrivals of individual vehicles at the intersection are specified by the instants at which they arrive either at the intersection, or at the tail of a queue of vehicles. Since in the model vehicles possess zero length, the latter is geometrically equivalent to arrival at the intersection. The idea of a queue of vehicles, as a time-ordered group of one or more stationary vehicles in a particular lane, is however preserved.

In a given lane the set of arrival instants forms a stochastic process in time, and the intervals between arrivals follow independent exponential distributions, with parameter λ_i for lane i, i = 1,2.

With probability p_i an arriving vehicle in lane i is of type I and with probability q_i it is of type II, where $p_i + q_i = 1$, i = 1,2. The type of this vehicle is independent of the type of previous arrivals.

(iv) Vehicle-interactions.

To simulate the movement of individual vehicles through an actual intersection, it is assumed in the model that as soon as a vehicle is 'ready' to move it does so and clears the intersection instantaneously. It is 'ready' to move only when it is the leading vehicle of a queue (which could consist solely of this vehicle), and the driver of the vehicle has decided to move. His decision is made instantaneously upon his arrival in the leading position in accordance with the following rules.

- a) A type I vehicle at the head of a queue in either lane moves as soon as it attains that position.
- b) The driver of a type II vehicle at the head of a queue in lane i (i = 1,2) takes into account in making his decision only

the type of the nearest opposing vehicle. If this is of type II he moves immediately. If it is of type I, with probability r_i he decides to move immediately, and with probability $1 - r_i$ he decides to await the passage through the intersection of the opposing type I vehicle. The driver of a vehicle which delays makes a fresh decision, following this same rule, as soon as the opposing type I vehicle has completed its crossing. Each decision made by the driver of a type II vehicle confronted by several type I vehicles in succession is independent of any previous decisions made by him.

These rules suffice to determine completely the behaviour of vehicles in the model. The only delays occurring are those due to the failure of a type II vehicle to move when faced with an approaching type I, i.e. those due to right-turning wehicles.

2.3. Steady-state behaviour of Model I.

2.3.1. Selection of regeneration points.

Before attempting to select suitable instants of time for consideration as regeneration points for the model, we observe certain general aspects of its behaviour which follow immediately from the description above. It is apparent that, at any instant of time, either there is a queue in one lane of the model, or there is no queue in either lane. A queue is necessarily headed by a
type II vehicle, and the nearest opposing vehicle is of type I. Queues may decrease in size only at the instants immediately following the departure of an opposing type I vehicle. At such an instant one or several vehicles may depart instantaneously, or the driver of the leading vehicle may decide not to move. Queues may increase in size at any instant.

It is convenient to choose as regeneration points the instants at which a vehicle crosses the intersection in either lane. Each such instant is described by (i) the lane from which the vehicle departs, (ii) the number of vehicles (if any) queueing at that instant, and (iii) the type of the nearest vehicle in the other lane, if there is no queue in that lane. To avoid ambiguity in the cases where several vehicles depart simultaneously, item (i) will be taken to refer to the lane which contains the single type I vehicle whose passage has released the queue, and it will be the initial or largest size of the queue which is quoted in item (ii). Item (111) is made necessary by the requirement that the description of the system at a regeneration point should be complete in the sense that no further information about its previous history is required to specify probabilities relating to its future behaviour. The omission of item (iii) from the description of the system would mean that this requirement was not always met, for example in the case where a II in lane 1 decided to move when opposed by a I in lane 2, and before the arrival of this I a further II arrived in lane 1. It is in fact exactly those situations which give rise to the necessity for item (iii) above which cause difficulty in the attempt to analyse, by a similar system of regeneration points, a more complicated model based on the gap acceptance methods of Cohen and Stam. For certain of these points to carry an adequate description of the system, it would be necessary for this description to include the length of the time interval due to elapse before the arrival of the nearest opposing vehicle. As the use of a continuous variable for descriptive purposes at a regeneration point would imply that the number of such points is not denumerable, the method of the Embedded Markov Chain cannot be used.

2.3.2. Notation for steady state analysis.

The following notation is used for reference to the set of regeneration points of \$2.3.1.

 $R_i(n)$ is the label given to an instant at which a vehicle departs from lane 1, and at which there are $n(\ge 1)$ vehicles queueing in the opposing lane. It is understood that the remarks of §2.3.1. concerning 1 and n will apply.

 $R_i(-1)$, $R_i(-2)$ refer to instants at which a vehicle departs from lane i with no queue in the opposing lane, in which the next arrival is of type I, II, respectively.

The interval between successive regeneration points will be

called a transition interval, and it may be partly specified by reference to the types of regeneration points at its extremities. Thus an interval commencing with an $R_i(m)$ and ending with an $R_j(n)$, (i, j = 1,2; m, n = -2,-1,1,2,...) will be referred to as of type $\{i, j; m, n\}$. A transition interval is defined to include the <u>initial</u> but not the final regeneration point, so that vehicles may leave the intersection only at the initial instant of a transition interval.

It is assumed that the behaviour of the system is stochastically stationary, and that parameter values are consistent with this assumption. We define 'state probabilities' as follows:

 $P_i(n)$ denotes the probability that a regeneration point selected at random is of type $R_i(n)$, i = 1, 2; n = -2, -1, 1, 2, ...

It is convenient also to define transition probabilities for the probability that a transition interval is of a particular type, conditional on the type of the initial regeneration point of the interval.

Additional variables may be used to specify a transition interval with greater precision, thus we refer to an interval of type $\{i, j; m, n\}$, of duration 7, and having a queue of length k in lane i immediately following the initial instant, as being of type $\{i, j; m, n; 7, k\}$. The probability density that an interval is of type $\{i, j; m, n; 7, k\}$, conditional on the initial regeneration

29.

point being of type $R_i(m)$, will be denoted by $\pi(i, j; m, n; \gamma, k)$; and $\pi(i, j; m, n)$ will denote the probability that an interval is of type $\{i, j; m, n\}$, conditional on the initial point being of type $R_i(m)$.

2.3.3. Transition probabilities.

The transition probabilities are determined by arguments similar to the following for $\pi(1,1; m,n)$, m,n > 0.

The procedure is to determine separately the quantities

 $\pi(1,1; m,n; 7, m-r), 1 \le r \le m-1;$ $\pi(1,1; m,n; 7, m);$ $\pi(1,1; m,n; 7, 0);$

and to make use of the fact that

$$\pi(1,1; m,n) = \int_{0}^{00} \sum_{r=0}^{m} \pi(1,1; m,n; \gamma, m-r) d\gamma$$

The quantity $\pi(1,1; m,n; \gamma, m-r)$, $1 \le r \le m-1$, refers to transitions from an $R_1(m)$ to an $R_1(n)$. It is the probability density for the joint occurrence of the events listed below, conditional on the presence of a queue of m vehicles in lane 2 and the departure of a type I vehicle from lane 1 at the initial instant. These events are

 (i) the next vehicle in lane 1 is of type I and it arrives at a time 7 after the initial instant;

- (ii) the leading vehicle (of type II) in lane 2 moves prior to the arrival of this type I vehicle, as do the following r-1 vehicles (of assorted type) in the queue - i.e. they all move at the initial instant;
- (iii) the next vehicle in the queue is of type II and the driver decides to await the passage through the intersection of the opposing type I vehicle;
- (iv) the number of arrivals in lane 2 during the subsequent time interval 7 is n-m+r.

It follows that

 $\pi(1,1; m,n; \gamma, m-r) = \lambda_1 p_1 e^{-\lambda_1 \gamma} \cdot r_2 (p_2 + q_2 r_2)^{r-1} \cdot q_2 (1-r_2) \cdot \frac{e^{-\lambda_2 \gamma} (\lambda_2 \gamma)^{n-m+r}}{(n-m+r)!} \cdot \epsilon (n-m+r),$ $1 \le r \le m-1,$

where e(n) = 1, $n \ge 0$,

= 0, n < 0.

Similarly $\pi(1, 1; m, n; \gamma, m)$ and $\pi(1, 1; m, n; \gamma, 0)$ are evaluated. Writing $f_i = p_i + q_i r_i$, $\lambda = \lambda_1 + \lambda_2$, we have $\pi(1, 1; m, n; \gamma, m) = p_1 \lambda_1 (1-r_2) e^{-\lambda \gamma} \frac{(\lambda_2 \gamma)^{n-m}}{(n-m)!} e(n-m)$, $\pi(1, 1; m, n; \gamma, 0) = \int_0^{\infty} \lambda_1 p_1 e^{-\lambda_1 (t+\gamma)} \cdot \lambda_2 q_2 e^{-\lambda_2 t} (1-r_2) e^{-\lambda_2 \gamma} \frac{(\lambda_2 \gamma)^{n-1}}{(n-1)!}$

. f2 m-1 r2 dt,

and so

R.,

$$\begin{aligned} & \pi(1,1; \quad m,n) = p_1 \lambda_1 (1-r_2) \left\{ \frac{1}{\lambda} \left(\frac{\lambda_2}{\lambda} \right)^{n-m} e(n-m) \\ & + \frac{r_2 q_2 f_2^{m-n-1}}{\lambda_1 + \lambda_2 q_2 (1-r_2)} \left[1 - \left(\frac{\lambda_2 f_2}{\lambda} \right)^n \right] e(n-n-1) \\ & + \left(\frac{\lambda_2}{\lambda} \right)^{n-m+1} \frac{r_2 q_2}{\lambda_1 + \lambda_2 q_2 (1-r_2)} \left[1 - \left(\frac{\lambda_2 f_2}{\lambda} \right)^{m-1} \right] e(n-m) \\ & + \frac{r_2 q_2}{\lambda} f_2^{m-1} \left(\frac{\lambda_2}{\lambda} \right)^n \right]; \quad m, n > 0. \end{aligned}$$

The complete set of 36 transition probabilities for medel I is given in Appendix I. A check on the transition probabilities is provided by the set of consistency relations

oo 2

$$\Sigma \Sigma = \pi(i, j; m, n) = 1, \quad i = 1, 2; \quad m = -2, -1, 1, 2, ...$$

 $n=-2 \quad j=1$
 $n=0$

2.3.4. Evaluation of state probabilities.

The state probabilities satisfy the stationarity equations

$$\rho_{j}(n) = \sum_{\substack{m=-2 \ i=1 \\ m\neq 0}}^{\infty} \sum_{\substack{m=-2 \ i=1 \\ m\neq 0}}^{\infty} \pi(i, j; m, n) \rho_{i}(n), \ j = 1, 2; \ n = -2, -1, 1, 2, \dots,$$
(1)

together with the normalising relation

These equations are now solved for the state probabilities.

If we define
$$\widetilde{H}_{1}(x) = \sum_{n=1}^{\infty} \rho_{1}(n) x^{n}$$
, $i = 1, 2; |x| \leq 1$, multiplication
of each of the equations of (i) by an appropriate power of x followed
by summation reduces the set (1) to the equivalent set
 $\lambda_{1}p_{2}r_{1}\rho_{2}(-2) = [\lambda_{2} + \lambda_{1}q_{1}(1-r_{1})]\rho_{1}(-1) + \lambda_{1}p_{2}\rho_{2}(-1)$
 $+ \lambda_{1}p_{2}q_{1}r_{1}\widetilde{H}_{1}(1) + (\lambda_{1}p_{1}p_{2}r_{2}/f_{2})\widetilde{H}_{1}(f_{2}) + \lambda_{1}p_{2}r_{1}\widetilde{H}_{2}(f_{1}) = 0,$ (3)
 $\widetilde{K}_{1}(x) \left[\frac{p_{1}\lambda_{1}(1-r_{2})(p_{2}-x)}{(\lambda-\lambda_{2}x)(f_{2}-x)} - 1 \right] + \frac{\lambda_{1}\lambda_{2}(1-r_{2})}{\lambda} \cdot \frac{x}{(\lambda-\lambda_{2}x)} \cdot \left[p_{1}\rho_{1}(-2)+q_{2}\rho_{2}(-1) \right]$
 $+ \widetilde{K}_{1}(f_{2}) \frac{p_{1}\lambda_{1}(1-r_{2})r_{2}q_{2}Q_{2}}{f_{2}} \cdot \frac{x}{(\lambda-\lambda_{2}x)} \left(\frac{1}{f_{2}-x} + \frac{\lambda_{2}}{\lambda} \right)$
 $+ \widetilde{K}_{2}(1) \cdot \frac{p_{1}q_{2}(1-r_{2})\lambda_{1}\lambda_{2}}{\lambda} \cdot \frac{x}{(\lambda-\lambda_{2}x)} = 0,$ (4)
and
 $\lambda_{2}\rho_{1}(-2)-\lambda_{1}q_{2}\rho_{2}(-2)-\lambda_{1}q_{2}\rho_{2}(-1)-\lambda_{1}q_{1}q_{2}\widetilde{H}_{1}(1)-\lambda_{1}q_{2}\widetilde{H}_{2}(1)-\frac{\lambda_{1}q_{2}p_{1}r_{2}}{f_{2}} \widetilde{K}_{1}(f_{2}) = 0.$
(5)
Equation (2) is
 $\rho_{1}(-2)+\rho_{2}(-2)+\rho_{1}(-1)+\rho_{2}(-1)+\widetilde{H}_{1}(1)+\widetilde{H}_{2}(1) = 1.$ (6)

Corresponding to each of equations (3), (4), (5) there is an equation obtained by interchanging lane suffices. These equations are labelled (3^{K}) , (4^{K}) , (5^{K}) respectively.

The construction of equations (3), (4), (5) has introduced certain constants which are themselves combinations of the unknown state probabilities. These are $\tilde{R}_1(1)$, $\tilde{R}_2(1)$, $\tilde{H}_1(f_2)$ and $\tilde{R}_2(f_1)$. Before proceeding further we evaluate these constants, together with the quantities $\rho_1(-2)$, $\rho_2(-2)$, $\rho_1(-1)$, $\rho_2(-1)$. For this purpose, we have equations (3), (5^E), (5), (5^E), together with an additional pair, found by setting x = 1 in (4), (4^E),

$$\lambda_{2}(1-r_{2}) \left[p_{1}\rho_{1}(-2) + q_{2}\rho_{2}(-1) \right] - \frac{p_{1}r_{2}}{f_{2}} (\lambda_{1}+\lambda_{2}f_{2})\widetilde{R}_{1}(f_{2}) + p_{1}q_{2}(1-r_{2})\lambda_{2}\widetilde{R}_{2}(1) - q_{1}\lambda\widetilde{R}_{1}(1) = 0, \qquad (7)$$

with a corresponding equation (7^{H}) .

When the coefficients of these six equations are inspected, it is found that only five are linearly independent, so that, on taking into account equation (6), we have in all six independent relations between the eight unknown quantities. A further independent pair of equations may be derived as follows by an argument based on the fact that the functions $\overline{R}_i(z)$ are analytic within the unit circle.

Writing k for $\lambda_2 f_2/\lambda$, so that k < 1, equations (4) and (7) reduce to

$$\widetilde{R}_{1}(\mathbf{x}) = \frac{\mathbf{x}\ell(\mathbf{x})}{(\mathbf{x}-\alpha)(\mathbf{x}-\beta)} + \frac{\alpha(f_{2}-\mathbf{x})(\mathbf{x}-1)}{(\mathbf{x}-\alpha)(\mathbf{x}-\beta)} \widetilde{R}_{1}(\mathbf{k}\mathbf{x}), \quad (8)$$
where $\ell(\mathbf{x}) = \frac{\lambda_{1}}{\lambda_{2}} \frac{\mathbf{p}_{1}\mathbf{r}_{2}}{f_{2}} (1-\mathbf{x}) \widetilde{R}_{1}(f_{2}) + \frac{\lambda_{1}}{\lambda_{2}} q_{1}(f_{2}-\mathbf{x}) \widetilde{R}_{1}(1),$

$$\mathbf{c} = \mathbf{p}_{1}q_{2}\mathbf{r}_{2}(1-\mathbf{r}_{2})\lambda/f_{2}(\lambda_{1}+\lambda_{2}q_{2}(1-\mathbf{r}_{2})),$$
and $\lambda_{2}(\mathbf{x}-\alpha)(\mathbf{x}-\beta) = \lambda_{2}\mathbf{x}^{2}-\mathbf{x}(\lambda_{2}f_{2}+\lambda-\lambda_{1}\mathbf{p}_{1}(1-\mathbf{r}_{2}))+\lambda f_{2}-\mathbf{p}_{1}\mathbf{p}_{2}\lambda_{1}(1-\mathbf{r}_{2}),$

$$\approx S(\mathbf{x}) \text{ say.}$$

Since $S(f_2) > 0$, S(1) < 0 and the coefficient of x^2 in S(x) is positive, it follows that one root of S(x) = 0 lies between f_2 and 1 and the other is greater than 1, i.e. a < 1, $\beta > 1$.

Equation (8) may be further rewritten

$$\widetilde{R}_{1}(x) = h(x) + g(x) \widetilde{R}_{1}(kx),$$

to which a formal solution, obtained by repeated substitution, is

$$\widetilde{R}_{1}(x) = h(x) + g(x)h(kx) + g(x)g(kx)h(k^{2}x) + \dots$$

$$= \frac{x}{(x-\alpha)(x-\beta)} \left\{ \ell(x) + \sum_{n=1}^{\infty} \frac{k^{n}c^{n}\ell(k^{n}x) \prod_{i=0}^{n-1} (f_{2}-k^{i}x)(k^{i}x-1)}{\prod_{i=1}^{n} (k^{i}x-\alpha)(k^{i}x-\beta)} \right\}$$

By definition, $\tilde{R}_1(z)$ is an analytic function of z at $z = \alpha$, so that its numerator must vanish at $z = \alpha$, i.e.

$$\ell(\alpha) + \sum_{n=1}^{\infty} k^n c^n \ell(k^n \alpha) \prod_{i=0}^{n-1} \left\{ (f_2 - k^i \alpha) (k^i \alpha - 1) / (k^{i+1} \alpha - \alpha) (k^{i+1} \alpha - \beta) \right\} = 0,$$

which is equivalent to

$$\frac{P_{1}r_{2}}{r_{2}}\tilde{R}_{1}(r_{2})\left\{ \begin{array}{l} & & \\ 1-\alpha+\sum k^{n}c^{n}(1-k^{n}\alpha) \prod_{i=0}^{n-1} \left[(r_{2}-k^{i}\alpha)(k^{i}\alpha-1)/(k^{i+1}\alpha-\alpha)(k^{i+1}\alpha-\beta) \right] \right\} \\ +q_{1}\tilde{R}_{1}(1)\left\{ r_{2}-\alpha+\sum k^{n}c^{n}(r_{2}-k^{n}\alpha) \prod_{i=0}^{n-1} \left[(r_{2}-k^{i}\alpha)(k^{i}\alpha-1)/(k^{i+1}\alpha-\alpha)(k^{i+1}\alpha-\beta) \right] \right\} \\ = 0. \quad (9)$$

Equation (9) and the corresponding equation (9^{a}) constitute the additional pair of linear equations required to evaluate the eight

constants, $\tilde{R}_{1}(1), \dots, P_{2}(-1)$.

By means of equation (8) we may now calculate the state probabilities $\rho_1(n)$, $n \ge 1$. We define

$$y_{1} = \frac{p_{1}\lambda_{1}r_{2}}{f_{2}}\tilde{R}_{1}(f_{2}) + q_{1}\lambda_{1}f_{2}\tilde{R}_{1}(1),$$

$$y_{2} = \frac{p_{1}\lambda_{1}r_{2}}{f_{2}}\tilde{R}_{1}(f_{2}) - q_{1}\lambda_{1}\tilde{R}_{1}(1),$$

$$a_{0} = p_{1}\lambda_{1}(1-r_{2}) - \lambda_{2}f_{2} - \lambda,$$

$$a_{1} = \lambda f_{2} - p_{1}p_{2}\lambda_{1}(1-r_{2}),$$

$$b_{0} = p_{1}(1-r_{2})r_{2}q_{2}\lambda\lambda_{2}/f_{2}(\lambda_{1}+\lambda_{2}q_{2}(1-r_{2})),$$

$$b_{1} = -(1+f_{2})b_{0},$$

$$b_{2} = f_{2}b_{0}.$$
(9)

Writing for convenience $\rho_1(n) = u_n$, so that $\tilde{R}_1(x) = \sum_{i=1}^{\infty} u_i x^i$, (8) is equivalent to

$$u_{1}(a_{1} + kb_{2}) = y_{1},$$

$$u_{2}(a_{1} + k^{2}b_{2}) + u_{1}(a_{0} + kb_{1}) = y_{2},$$

$$u_{n}(a_{1} + k^{n}b_{2}) + u_{n-1}(a_{0} + k^{n-1}b_{1}) + u_{n-2}(\lambda_{2} + k^{n-2}b_{0}) = 0, n \ge$$
(10)

If we regard the first n equations of the set (10) as nonhomogeneous linear equations in the unknowns u_1, u_2, \dots, u_n , we may write the solution for u_n , $n \ge 3$, in the determinantal form 3.

37.

see e.g. Jordan (1950), p.587. In practice, however, the numerical evaluation of Δ_n for large n seems to be most easily achieved by regarding Δ_n as the solution of a system of recurrence relations, and so it is just as satisfactory to attempt the solution of (10) directly.

It is not possible to solve (10) by calculation of $u_1, u_2, u_3, ...$ from successive equations of the set, as it is found that an ultimately divergent sequence is obtained. This may be understood upon consideration of the asymptotic form of (10). The recurrence relation becomes almost exactly

$$\mathbf{a}_{1}\mathbf{u}_{n} + \mathbf{a}_{0}\mathbf{u}_{n-1} + \lambda_{2}\mathbf{u}_{n-2} = 0 \tag{(11)}$$

for moderately large n, and (11) has an auxiliary quadratic with roots a^{-1} , β^{-1} . Thus $u_n \sim A\beta^{-n} + Ba^{-n}$ and since $a^{-1} > 1$, $\beta^{-1} < 1$, it is the solution with B exactly zero which is required. This solution may not be reached by the forward solution of (10) from u_1, u_2 , owing to the accumulation of round-off error.

The procedure adopted here involves the backward solution of the system (10). We assume that, for some number N, the values of u_n (n > N) are sufficiently small to make a negligible contribution to the value of the expression for whose calculation the $\{u_n\}$ is required. In the determination of the mean delay it is found that the quantity most likely to be affected by inaccuracies in the $\{u_n\}$ is the second moment of the distribution $\{u_n\}$. This second moment is calculated repeatedly, for a selection of suitably spaced and increasing values of N, using the values of $\{u_n\}$ approximated by the method to be described. This process is continued until substantial increases in the value of N result in comparatively small changes in the second moment. At this point we consider N adequately determined.

B complete the approximate determination of $\{u_i\}$, we note that for large n, the ratio of u_{n-1} to u_n is effectively β . A set of quantities u_n° , $n = 1, \ldots N$, satisfying (10) and with $u_N^{\circ} = 1$, $u_{N-1}^{\circ} = \beta$, are found by backward solution of (10), the first pair of equations of the set being ignored. The first equation of (10) gives a value for u_1 , and the approximation to u_n is then

$$u_n = u_n^\circ \times u_1/u_1^\circ, \quad n \leq \mathbb{N},$$
$$= 0, \qquad n > \mathbb{N}.$$

Checks on these approximations are provided by

- (i) comparison of the value for u₂ with the value given by the the second equation of (10);
 N
- (ii) comparison of Σ u with the known value of $\widetilde{R}_1(1)$. n=1

2.3.5. The mean delay.

By 'delay' to a vehicle we understand the interval elapsing from its arrival at the tail of a queue, or at the intersection if no queue is present, until the instant it crosses the intersection. In this section the mean delay to vehicles in lane 2 is determined. In view of the Poisson arrival process in lane 2, the formula mean delay = (mean queue length) / (arrival rate).

(Kendall 1951)

may be used for this purpose, where 'mean queue length' refers to the mean queue length left by a departing vehicle.

The method used here to determine this mean queue length is to average its value for a particular type of transition interval over all transition intervals. The average formed is weighted for each interval by the number of vehicles leaving lane 2 during the interval, since by 'mean' we imply selection of a vehicle rather than a point in time. Thus the number of vehicles leaving lane 2 during an interval must be included in the description of the interval.

In any interval, the only instant at which departures in lane 2

occur is the initial instant. If there are n vehicles queueing in lane 2 just prior to the initial instant, and if in the initial instant the number of queueing vehicles is reduced to k_s the mean queue length as seen by a vehicle departing in this interval is $\frac{1}{2}(n+k-1)$.

We denote by $\pi(1, j; m, n; k)$ the probability that a regeneration point of type $R_1(m)$ is followed by a point of type $R_j(m)$, and that there are k vehicles in lane 2 immediately following the initial instant. If we further define by ω_2 the proportion of transition intervals having as initial point $R_2(n)$, n = -2, -1, 1, 2, ..., then Prob. [a vehicle selected at random in lane 2 departs during

an interval of type {1, j; m, n; k}

Thus the mean queue length left by a vehicle departing from lane 2

 $\begin{cases} \Sigma & \Sigma & (m+k-1)(m-k)\pi_{1}(m)\pi(1, j; m, n; k) \\ j, m, n & k=0 & \\ & \Sigma & \Sigma & (m^{0}-k^{0})\pi_{1}(m^{0})\pi(1, j^{0}; m^{0}, m^{0}; k^{0}) \\ & j^{0}, m^{0}, n^{0} & k^{0}=0 & \\ \end{cases}$

as no contribution is made to the numerator by those intervals which commence with the passage of zero or one vehicles in lane 2. From this expression numerical values for the mean queue length and hence mean delay may be derived.

It has been pointed out that the mean queue length as seen by a departing vehicle is in fact equal to the mean queue length in time. If we denote the respective means by $E(q_{d})$ and $E(q_{t})$, then in the equilibrium case, we have for the time interval (t, t+bt),

expected total delay experienced a expected total delay generated,

i.e. $E \{q_t \delta t\} = \lambda_2 \delta t E \{v\}$, where v is the delay of a given vehicle, so that $E \{q_t\} = \lambda_2 E(v) = E \{q_d\}$.

2.3.6. Numerical results.

Using a Ferranti Sirius computer equations (10) were solved, and the mean delays computed for several sets of parameter values. In table 2.1, λ_i refers to number of vehicles per unit time, and the mean delay is measured in these time units.

Table 2.1.

| 9 ⁹ 1 | ¥1 | λ ₁ | 9 ² 2 | r2 | ^ک 2 | Nean delay in lang 2. |
|------------------|----|----------------|------------------|----|----------------|--------------------------|
| •5 | .4 | 10 | •5 | .4 | 18 | . 201 |
| -5 | .4 | 26 | •5 | .4 | 10 | .030 |
| •5 | .4 | 10 | -5 | .1 | 10 | .207 |
| •5 | .4 | 10 | • 3 | .2 | 10 | .205 (a) |
| .5 | .4 | 10 | • 3 | .2 | 5 | .184 |
| -5 | .4 | 20 | - 3 | .2 | 10 | .092 (b) |

Results of calculations for mean delay.

2.4. Remarks on model I.

As explained in section 2.1, model I was constructed with mathematical tractability in mind, and in order to see whether an analysis of its steady-state behaviour might be possible. Although such an analysis has been achieved, it is of interest to assess the practical value of the model before it is abandened as merely a step towards more realistic ones. We consider this from two points of view, namely the plausibility of the initial assumptions and of the results of calculations of the mean delay.

2.4.1. Validity of the initial assumptions.

In the model, the replacement of vehicles by geometrical points is intended to simplify the mathematical description of both the arrival process and the mechanism of congestion. In an actual intersection, congestion largely results from the fact that the dimensions of vehicles and the widths of lanes are of the same order of magnitude. There is no provision in the model, other than the rule for delaying type II vehicles, for simulating this aspect of the socual situation. In fact, any unrealistic behaviour of the model resulting from the assumption of point-vehicles would tend to be emphasized by the additional assumptions of zero orcessing-times and decision-times for vehicles and drivers.

It is just possible, of course, that these assumptions may be reasonable approximations in some situations where, on the average, delay-times considerably exceed crossing-times. Again, the rule which allocates to right-turning vehicles the same probability of moving, no matter how near an opposing I might be, might be regarded as reasonable in a situation where the visibility of drivers is restricted, or where vehicles are moving very slowly.

The assumption of random arrivals may not be completely justified by the actual traffic flow at a particular intersection, but it is nevertheless the case that a fundamental simplification in the analysis of the model is dependent upon this assumption. It should

43.

therefore be retained in some form if at all possible. There is a considerable mass of experimental evidence of varying quality suggesting that the Poisson model is adequate for vehicle arrivals at points along roads some distance from any congestion mechanism. This form of vehicle distribution may be demonstrated to be the asymptotic form assumed by traffic possessing certain fairly plausible initial velocity and position characteristics. (See e.g. Haight (1963) Chapter 4).

The assumed constancy of arrival rates should reasonably approximate the actual situation, at least over limited periods of observation.

2.4.2. Predicted behaviour of model I.

It is characteristic of actual intersections that the total number of vehicles per unit time which may cross the intersection is limited. This leads to the idea of the capacity of an intersection, and traffic engineers are much concerned with an appropriate definition of this quantity. If the vehicular flow to the intersection exceeds its capacity, the behaviour of the system ceases to be stationary.

It is interesting to determine what restrictions may be necessary on the values of the parameters in order that the behaviour of the model should be stationary. We see that, with

A.L.

probability zero, queues increase indefinitely in this model (for non-zero p_i, q_i), since any queue may be completely released, instantaneously, by the arrival of an opposing II. Stationary behaviour of the system is therefore possible for any set of parameter values. This characteristic of the model may be regarded as a serious deficiency, and it would tend to restrict the successful application of the model to an actual intersection to those situations where traffic is relatively light.

Inspection of table 2.1 suggests that it may be possible, for at least some values of p_i, r_i , to decrease delay in lane 2 by increasing the arrival rate of vehicles in lane 1, other parameters remaining fixed. (c.f. rows (a), (b)). This apparently unrealistic behaviour may be understood by observing that by increasing λ_1 we merely offer more frequent crossing opportunities to a delayed II in lane 2, with a consequent reduction in mean waiting-time in lane 2. In practice an increase in λ_1 would imply a decrease in r_2 - the existence of a relationship between r_1 and λ_2 and between r_2 and λ_1 might be considered.

The behaviour of model I does not seen to correspond more closely with actual traffic than one would expect from the considerably idealised initial assumptions. The simplicity of the model would allow of its analysis by methods other than that of the Embedded Markov Chain, but it appears that the value of such an exercise would be mainly some of.

A description of model I is given in Reid (1967).

Chapter III

Construction and Analysis of Model II.

3.1. Derivation of the model.

A serious defect of model I is the assumption that a driver's decision to turn right is not affected by the proximity of an opposing vehicle. We have seen in section 2.3.1. that modifications to model I based on the gap acceptance function as used by, e.g. Weiss and Maradudin (1962), are unlikely to be tractable, partly because of the manner in which such a function requires drivers to make use of the entire available gap in reaching their decision.

It seems reasonable to suppose that the interest of a turning driver in a gap in the opposing lane is concerned solely with its adequacy for his intended crossing. We may imagine that he has in mind an interval a, which will be called his 'gap-requirement time', and which is the minimum time he requires for starting up and moving clear of the opposing stream of vehicles. His decision whether to cross is made by comparing the available gap with a: if it exceeds a, he crosses without besitation, otherwise postponing his decision until the opposing vehicle has cleared the intersection. If we regard the intersection as blocked in both lanes for the gaprequirement time a, and make the assumption of Poisson traffic, the interval from the end of this blocked period until the passage of

4.2.

the original opposing vehicle will have an exponential distribution. Thus any vehicles arriving at the intersection during this interval, possibly from a queue, will be confronted with random opposing arrivals, simplifying considerably the analysis of a model based on such a scheme.

It is a matter of observation, however, that drivers tend to overestimate the time they require to complete a crossing. Thus in an actual situation there is a certain interval T following the block caused by a turning vehicle which is the remainder of the interval estimated by the driver to be necessary for his movement. Although, in a model, T might be regarded as part of the gap requirement time a, this would imply that the driver of the vehicle queueing behind the vehicle which has crossed and given rise to T would not start to make his decision until the end of the interval T. This might be reasonable if he was required to make a distinct movement to the head of the queue before making his decision, and if the time he required for this movement was on the average about equal to T.

Although it would be possible to discuss methods of interpreting a in greater detail, possibly by investigating the relative sizes of separable components of a, and by considering the accuracy of a driver's assessment of the distance of an opposing vehicle, it is felt that little would be gained because of the impracticability of

4%

measurements of these quantities. We leave further discussion of the suitability of this method of describing the gap-acceptance mechanism until chapter 5. Similar types of gap-acceptance mechanism have been proposed by Gaver (1965) and Weiss (1965).

In model II the driver of a turning vehicle will be taken to make use of a gap requirement time in the manner described above, so long as his position at the head of a queue is a consequence either of his arrival at the intersection with no queue in either lane, or of a previous decision which he has made not to nove in the face of an opposing vehicle. Drivers in a given lane are assumed to have access to a population of gap-requirement times, from which a particular driver selects one for a given situation uninfluenced by any previous choices of himself or other drivers.

Those drivers in model II who arrive at the head of a queue when one or more preceding members of the queue move off, make an immediate decision in the same way as drivers in model I, and if they decide to cross, do so instantaneously. This assumption is intended to reflect the tendency of actual drivers to 'follow-on', i.e. to follow instantaneously almost without regard to encoming traffic a preceding vehicle which is making the crossing. One would expect a model incorporating such an assumption to be easier to analyze than one in which all turning drivers made use of a gap requirement criterion.

48.

3.2. Description of model II.

(i) Layout, vehicles, vehicle-arrivals.

Model II differs from model I only in the manner in which vehicles interact at the intersection. The description of the layout, vehicles, and arrival process is as given in section 2.2, (i), (ii), (iii).

(ii) Vehicle-interactions.

We distinguish the vehicles which arrive at the intersection to find one or more preceding vehicles delayed in the same lane, from those whose arrival at the intersection is uninterrupted by preceding vehicles in their lane : we refer to arrivals 'from a queue' in the former case and 'in free flow' in the latter. 'Crossing' and 'turning' refer to movements complete to the extent that the intersection is left clear.

The rules for the interaction of vehicles at the intersection are :

- (a) A type I vehicle, which arrives at the intersection either in free flow or from a queue, crosses immediately.
- (b) A type II wehicle which arrives at the intersection in free flow or from a queue, or which has already delayed its departure from the head of a queue by either of rules
 (c), (d) below, turns immediately when the nearest vehicle in the opposite lane is also of type II.

(c) The driver of a type II vehicle in lane i which arrives at the intersection in free flow, or which has already delayed its crossing by either of the rules to be described here and in rule (d), acts as follows when the nearest vehicle in the opposite lane is of type I.

He measures the time interval due to elapse, of length \mathcal{T} , from the instant at which his vehicle is in a position to move (i.e. either its instant of arrival or the instant at which an opposing type I vehicle departs), until the arrival of the opposing vehicle. He selects a gap-requirement time' a from a distribution with density $f_i(a)$. If $a \leq \tau$, he delays for an interval a and then instantaneously completes his turn, thus blocking lane i for an interval of length a. If $a \geq \tau$ he awaits the arrival and crossing of the opposing type I vehicle, before again assessing the situation by means of rules (b), (c). Each decision made by a driver in this way is independent of any previous decisions made by him or other drivers.

(d) A type II vehicle in lane i, which arrives at the intersection from a queue and which finds the nearest opposing vehicle to be of type I, turns immediately with probability k₁; or awaits the crossing of the type I vehicle with probability i - k₁; whereupon it uses whichever of rules (b), (c) is appropriate.

50.

The quantity k_i is the 'follow-on probability'. Each decision made by a driver in this way is independent of any previous decisions made by him or others.

Rules (a) - (d) suffice to determine completely the interaction of vehicles in the model. As in model I, the only delays to M-vehicles are those due to the presence of right-turning vehicles.

3.3. Steady-state behaviour of Model II.

3.3.1. Selection of regeneration points : notation.

It is obvious that, at any instant of time, a queue may exist in at most one lane of the model. Queues may be augmented at any instant of time. The instants at which queues may decrease in size or vanish are of two types. A queue in lane i may disappear immediately after the crossing of an opposing type I vehicle, if the next opposing vehicle is of type II. Otherwise, a delayed type II vehicle may turn after pausing for its gap-requirement time, to be followed immediately by several other vehicles of assorted type. Thus the complete or partial release of a queue in model II is not always immediately preceded by the crossing of a I in the other lane. Recognition of this aspect of the behaviour of the model is of assistance in the construction of an Embedded Markov Chain. It is assumed that the system has attained a state of stochastic stationarity. As in model I we identify as regeneration points those instants at which a vehicle crosses the intersection in either lane. In the case where several vehicles depart simultaneously, we regard the 'first' vehicle to cross the intersection, in the sense of section 2.2 (iii), as defining the regeneration point. The description of the system at a regeneration point must include the following information :

- (i) the lane from which the first vehicle departs,
- (ii) if a queue exists, its lane and size,
- (iii) if there is no queue, the type of the nearest vehicle in the lane other than that from which the vehicle departs.

The notation used to represent regeneration points is as follows :

- $R_i(n)$ refers to an instant at which a vehicle necessarily of type I, crosses in lane i, in the presence of a queue of n vehicles in the <u>other</u> lane (n = 1, 2, ...);
- $S_{i}(n)$ refers to an instant at which a vehicle necessarily of type II, crosses in lane 1, in the presence of an <u>additional</u> n queueing vehicles in the same lane (n = 1, 2, ...).

 $\begin{array}{c} R_{i}(-1) \\ R_{i}(-2) \end{array} \begin{array}{c} \text{refer to instants at which a vehicle of either type crosses} \\ I \\ \text{in lane i, there being nonliqueue present, and the next opposing} \\ \text{vehicle being of type} \begin{cases} I \\ II \end{cases} \end{array}$

Thus for instance $S_1(n)$ refers to an instant at which a vehicle of

type II turns from lane 1, after delaying in this lane for its gap-requirement time a, in the presence of a further queue of n vehicles in lane 1. The next arrival in lane 2 is a type I vehicle.

Because of the properties of the exponential distribution, the description of the system at each regeneration point is sufficiently complete to enable all subsequent probabilistic behaviour of the model to be inferred without knowledge of its previous history. Vehicles cross the intersection only at the regeneration points : they may arrive at any instant of time.

We define the state probabilities for the Embedded Markov Chain as follows :

$$\begin{array}{c}
P_{i}(n) \\
\sigma_{i}(n) \\
P_{i}(-1) \\
P_{i}(-2)
\end{array}$$
is the probability that a randomly
$$\begin{array}{c}
R_{i}(n) \\
S_{i}(n) \\
R_{i}(-1) \\
R_{i}(-2)
\end{array}$$

$$\begin{array}{c}
R_{i}(n) \\
R_{i}(-1) \\
R_{i}(-2)
\end{array}$$

(n = 1, 2, ...).

3.3.2. Transition Probabilities.

We now consider in detail the chain of events which determines the type of regeneration point which follows a regeneration point of given type, and evaluate corresponding transition probabilities. There are 64 types of transition such as $R_1(n) \gg R_1^{-1}(m)$, but by symmetry we may reduce the number for study to 32. The example below illustrates the method :

 $S_1(m) \rightarrow R_2(n)$

At the initial instant a type II vehicle is instantaneously completing its turn from lane 1, with a further queue of m vehicles in this lane; the driver of this type II vehicle has already used the information that the next arrival in lane 2 will be of type I.

We write for brevity "at τ " for "in $(\tau, \tau + d\tau)$ ". For the next regeneration point to be of type $R_2(n)$, one of the following sequences of events must occur.

Either $r(0 \le r \le m-1)$ vehicles depart from lane 1 immediately following the initial vehicle, after which the driver of the $(r+1)^{th}$ vehicle decides to await the arrival of the opposing type I vehicle, distant time τ , and during this time a further n-m+r vehicles arrive in lane 1.

Or, all the vehicles in lane 1 cross or turn immediately and the next vehicle in lane 1 (due an interval γ_1 later) is of type II, and arrives prior to the arrival of the type I vehicle in lane 2 (which occurs after an interval $\tau + \tau_1$, say). The driver of this type II vehicle decides to await the arrival of the opposing type I, and in the remaining interval, of length τ , a further n-1 vehicles accumulate in lane 1.

With p_i , q_i , λ_i , λ , ϵ , defined as for model I (section 2.2 (iii), 2.3.3), if we write $F_i(\tau) = \int_0^{\tau} f_i(t) dt$, $s_i = p_i + q_i k_i$, the probability of this transition is

$$\int_{0}^{\infty} \sum_{r=0}^{n-1} \lambda_{2} e^{-\lambda_{2} \mathcal{T}} (\mathbf{p}_{1}+q_{1}\mathbf{k}_{1})^{r} q_{1} (1-\mathbf{k}_{1}) e^{-\lambda_{1} \mathcal{T}} \frac{(\lambda_{1} \mathcal{T})^{n-m+r}}{(n-m+r)!} d\tau e^{(n-m+r)}$$

$$+ \int_{0}^{\infty} \int_{0}^{\infty} \lambda_{2} e^{-\lambda_{2} (\mathcal{T}+\mathcal{T}_{1})} (\mathbf{p}_{1}+q_{1}\mathbf{k}_{1})^{m} e^{-\lambda_{1} \mathcal{T}_{1}} \lambda_{1} q_{1} \left[1-\mathbf{F}_{1}(\mathcal{T})\right] e^{-\lambda_{1} \mathcal{T}} \frac{(\lambda_{1} \mathcal{T})^{n-1}}{(n-1)!} d\tau d\tau_{1}$$

$$= \frac{\lambda_{2} q_{1} (1-\mathbf{k}_{1})}{\lambda} \left(\frac{\lambda_{1}}{\lambda}\right)^{n-m} \sum_{r=0}^{m-1} \left(\frac{\mathbf{e}_{1} \lambda_{1}}{\lambda}\right)^{r} e^{(n-m+r)}$$

$$+ \frac{\lambda_{2} q_{1}}{\lambda} \frac{\lambda_{1}^{n} \mathbf{e}_{1}^{m}}{(n-1)!} \int_{0}^{\infty} \left[1-\mathbf{F}_{1}(\mathcal{T})\right] e^{-\lambda \mathcal{T}} \mathcal{T}^{n-1} d\tau .$$

It is found that of the 32 basic types of transition, 9 have probability 0. The transition $R_1(m) \Rightarrow R_2(-1)$ has positive probability only in the case m = 1. The complete set of transition probabilities for model II is given in Appendix I. The transition probabilities may be verified by use of the consistency relations of section 2.3.3.

3.3.3. Determination of State Probabilities.

The state probabilities satisfy the usual set of stationarity equations together with a normalising condition.

We define

and

$$\widetilde{H}_{i}(x) = \sum_{n=1}^{\infty} x^{n} \rho_{i}(n), \quad |x| \leq 1;$$

$$\widetilde{S}_{i}(x) = \sum_{n=1}^{\infty} x^{n} \sigma_{i}(n), \quad |x| \leq 1;$$

$$f_{i}^{x}(s) = \int_{0}^{\infty} e^{-st} f_{i}(t) dt, \quad \mathcal{R}(s) > 0$$

The stationarity equations then reduce to equations (1) to (4) below, together with equations (1^{N}) to (4^{N}) obtained from these by an interchange of lane suffices. The equations are

$$(\lambda - \lambda_{2} \mathbf{x}) \tilde{\mathbf{R}}_{1}(\mathbf{x}) = \lambda_{1} \left[1 - f_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) \right] \left\{ \frac{\lambda_{2}}{\lambda} \mathbf{x} (\mathbf{p}_{1} \rho_{1}(-2) + q_{2} \rho_{2}(-1)) + \mathbf{p}_{1} \tilde{\mathbf{R}}_{1}(\mathbf{x}) \right.$$

$$+ \frac{\lambda_{2}}{\lambda} \mathbf{p}_{1} q_{2} \mathbf{x} \tilde{\mathbf{R}}_{2}(1) + \frac{\lambda_{2}}{\lambda} q_{2} \mathbf{x} \tilde{\mathbf{S}}_{2}(\mathbf{s}_{2}) \right\}$$

$$+ \frac{\lambda_{1} q_{2}(1 - \mathbf{k}_{2}) \mathbf{x}}{(\mathbf{s}_{2} - \mathbf{x})} \left\{ \tilde{\mathbf{S}}_{2}(\mathbf{s}_{2}) - \tilde{\mathbf{S}}_{2}(\mathbf{x}) \right\},$$

$$(1)$$

(by considering transitions into $R_1(n)$);

$$\widetilde{S}_{1}(\mathbf{x}) = \frac{\lambda_{1}}{\lambda} \left(f_{1}^{\#}(\lambda - \lambda_{1}\mathbf{x}) - f_{1}^{\#}(\lambda) \right) \left\{ q_{1}\rho_{1}(-1) + p_{2}\rho_{2}(-2) + q_{1}p_{2}\widetilde{R}_{1}(1) \right. \\ \left. + q_{1}\widetilde{S}_{1}(\mathbf{s}_{1}) \right\} \\ \left. + \frac{p_{2}}{\mathbf{x}} \quad \widetilde{R}_{2}(\mathbf{x}) f_{1}^{\#}(\lambda - \lambda_{1}\mathbf{x}) - p_{2}f_{1}^{\#}(\lambda)\rho_{2}(1) \right],$$

(transitions into S₁(n));

$$\lambda \rho_{1}(-1) = \lambda_{1}(p_{1}+q_{1}f_{1}^{*}(\lambda))(\rho_{1}(-1)+\tilde{S}_{1}(s_{1}))+\lambda_{1}p_{2}(\rho_{2}(-1)+f_{1}^{*}(\lambda)\rho_{2}(-2) +\tilde{S}_{2}(s_{2})+q_{1}f_{1}^{*}(\lambda)\tilde{R}_{1}(1))+\lambda p_{2}f_{1}^{*}(\lambda)\rho_{2}(1), \qquad (3)$$

(transitions into $R_1(-1)$); and

$$\lambda_{2} P_{1}(-2) = \lambda_{1} q_{2} (P_{2}(-1) + P_{2}(-2) + q_{1} \tilde{R}_{1}(1) + \tilde{R}_{2}(1) + \tilde{S}_{2}(B_{2})),$$
(transitions into $R_{1}(-2)$). (4)

Equations (1) and (2^{E}) and (1^{E}) and (2) may be solved simultaneously for $\tilde{R}_{i}(x)$, $\tilde{S}_{i}(x)$, and these functions are then determined as rational functions of x and known transforms, but containing as unknown quantities the set of constants $\rho_{1}(-2)$, $\rho_{2}(-2)$, $\rho_{1}(-1)$, $\rho_{2}(-1)$, $R_{1}(1)$, $\tilde{R}_{2}(1)$, $\tilde{S}_{1}(s_{1})$, $\tilde{S}_{2}(s_{2})$, $\rho_{1}(1)$ and $\rho_{2}(1)$.

It is convenient to use equations (3), (3^{m}) to substitute for the last pair of this set in terms of the others, leaving eight unknowns. We proceed to find a nonsingular set of eight equations linear in these unknowns, which we shall denote by $u_1, u_2, \ldots u_8$, respectively.

Setting x = 1 in the derived expression for $\tilde{R}_1(x)$ gives

 $p_1\lambda_2u_1 + p_1\lambda_2u_3 - \lambda_1u_4 - \lambda q_1u_5 + p_1q_2\lambda_2u_6 + p_1\lambda_2u_7 - \lambda_1u_8 = 0$, (5) together with a corresponding equation (5^N) derived from $\tilde{R}_2(x)$. The set (4), (4^N), (5), (5^N) are linearly dependent, since the coefficients of u, sum to zero (i = 1, 2, ... 8).

If in the normalising condition $P_1(-2) + P_2(-2) + P_1(-1) + P_2(-1) + \tilde{R}_1(1) + \tilde{R}_2(1) + \tilde{S}_1(1) + \tilde{S}_2(1) = 1$, the last two terms of the L.H.S. are replaced by formulae derived from the expressions for $\tilde{S}_i(x)$, we have a further equation in the $\{u_i\}$. Writing

$$\xi_{1} = \lambda_{1}(s_{2} - 1)(q_{1} + p_{1}f_{2}^{*}(\lambda_{1})),$$

$$\phi_{1} = \lambda_{1}(s_{2} - 1)(f_{2}^{*}(\lambda_{1}) - q_{1}f_{2}^{*}(\lambda) - p_{1}f_{2}^{*}(\lambda_{1})f_{2}^{*}(\lambda)),$$

with similar definition of x_2 , ϕ_2 for interchanged lane suffices, we have

$$\sum_{i=1}^{8} o_i u_i = 1, \qquad (6)$$

where $c_1 = 1 + \frac{\lambda_2 P_1}{\lambda \lambda_1 q_1 (e_2 - 1)} \left\{ \phi_1 + f_2^{E}(\lambda) \zeta_1 \right\}$

$$c_{3} = 1 + \frac{1}{\lambda\lambda_{1}q_{1}(a_{2}-1)} \left\{ \lambda_{2}p_{1}\xi_{1} + \lambda_{1}q_{1}\phi_{2} - \lambda_{1}\xi_{2} \left[\frac{\lambda_{1}}{\lambda_{1}} - (p_{1}+q_{1}f_{1}^{\#}(\lambda)) \right] \right\},$$

$$c_{5} = 1 + \frac{\lambda_{1}q_{1}p_{2}}{\lambda\lambda_{2}q_{2}(a_{1}-1)} \left\{ \phi_{2} + f_{1}^{\#}(\lambda)\xi_{2} \right\},$$

$$c_{7} = \frac{\lambda_{2}p_{1}\xi_{1}}{\lambda\lambda_{1}q_{1}(a_{2}-1)} + \frac{\lambda_{1}}{\lambda\lambda_{2}q_{2}(a_{1}-1)} \left\{ q_{1}\phi_{2} + \frac{\lambda\lambda_{2}p_{2}(1-a_{1})f_{1}^{\#}(\lambda_{2})}{\lambda_{1}} + \left[p_{1}+q_{1}f_{1}^{\#}(\lambda) \right] \xi_{2} \right\},$$

and $c_{2r} = c_{2r-1}$ with lane subscripts interchanged, $r = 1, \dots, 4$. The statement $\tilde{S}_2(0) = 0$ is equivalent to

$$p_{1}s_{2}\lambda f_{2}^{\Xi}(\lambda)u_{1} + p_{1}s_{2}(\lambda-\lambda_{1}p_{1}+\lambda_{1}p_{1}f_{2}^{\Xi}(\lambda))u_{3} + s_{2}\left[\lambda_{1}p_{1}(1-f_{2}^{\Xi}(\lambda))(\frac{\lambda}{\lambda_{2}} - p_{2})\right]$$
$$-\lambda\left(\frac{\lambda}{\lambda_{2}} - p_{2} - q_{2}f_{2}^{\Xi}(\lambda)\right) u_{4} + p_{1}q_{2}s_{2}\lambda f_{2}^{\Xi}(\lambda)u_{6} + p_{1}s_{2}(\lambda-\lambda_{1}p_{1}+\lambda_{1}p_{1}f_{2}^{\Xi}(\lambda))u_{7}$$
$$+ \left[\frac{\lambda\lambda_{1}}{\lambda_{2}}p_{1}(1-s_{2})f_{2}^{\Xi}(\lambda) + \lambda s_{2}q_{2}f_{2}^{\Xi}(\lambda) + s_{2}p_{2}(\lambda-\lambda_{1}p_{1}+\lambda_{1}p_{1}f_{2}^{\Xi}(\lambda))\right] u_{8} = 0. \quad (7)$$

There is a corresponding equation (7^*) .

The expressions for $\tilde{H}_1(x)$, $\tilde{S}_2(x)$ have a common denominator $D_1(x)$, where $D_1(x) = (s_2 - x)(\lambda - \lambda_1 p_1 - \lambda_2 x) + \lambda_1 p_1(1 - x) f_2^{\mathbb{B}}(\lambda - \lambda_2 x)$, after removal of irrelevant factors. Since $D_1(x)$ is continuous and $D_1(0) > 0$, $D_1(1) < 0$, there is at least one zero of $D_1(x)$ in (0, 1). It is shown in section 3.3.6 that $D_1(x)$ has exactly one zero in this interval, at say $x = a_1$. It follows by the definition of the functions $\tilde{R}_1(x)$, $\tilde{S}_2(x)$ that the numerators of the expressions for these functions must also vanish at $x = a_1$. If we write

 $\Psi_1 = a_1 - s_2 + (1-a_1)f_2^{\#}(\lambda-\lambda_2a_1)$

the condition for
$$R_1(x)$$
 is

$$p_{1}\psi_{1}u_{1} + p_{1}(1-s_{2})u_{3} + (q_{2}\psi_{1} - (1-s_{2})(\frac{\lambda}{\lambda_{2}} - p_{2}))u_{4} + p_{1}q_{2}\psi_{1}u_{6} + p_{1}(1-s_{2})u_{7} + (q_{2}\psi_{1} - (1-s_{2})(\frac{\lambda}{\lambda_{2}} - p_{2}))u_{8} = 0.$$
(8)

A similar equation (8^{m}) may be derived by consideration of $\tilde{R}_{2}(x)$ at $x = a_{2}$; the equations based on the numerators of $\tilde{S}_{2}(a_{1})$, $\tilde{S}_{1}(a_{2})$ give no new information. (This statement and the next are based on numerical work: complexity has so far defeated analytic investigation.)

Equations (4), $(4^{\overline{x}})$, (5), (6), (7), $(7^{\overline{x}})$, (8), $(8^{\overline{x}})$ constitute a linearly independent set from which $\{u_i\}$ may be determined. We have thus determined $\rho_i(-2)$, $\rho_i(-1)$, $\overline{R}_i(x)$ and $\overline{S}_i(x)$, which are required for the determination of the distribution of delay.

3.3.4. The distribution of delay.

The distribution function for the delay experienced by vehicles

in lane i is denoted by $B_i(x)$, where it is understood as in §2.3.5 that the delay to a vehicle is the total time during which the vehicle is at rest at the intersection. The delay to vehicles in lane 2 is considered in this section.

Let us define

 $v_r = \text{prob.} \left[a \text{ wehicle departing in lane 2 leaves a queue of} \right]$ length r in lane 2, $\Delta_2(x) = \sum_{r=0}^{\infty} v_r x^r$, $|x| \leq 1$, r=0

and
$$B_2^{\overline{R}}(s) = \int_0^{\infty} e^{-st} dB_2(t)$$
.

Since the queue in lane 2 left by a driver departing from this lane consists of vehicles which have arrived during his delay time, of duration v, say, the distribution of the number of queueing vehicles, n, conditional on v is

$$\frac{(\lambda_2 \mathbf{v})^n}{n!} = e^{-\lambda_2 \mathbf{v}}, \quad n = 0, 1, 2, ...,$$

and the unconditional distribution of n has probability generating function

$$\sum_{n=0}^{\infty} x^{n} \int_{0}^{\infty} \frac{(\lambda_{2} v)^{n}}{n!} e^{-\lambda_{2} v} dB_{2}(v)$$
$$= B_{2}^{\mathbb{M}}(\lambda_{2}(1-x)) = \Lambda_{2}(x).$$

Thus

$$B_2^{x}(x) = \Lambda_2\left(\frac{\lambda_{2-x}}{\lambda_2}\right), \quad |x| \leq 1, \text{ a standard result.}$$

We now construct an expression for $\Lambda_2(x)$.

The only regeneration points at which a vehicle departing from lane 2 may leave a non-empty queue in this lane are of type $R_1(n)$ or $S_2(n)$. Let k denote the total number of departures from lane 2 associated with such a point. For $R_1(n)$, k assumes one of the values zero and n, since at such a regeneration point the only way in which a non-zero number of vehicles may depart in lane 2 is by the next arrival in lane 1 being of type II. For departures in lane 2, if H_2 is a normalising constant,

prob. remaining queue is of length r and regeneration point is of type $R_{1}(n)$

$$\frac{1}{n} \times \frac{n\rho_1(n)q_1}{H_2} \in (n-1-r)$$

Similarly,

5-8

prob. remaining queue is of length r, regeneration point is of type $S_2(n)$, with a total of k departures, $1 \le k \le n$

$$= \frac{1}{k} \times \frac{k\sigma_2(n)s_2^{k-1}(1-s_2)}{H_2} e(n-r) e(r-n+k-1)$$

and

prob. remaining queue is of length r, regeneration point is of type $S_2(n)$, total of n+1 departures

$$= \frac{1}{(n+1)} \times \frac{(n+1)\sigma_2(n)s_2^n}{H_2} \varepsilon(n-r).$$

Here

$$H_{2} = \rho_{2}(-2) + \rho_{2}(-1) + \sum_{n=1}^{\infty} \rho_{2}(n) + \sum_{n=1}^{\infty} \sum_{k=1}^{n} k \sigma_{2}(n) s_{2}^{k-1}(1-s_{2})$$

+ $\sum_{n=1}^{\infty} (n+1) \sigma_{2}(n) s_{2}^{n} + q_{1} \sum_{n=1}^{\infty} n \rho_{1}(n)$
= $\rho_{2}(-2) + \rho_{2}(-1) + \widetilde{H}_{2}(1) + \frac{1}{(1-s_{2})} \left\{ \widetilde{S}_{2}(1) - s_{2} \widetilde{S}_{2}(s_{2}) \right\} + q_{1} \widetilde{H}_{1}(1),$

where the prime denotes differentiation, (c.f. 'length-biased sampling' in Cox and Lewis (1966), p. 61). H_2 is the mean number of vehicles departing from lane 2 at a randomly chosen regeneration point. Thus we have after a few lines of algebra

$$H_{2}A_{2}(\mathbf{x}) = \frac{q_{1}}{(1-\mathbf{x})} \left\{ \widetilde{R}_{1}(1) - \widetilde{R}_{1}(\mathbf{x}) \right\} + \frac{1}{(s_{2}-\mathbf{x})} \left\{ s_{2}\widetilde{S}_{2}(s_{2}) - \mathbf{x}\widetilde{S}_{2}(\mathbf{x}) \right\} \\ + \rho_{2}(-2) + \rho_{2}(-1) + \widetilde{R}_{2}(1),$$

and $B_2^{\overline{M}}(s)$ is therefore determined.

It may be deduced from this that the mean delay to arrivals in lane 2 is

$$\frac{1}{2H_2\lambda_2} \left\{ q_1 \tilde{R}_1^{\prime\prime} (1) + \frac{2s_2}{(1-s_2)^2} \left[\tilde{S}_2(s_2) - \tilde{S}_2(1) \right] + \frac{2}{(1-s_2)} \tilde{S}_2^{\prime} (1) \right\},$$

and that the second moment about the origin of the delay distribution
is, writing $\delta_2 = \lambda_2(1-s_2),$

$$\frac{1}{H_2} \left\{ \frac{q_1}{3\lambda_2^{2}} \widetilde{R}_1^{\prime\prime\prime\prime}(1) - \frac{2\lambda_2 B_2}{\delta_2^{3}} \left[\widetilde{S}_2(B_2) - \widetilde{S}_2(1) \right] - \frac{2B_2}{\delta_2^{2}} \widetilde{S}_2^{\prime\prime}(1) + \frac{1}{\lambda_2 \delta_2} \widetilde{S}_2^{\prime\prime\prime}(1) \right\} .$$
3.3.5. Numerical Results.

A Ferranti Sirius computer was used to calculate the mean and variance of the distribution of delay to vehicles in lane 2, for a selection of parameter values, under the assumption that the distribution of gap-requirement times had a translated negative exponential form. i.e.

$$T_i(\alpha) = 0, \quad \alpha \in T_i,$$

= $v_i \exp(-v_i(\alpha - T_i)), \quad \alpha > T_i.$

This choice of distribution was suggested by Herman and Weiss (1961), and is discussed in chapter 5.

For each of the sets of parameter values the value of k_i is chosen to be equal to the probability that a II at the intersection, faced with an opposing I sampled from the opposing stream, would accept the gap with which it is presented if it were to make use of the gap-requirement distribution for its lane. Thus k_i is fixed in terms of λ_i , ν_i , T_i , and, e.g. we set $k_1 = f_1^{E}(\lambda_2)$.

Results are quoted only for the case of lane-symmetric traffic, and are shown in table 3.1.

Comments on table 3.1.

An example of the effects of changes in the value of λ_i on the delay distribution is given in rows 2, 5, 19, 13, 20, 21. It appears that, as λ_i increases from the value 0.1, the mean delay

Table 3.1.

Results of calculations,

| row P _i | p | arane | têr v | alues | † | delay dista. in lane i | | | |
|--|---|------------------|-----------|-------|----------------------|---|---|--|--|
| | Pi | λ _i | Vi | Ti | k _{j.} | mean | variance | | |
| 1 2 3 | .2 .5 .8 | .1 | 1.0 | 2.5 | .708 | • 596 • 864 • 005 | 10.62 91.37 1214.85 | | |
| 4 5 6 7 8 9 | .2 .5 .8 .9 .93 .97 | .3 | | | .363 | .504 1.236 1.372 .864 .610 .21 | 2.50 7.82 22.93 70.72 142.33 | | |
| 10 11 12 13 14 15 16 17 18 | .10 .20 .35 .50 .65 .90 .95 .98 .99 | .9 | | | . 056 | .112 .240 .467 .772 1.239 2.142 4.505 3.853 2.494 | .24 .55 1.22 2.44 5.23 14.56 84.25 99.90 | | |
| 19 20 21 | •5 | .6 1.5 2.5 | | | .139 .009 .001 | 1.014 .494 .300 | 4.15 1.04 .39 | | |
| 22 23 | | .3 | .3 .05 | | .236 | 1.697 2.287 | 12.56 22.39 | | |
| 24 | | .9 | 1.0 | 5 | .006 | . 833 | 2.94 | | |

† Certain values are cmitted for clarify - in such cases the relevant value is the nearest above in the same column.

For this set of parameter values it was not possible to obtain a reliable value for the variance because of round-off error.

increases, passes through a maximum and then decreases. To explain this, we observe that for those sets of parameter values which correspond to mean delay decreasing with λ_i , the values of λ_i are such that the mean inter-arrival interval is smaller than T₄, so that it is unlikely that type II vehicles are released by either the gap-acceptance or follow-on mechanism. In this situation larger values of λ_i correspond merely to the more frequent arrivals of type II vehicles, with consequent opportunities for the complete dispersion of any opposing queues. It is therefore plausible that mean delay should be a decreasing function of λ_{i} . For comparatively small values of λ_i , by contrast (for the same values of p_i , v_i , T_i), the gap-acceptance and follow-on probabilities play an appreciable part in dispersing queues, and an increase in the value of λ_{i} , while implying the more frequent arrival of type II vehicles, may increase congestion, on balance, by sufficiently decreasing the probability that a type II vehicle at the head of a queue accepts the gap to an opposing I. We would expect this effect to be reduced for smaller values of p_i, and the calculations of rows 1, 4, 11; 3, 6, 15, support this conjecture.

With regard to the effect of the parameter p_i on the flow of vehicle through the intersection, we intuitively expect the mean delay to tend to zero as p_i tends to zero or unity, since in these limiting cases there should be no interference between vehicles each of the same type. This behaviour is reflected in rows 1, 2, 3; 4 - 9; 10 - 18. In each case, so far as it has been calculated, the variance is an increasing function of p_i . Computational difficulties prevent the determination of the variance for values of p_i close to unity.

Inspection of rows 5, 22, 23 confirms that, as expected, decreasing v_i results in an increase in the mean of the delay distribution. A similar result follows from an increase in T_i (rows 13, 24).

3.3.6. Proof of result used in section 3.3.3.

To show that $D_1(x)$ has only one zero in (0, 1).

If not, since $D_1(0) > 0$, $D_1(1) < 0$ and $D_1(x)$ is continuous, there are at least three zeros in this interval, so that $D_1^{\circ\circ}(x)$ must vanish at an interior point.

Now

$$D_{1}^{\circ \circ}(\mathbf{x}) = 2\lambda_{2} + \lambda_{1}P_{1}(-2 \frac{d}{d\mathbf{x}} f_{2}^{\mathbb{H}}(\lambda - \lambda_{2}\mathbf{x}) + (1 - \mathbf{x}) \frac{d^{2}}{d\mathbf{x}^{2}} f_{2}^{\mathbb{H}}(\lambda - \lambda_{2}\mathbf{x})),$$

and $\frac{d^2}{dx^2} f_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}) = \lambda_2^2 \int_0^\infty e^{-(\lambda - \lambda_2 \mathbf{x})u} u^2 f_2(u) du,$ > O_0

Also since us $k^{-1} \leq k^{-1} = 1$ for k > 0, u > 0, and $\int_{0}^{\infty} f_{2}(u) du = 1$,

$$\frac{\mathrm{d}}{\mathrm{d}\mathbf{x}} f_2^{\mathbf{x}} (\lambda - \lambda_2 \mathbf{x}) = \lambda_2 \int_0^\infty e^{-(\lambda - \lambda_2 \mathbf{x}) \mathbf{u}} u f_2(\mathbf{u}) \mathrm{d}\mathbf{u},$$
$$\leq \lambda_2 (\lambda - \lambda_2 \mathbf{x})^{-1} e^{-1}.$$

It follows that, for $0 \le x \le 1$,

$$D_{1}^{\circ \circ}(\mathbf{x}) > 2\lambda_{2} - 2\lambda_{1}P_{1}\lambda_{2}(\lambda-\lambda_{2}\mathbf{x})^{-1}e^{-1},$$

$$= \frac{2}{e(\lambda-\lambda_{2}\mathbf{x})} \quad (\lambda_{2}(\lambda-\lambda_{2}\mathbf{x})e - \lambda_{1}\lambda_{2}P_{1}),$$

$$= \frac{2}{e(\lambda-\lambda_{2}\mathbf{x})} \quad (\lambda_{1}\lambda_{2}q_{1}+\lambda_{1}\lambda_{2}(e-1) + \lambda_{2}^{2}e(1-\mathbf{x})),$$

$$> 0,$$

Hence $D_1(x)$ has only one zero in 0 < x < 1.

3.4. Remarks on Model II.

The use of the gap-acceptance mechanism in this model should result in a model which is both more flexible, in that greater allowance may be made for variations among individual drivers, and more realistic, than model I. In model II, the driver of a stationary right-turning vehicle may take a non-zero interval of time both to assess the situation with which he is confronted, and to complete his turn if he so decides.

On the other hand it is apparent that there are several aspects of the behaviour of vehicles at an actual intersection which are not represented in the model. For instance, in rule (ii) of section 3.2., when a type II vehicle in lane 1 is about to execute its turn, the nearest approaching vehicle in lane 2 being also of type II, no account is taken of the proximity of the vehicle immediately following the type II vehicle in lane 2. Also in rule (iv) of the same section, one would prefer the number of vehicles simultaneously 'following-on' to be limited in some way by the size of the available gap in the opposite stream. although the rule as it stands might be credible in those actual situations where drivers of type I vehicles occasionally give way to drivers of opposing type II vehicles. The seriousness of omissions such as these will vary with actual intersections under study.

The fact that the behaviour of model II is ergodic for all reasonable sets of parameter values, which is an immediate consequence of its property that queues of any length may be dissipated instantaneously, is a more general difficulty. As in model I this characteristic would make the model unsuitable for the simulation of high-flow situations.

It is interesting that model II, which appears to be more realistic than model I, leads to an analysis which is little more difficult than that of the earlier model.

Chapter IV

Construction and Analysis of Model III.

4.1. Derivation of the model.

Perhaps the most important defect of models I and II is their inability to reflect the non-stationary behaviour of an actual intersection under certain traffic conditions, and model III is particularly concerned with an attempt to remedy this deficiency.

We recall that the stationary character of the previous models is a consequence of the immediate dissipation of any queues present by the certain arrival of a type II vehicle in the opposing lane. A method is required which would, at least in some cases, replace this instantaneous process by a more gradual dissipation of the queue over a non-zero interval of time. One way of arranging this is to require at least some of the vehicles in a queue, for preference the type II vehicles, to delay and to use gap-requirement times when faced with opposing II's. The stipulation that all type II vehicles should make their decision in this way would lead to unrealistic behaviour. An example would be the situation where the driver of a II in lane 1, having arrived at the intersection without waiting in a queue, selects a gap-requirement time and on the basis of this awaits the passage of an approaching II in lane 2 before re-assessing the situation.

A type II vehicle will, therefore, be taken to make use of a

69.

gap-requirement criterion only when it is faced with an opposing I, or when the type II vehicle arrives at the intersection from a queue, and it is faced with an opposing II. There will be no 'follow-on' behaviour of the kind occurring in model II, where it is possible for a turning vehicle to follow immediately the preceding member of a queue through the intersection. The selection of gap-requirement times by type II vehicles faced with an opposing I will also be influenced by whether these vehicles arrive at the intersection from free flow or from a queue.

4.2. Description of model III.

- (i) The layout is as in models I, II.
- (ii) <u>Vehicles.</u> As in the previous models, these will be regarded as geometrical points of types I, II. Vehicles are further classified as to whether their arrival at the intersection is or is not consequent upon delay in a queue - the former are of type S (stationary) and the latter of type M (moving).
- (iii) <u>Vehicle-arrivals</u> occur as in models I, II.
- (iv) <u>Vehicle-interactions.</u> The behaviour of vehicles at the intersection is described below. We denote by "g. r. t.: f" the result that the vehicle at the intersection uses a gaprequirement time (as defined in section 3.2.(ii)(c)), chosen from a population with density f(t).

| Type of vehicle at intersection in lane i | Type of nearest vehicle in opposing lane | Result | | | | |
|--|---|--------------------------------------|--|--|--|--|
| I, M or S | I or II | No delay to veh. at intersection. | | | | |
| II, M | I | g.r.t. : f _i | | | | |
| II, M | II | No delay to veh. at intersection. | | | | |
| II, S | I | g.r.t. : g _i | | | | |
| II, S | II | g.r.t. : h _i | | | | |

This system of rules describes the behaviour of traffic in any situation which can arise. It is not possible for queues to exist simultaneously in both lanes, since any queue requires as its leading vehicle either a II (M or S) faced with an opposing I or a II (S) faced with an opposing II. The only delays occurring in the model are those caused by right-turning vehicles.

4.3. Steady-state behaviour of model III.

4.3.1. Selection of regeneration points : notation.

As in model II, it is possible for a queue in either lane in model III to be augmented at any instant. The instants at which queues decrease in size may be classified once again into those at which the initial departure (in the sense of section 2.3.1.) occurs in the lane in which the queue exists and those at which this departure occurs in the lane opposite the queue. It is of importance in assessing the future behaviour of the system that the type of the opposing vehicle should be known in the former case, since this vehicle may be of type I or II in model III, and this information would already have been used by the driver of the leading vehicle of the queue in making a decision to cross.

We shall consider the behaviour of the system under the assumption of stochastic stationarity, and we identify as regeneration points those instants at which a vehicle crosses the intersection in either direction, by noting at each such instant the following information :

- (i) the lane from which the 'initial' vehicle departs,
- (ii) if a queue exists, its lane and size,
- (iii) if the queue is in the same lane as (i), the type of the nearest opposing vehicle.
- (iv) if there is no queue, the type of the nearest opposing vehicle. The notation used is as follows :
- $R_i(n)$ refers to an instant at which a vehicle which may be of either type crosses in lane i, in the presence of a queue of n vehicles in the <u>other</u> lane, (n = 1, 2, ...);
- $S_{i}(n)$ refers to an instant at which a vehicle necessarily of type II crosses in lane i, in the presence of an additional n queueing $T_{i}(n)$ vehicles in lane, i, the nearest vehicle in the other lane being of type $\{I, (n = 1, 2, ...)\}$
- $\begin{array}{c} R_{i}(-1) \\ \text{ refers to instants at which a vehicle crosses in lane i, there being no queue present, and the nearest opposing vehicle being of type {I . II } \end{array}$

The remarks of section 3.3.1 concerning the validity of regeneration points are relevant here. State probabilities for this embedded Markov Chain are defined as follows :

$$\begin{array}{c} P_{i}(n) \\ c_{i}(n) \\ t_{i}(n) \\ r_{i}(n) \\ P_{i}(-1) \\ P_{i}(-2) \end{array}$$
 is the probability that a randomly
$$\begin{array}{c} B_{i}(n) \\ S_{i}(n) \\ T_{i}(n) \\ T_{i}(n) \\ B_{i}(-1) \\ B_{i}(-2) \end{array}$$

4.3.2. Transition Probabilities.

There are 100 types of transition such as $R_2(m) \rightarrow T_1(n), R_1(-1) \rightarrow S_2(n)$. Considerations of symmetry reduce the number for study to 50.

As an example we evaluate the conditional probability that a point of type $S_1(m)$ is followed by a point of type $R_1(-1)$. The sequences of events which would give rise to this situation are as follows (writing 'at x' for 'in (x, x+dx)').

(a)Of the m vehicles queueing in lane 1 immediately following the departure of the initial II from this lane, the first m-1 are of type I and therefore cross immediately. The remaining vehicle is of type II, its driver measures the interval until the arrival of the nearest opposing vehicle (of necessity of type I) as $\mathbb{Z}+\alpha$, and selects a gap-requirement time of length α , using the density $g_1(\alpha)$. It thereupon crosses after an interval α , during which there is no arrival in lane 1, causing an $R_1(-1)$.

- (b) All m+1 vehicles at the intersection in lane 1 cross immediately, i.e. they are all of type I except the first. The next arrival at the intersection is in lane 1 and is of type I.
- (c) All m+1 vehicles at the intersection in lane 1 cross immediately. The next arrival at the intersection is in lane 1, is of type II, and is at time τ (measured from the initial $S_1(m)$). This II selects a gap-requirement time of length α , using the density $f_1(\alpha)$. The next arrival in lane 2, of type I, is at time $\tau+\alpha+\tau_1$. No vehicles arrive in lane 1 during the interval α for which the II is stationary, and it is the departure of the II which causes the $R_1(-1)$.

The transition probability associated with $S_1(m) \rightarrow R_1(-1)$ is therefore

$$\frac{\lambda_1}{\lambda} p_1^{m+1} + q_1 p_1^{m-1} \int_0^\infty g_1(\alpha) e^{-\lambda \alpha} \partial \alpha + p_1^m q_1 \frac{\lambda_1}{\lambda} \int_0^\infty f_1(\alpha) e^{-\lambda \alpha} d\alpha.$$

The complete set of transition probabilities is listed in Appendix I. These may be checked by the consistency relations of section 2.3.3. Of the 50 types of transition, it is found that 23 have zero probability. The transitions $R_1(m) \rightarrow R_2(-2)$, $R_2(m) \rightarrow R_1(-2)$, $R_1(m) \rightarrow R_2(-1)$, $R_2(m) \rightarrow R_1(-1)$ are possible only when m = 1. 4.3.3. Determination of state probabilities.

We define

$$\widetilde{R}_{i}(x) = \sum_{n=1}^{\infty} x^{n} \rho_{i}(n), \quad \widetilde{S}_{i}(x) = \sum_{n=1}^{\infty} x^{n} \sigma_{i}(n), \quad \widetilde{T}_{i}(x) = \sum_{n=1}^{\infty} x^{n} \tau_{i}(n)$$

$$\|x\| \leq 1;$$
and
$$f_{i}^{\mathcal{X}}(s) = \int_{0}^{\infty} e^{-st} f_{i}(t) dt,$$

$$g_{i}^{\mathcal{X}}(s) = \int_{0}^{\infty} e^{-st} g_{i}(t) dt,$$

$$h_{i}^{\mathcal{X}}(s) = \int_{0}^{\infty} e^{-st} h_{i}(t) dt,$$

$$\int_{0}^{\infty} e^{-st} h_{i}(t) dt,$$

Then after some rearrangement the stationarity equations reduce to equations (1) to (5) below, together with a corresponding set $(1^{\frac{\pi}{2}})$ to $(5^{\frac{\pi}{2}})$ with interchanged lane suffices.

If we write $\S_2(\mathbf{x}) = 1 - g_2^{\mathbb{X}} (\lambda - \lambda_2 \mathbf{x}),$ $\eta_2(\mathbf{x}) = 1 - h_2^{\mathbb{X}} (\lambda - \lambda_2 \mathbf{x}),$ $\varphi_2(\mathbf{x}) = 1 - f_2^{\mathbb{X}} (\lambda - \lambda_2 \mathbf{x}),$

the equations are

-

$$\widetilde{R}_{1}(\mathbf{x})(\lambda-\lambda_{2}\mathbf{x}-\lambda_{1}\mathbf{p}_{1}\mathbb{F}_{2}(\mathbf{x})-\lambda_{1}\mathbf{q}_{1}\mathbb{T}_{2}(\mathbf{x})) - \widetilde{S}_{2}(\mathbf{x}) \frac{\lambda_{1}\mathbf{q}_{2}\mathbf{x}}{\mathbf{x}-\mathbf{p}_{2}} \mathbb{F}_{2}(\mathbf{x})$$

$$-\tilde{T}_{2}(x) \frac{\lambda_{1}q_{2}x}{x-p_{2}} \eta_{2}(x) - \Lambda_{1}(x), \qquad (1)$$

$$-\tilde{R}_{1}(x)\frac{p_{1}g_{2}^{\pi}(\lambda-\lambda_{2}x) + \tilde{S}_{2}(x)\left(1-\frac{q_{2}}{x-p_{2}}g_{2}^{\pi}(\lambda-\lambda_{2}x)\right) = B_{1}(x), \quad (2)$$

$$\begin{split} & -\overline{\mathbb{E}}_{\eta} \left(-\overline{\mathbb{E}}_{\eta} \left(x \right) \frac{q_{1}}{x} h_{2}^{\mathbb{E}} \left(\lambda - \lambda_{2} x \right) + \overline{\mathbb{T}}_{2} \left(x \right) \left(1 - \frac{q_{2}}{x - p_{2}} h_{2}^{\mathbb{E}} \left(\lambda - \lambda_{2} x \right) \right) = C_{\eta} \left(x \right), \end{split}$$

$$B_{1}(x) = -p_{1}\rho_{1}(1)g_{2}^{\Xi}(\lambda) + \tilde{S}_{2}(p_{2}) \left[\frac{\lambda_{2}q_{2}}{\lambda} (f_{2}^{\Xi}(\lambda-\lambda_{2}x)-f_{2}^{\Xi}(\lambda)) - \frac{q_{2}}{x-p}(\lambda-\lambda_{2}x) - \frac{q_{2}}{x-p}(\lambda)) \tilde{T}_{1}(p_{1}) + \frac{\lambda_{2}(f_{2}^{\Xi}(\lambda-\lambda_{2}x)-f_{2}^{\Xi}(\lambda))(\rho_{2}(-1)q_{2}+\rho_{1}(-2)p_{1}),}{\lambda} + \frac{\lambda_{2}(f_{2}^{\Xi}(\lambda-\lambda_{2}x)-f_{2}^{\Xi}(\lambda))(\rho_{2}(-1)q_{2}+\rho_{1}(-2)p_{1}),}{\lambda} + \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda-\lambda_{2}x) + \frac{q_{2}}{p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda-\lambda_{2}x) + \frac{q_{2}}{p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda-\lambda_{2}x) + \frac{q_{2}}{p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda-\lambda_{2}x) + \frac{q_{2}}{p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda-\lambda_{2}x) + \frac{q_{2}}{p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda) - \frac{q_{2}}{x-p_{2}}h_{2}^{\Xi}(\lambda)$$

76.

Equations (1), (2), (3) may be solved simultaneously to give $\tilde{R}_1(x)$, $\tilde{S}_2(x)$, $\tilde{T}_2(x)$ in terms of the constants $\tilde{S}_1(p_1)$, $\tilde{S}_2(p_2)$, $\tilde{T}_1(p_1)$, $\tilde{T}_2(p_2)$, $\rho_1(1)$, $\rho_2(1)$, $\rho_1(-1)$, $\rho_2(-1)$, $\rho_1(-2)$, $\rho_2(-2)$, which we denote by u_1 , \dots u_{10} respectively. In similar menner $\tilde{R}_2(x)$, $\tilde{S}_1(x)$, $\tilde{T}_1(x)$ are determined. We proceed to evaluate these constants.

If we write

$$J_{1}(\mathbf{x}) = \lambda - \lambda_{2} \mathbf{x} - \lambda_{1} p_{1} \frac{\kappa}{2} (\mathbf{x}) - \lambda_{1} q_{1} \frac{\eta}{2} (\mathbf{x}),$$

$$D_{1}(\mathbf{x}) = \mathbf{x} - p_{2} - q_{2} g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}),$$

$$E_{1}(\mathbf{x}) = \mathbf{x} - p_{2} - q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}),$$

equations (1) to (3) may be written

$$\begin{bmatrix} (\mathbf{x}-\mathbf{p}_{2})\mathbf{J}_{1}(\mathbf{x}) & -\lambda_{1}\mathbf{q}_{2}\mathbf{x}\tilde{\mathbf{g}}_{2}(\mathbf{x}) & -\lambda_{1}\mathbf{q}_{2}\mathbf{x}\tilde{\mathbf{q}}_{2}(\mathbf{x}) \\ -\frac{\mathbf{p}_{1}}{\mathbf{x}}(\mathbf{x}-\mathbf{p}_{2})\mathbf{g}_{2}^{\mathbf{H}}(\lambda-\lambda_{2}\mathbf{x}) & \mathbf{D}_{1}(\mathbf{x}) & \mathbf{0} \\ -\frac{\mathbf{q}_{1}}{\mathbf{x}}(\mathbf{x}-\mathbf{p}_{2})\mathbf{h}_{2}^{\mathbf{H}}(\lambda-\lambda_{2}\mathbf{x}) & \mathbf{0} & \mathbf{E}_{1}(\mathbf{x}) \end{bmatrix} \begin{bmatrix} \mathbf{\tilde{R}}_{1}(\mathbf{x}) \\ \mathbf{\tilde{s}}_{2}(\mathbf{x}) \\ \mathbf{\tilde{R}}_{2}(\mathbf{x}) \\ \mathbf{\tilde{R}}_{2}(\mathbf{x}) \end{bmatrix} \begin{bmatrix} \mathbf{\tilde{R}}_{1}(\mathbf{x}) \\ \mathbf{\tilde{R}}_{2$$

and if we denote by $\begin{bmatrix} r\\ ij \end{bmatrix}$ the matrix of minors of the left hand matrix,

$$\widetilde{R}_{1}(x) = \frac{A_{1}(x)\left[\frac{\pi}{11}(x) - B_{1}(x)\right]_{21}(x) + C_{1}(x)\left[\frac{\pi}{31}(x)\right]}{\bigtriangleup_{1}(x)}$$
(7)

where $\Delta_1(\mathbf{x}) = (1-\mathbf{x}) \left\{ (\lambda_1 + \lambda_2 q_2) q_2 g_2^{\mathbf{x}} (\lambda - \lambda_2 \mathbf{x}) h_2^{\mathbf{x}} (\lambda - \lambda_2 \mathbf{x}) - (\mathbf{x} - \mathbf{p}_2) \left[(q_1 \lambda_1 + q_2 \lambda_2) h_2^{\mathbf{x}} (\lambda - \lambda_2 \mathbf{x}) + (\mathbf{p}_1 \lambda_1 + q_2 \lambda_2) g_2^{\mathbf{x}} (\lambda - \lambda_2 \mathbf{x}) \right] + \lambda_2 (\mathbf{x} - \mathbf{p}_2)^2 \right\}.$

In order that $\lim_{x \to 1} \tilde{R}_{i}(x) \leq c_{0}$, the numerator of the expression $x \to 1$

(7) must vanish at x = 1. This leads to

$$A_1(1) + \lambda_1(B_1(1) + C_1(1)) = 0,$$

or equivalently

$$q_{2}\left(\frac{\lambda_{2}}{\lambda}\left(1-f_{2}^{*}(\lambda)\right)-\frac{1}{q_{2}}-\frac{g_{2}^{*}(\lambda)}{p_{2}}\right)u_{2}+p_{1}\frac{\lambda_{2}}{\lambda}\left(1-f_{2}^{*}(\lambda)\right)u_{3}$$

$$-q_{2}\left(\frac{1}{q_{2}}+\frac{h_{2}(\lambda)}{p_{2}}\right)u_{4}-\left(p_{1}g_{2}^{*}(\lambda)+q_{1}h_{2}^{*}(\lambda)\right)u_{5}$$

$$+\frac{\lambda_{2}}{\lambda}\left(1-f_{2}^{*}(\lambda)\right)\left(q_{2}u_{8}+p_{1}u_{9}\right)=0,$$
(8)

with a corresponding equation (8^{Ξ}) . The same equation (8) arises from similar consideration of $\tilde{S}_2(1)$, $\tilde{T}_2(1)$. Equations (4), (4^{Ξ}) , (5), (5^{Ξ}) , (8), (8^{Ξ}) are linearly dependent.

The conditions $\tilde{H}_{1}(0) = \tilde{S}_{2}(0) = \tilde{T}_{2}(0) = 0$ are all equivalent to $\lambda_{1}q_{2}\left(\frac{\lambda_{2}}{\lambda}\phi_{2}(0) + \frac{\tilde{S}_{2}(0)}{p_{2}}\right)u_{2} + \frac{\lambda_{1}\lambda_{2}p_{1}}{\lambda}\phi_{2}(0)u_{3} + \frac{\lambda_{1}q_{2}}{p_{2}}\eta_{2}(0)u_{4}$ $-J_{1}(0)u_{5} + \frac{\lambda_{1}\lambda_{2}}{\lambda}q_{2}\phi_{2}(0)u_{8} + \frac{\lambda_{1}\lambda_{2}}{\lambda}p_{1}\phi_{2}(0)u_{9} = 0,$ (9)

with a corresponding equation.

By definition, the functions $\tilde{R}_1(x)$, $\tilde{S}_2(x)$, $\tilde{T}_2(x)$ are bounded for $0 \le x \le 1$. Since $\Delta_1(x)$ may have one or more zeros in this interval, further conditions may have to be met by the numerators of these functions.

Let us suppose for the present that $\triangle_1(\mathbf{x})$ (and $\triangle_2(\mathbf{x})$) vanish exactly once for $0 < \mathbf{x} < 1$ at $\mathbf{x} = a_1, (a_2)$. We require the numerators of each of $\widetilde{\mathbf{H}}_1(a_1)$, $\widetilde{\mathbf{S}}_2(a_1)$, $\widetilde{\mathbf{T}}_2(a_1)$, expressed in the form of (7), to vanish. This results in a further three linear conditions on \mathbf{u}_1 , ..., \mathbf{u}_{10} . It has been found numerically that the first of these conditions yields no new information about \mathbf{u}_1 , ... \mathbf{u}_{10} . The second does yield a further linearly independent relationship, and the third is linearly dependent on the second and the previous conditions on \mathbf{u}_1 , ... \mathbf{u}_{10} . Although an algebraic demonstration of these remarks is desirable, the complexity of the resulting equations has been such as to resist all attempts to construct one.

Thus we have, writing $\theta_i(x) = f_i^{\Xi}(\lambda - \lambda_i x) - f_i^{\Xi}(\lambda)$,

$$u_{2} \cdot \lambda_{1} q_{1} q_{2} h_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) \left[b_{1} (\alpha_{1}) \left\{ \frac{\lambda_{2}}{\lambda} q_{2} (\alpha_{1}) - \frac{\lambda_{2} (\alpha_{1})}{\alpha_{1} - p_{2}} \right\} + \xi_{2} (\alpha_{1}) \left\{ \frac{\lambda_{2} q_{2}}{\lambda} \theta_{2} (\alpha_{1}) - \frac{q_{2}}{\alpha_{1} - p_{2}} g_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) - \frac{q_{2}}{p_{2}} g_{2}^{\mathbb{B}} (\lambda) \right\} + \left[p_{1} u_{3} + q_{2} u_{8} + p_{1} u_{9} \right] \frac{\lambda_{1} \lambda_{2}}{\lambda} q_{1} h_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1})$$

$$\left\{ b_{1} (\alpha_{1}) \phi_{2} (\alpha_{1}) + q_{2} \xi_{2} (\alpha_{1}) \theta_{2} (\alpha_{1}) \right\} - u_{4} \left[\frac{q_{1}}{\alpha_{1} - p_{2}} h_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) b_{1} (\alpha_{1}) \lambda_{1} q_{2} \gamma_{2} (\alpha_{1}) \right]$$

$$+ \left\{ J_{1} (\alpha_{1}) b_{1} (\alpha_{1}) - \lambda_{1} p_{1} q_{2} g_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) \xi_{2} (\alpha_{1}) \right\} \left\{ \frac{q_{2} h_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) b_{1} (\alpha_{1}) \lambda_{1} q_{2} \gamma_{2} (\alpha_{1}) \right\}$$

$$- q_{1} u_{5} \left\{ \lambda_{1} q_{2} \xi_{2} (\alpha_{1}) h_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) p_{1} g_{2}^{\mathbb{B}} (\lambda) + h_{2}^{\mathbb{B}} (\lambda) \left\{ J_{1} (\alpha_{1}) b_{1} (\alpha_{1}) - \lambda_{1} p_{1} q_{2} g_{2}^{\mathbb{B}} (\lambda - \lambda_{2} \alpha_{1}) \right\}$$

$$= 0, \qquad (10)$$

with a corresponding equation (10^{\pm}) , obtained from $T_1(a_2)$.

The normalising condition is

,

$$\tilde{R}_{1}(1)+\tilde{S}_{2}(1)+\tilde{T}_{2}(1)+\tilde{R}_{2}(1)+\tilde{S}_{1}(1)+\tilde{T}_{1}(1)+u_{7}+u_{8}+u_{9}+u_{10}=1.$$
(11)

By means of L'Hopital's rule we find that

$$\tilde{B}_{1}(1)+\tilde{S}_{2}(1)+\tilde{T}_{2}(1) = \frac{1}{\Delta_{1}(1)} \sum_{i=1}^{3} (A_{1}^{i}\Gamma_{1i}-B_{1}^{i}\Gamma_{2i}+C_{1}^{i}\Gamma_{3i}+A_{1}\Gamma_{1i}) -B_{1}\Gamma_{2i}+C_{1}\Gamma_{3i}),$$

where we write $A_1^{i} = A_1^{i}(1)$, $\Gamma_{21}^{i} = \Gamma_{21}^{i}(1)$ etc., for brewity, i.e.

$$\begin{split} \widetilde{\mathbb{R}}_{1}(1)+\widetilde{S}_{2}(1)+\widetilde{\mathbb{T}}_{2}(1) &= \frac{1}{\Delta_{1}^{4}(1)} \left\{ u_{2} \left[\frac{\lambda_{1}\lambda_{2}}{\lambda} q_{2}\varphi_{2}^{*}(1)+\frac{\lambda_{1}\widetilde{\mathbb{F}}_{2}(1)}{q_{2}} - \lambda_{1}\widetilde{\mathbb{F}}_{2}^{*}(1) \right\} \Sigma \Gamma_{11}^{*}(-1)^{1+1} \\ &+ \left\{ \frac{\lambda_{2}^{2}}{\lambda} q_{2}f_{2}^{*} \frac{\pi^{*}(\lambda_{1})}{q_{2}} - \frac{8\pi^{*}(\lambda_{1})}{q_{2}} - \lambda_{2}g_{2}^{*}(\lambda_{1}) \right\} \Sigma \Gamma_{21}^{*}(-1)^{1+1} \\ &+ \left[\frac{\lambda_{1}\lambda_{2}}{\lambda} q_{2}\varphi_{2}(1) - \lambda_{1}\widetilde{\mathbb{F}}_{2}(1) \right] \Sigma \left[\Gamma_{11}^{*} + \Gamma_{11}^{*}(\cdot)(-1)^{1+1} \\ &- \left[\frac{\lambda_{2}q_{2}}{\lambda} q_{2}(1) - g_{2}^{*}(\lambda_{1}) - \frac{q_{2}}{p_{2}} g_{2}^{*}(\lambda) \right] \Sigma \Gamma_{21}^{*}(-1)^{1+1} \\ &+ \left[p_{1}u_{3}+q_{2}u_{8}+p_{1}u_{9} \right] \left[\frac{\lambda_{1}\lambda_{2}}{\lambda} p_{1}\varphi_{2}^{*}(1)\Sigma \Gamma_{11}(-1)^{1+1} + \frac{\lambda_{2}^{2}}{\lambda} p_{1}f_{2}^{*}(\lambda_{1})\Sigma \Gamma_{21}^{*}(-1)^{1+1} \\ &+ \frac{\lambda_{1}\lambda_{2}}{\lambda} p_{1}\varphi_{2}(1)\Sigma \left[\Gamma_{11}^{*} + \Gamma_{11}^{*}(\cdot)(-1)^{1+1} - \frac{p_{1}\lambda_{2}}{\lambda} \theta_{2}(1)\Sigma \Gamma_{21}^{*}(-1)^{1+1} \\ &+ u_{4} \left[\lambda_{1} \left\{ \frac{\eta_{2}(1)}{q_{2}} - \eta_{2}^{*}(1) \right\} \Sigma \Gamma_{11}^{*}(-1)^{1+1} + \left\{ \frac{h_{2}^{*}(\lambda_{1})}{q_{2}} + \lambda_{2}h_{2}^{*}(\lambda_{1}) \right\} \Sigma \Gamma_{31}^{*}(-1)^{1+1} \\ &- \lambda_{1}\eta_{2}(1)\Sigma \left[\Gamma_{11}^{*} + \Gamma_{11}^{*}(\cdot)(-1)^{1+1} - \left\{ h_{2}^{*}(\lambda_{1}) + \frac{q_{2}}{p_{2}} h_{2}^{*}(\lambda) \right\} \Sigma \Gamma_{31}^{*}(-1)^{1+1} \\ &+ u_{5} \left[p_{1}g_{2}^{*}(\lambda)\Sigma \Gamma_{21}^{*}(-1)^{1+1} - q_{1}h_{2}^{*}(\lambda)\Sigma \Gamma_{31}^{*}(-1)^{1+1} \right] \right\} . \end{split}$$

We may now solve equations (4), (4^{Ξ}) , (5), (5^{Ξ}) , (9), (9^{Ξ}) , (10), (10^{Ξ}), (8), (11) for $\{u_{ij}\}$, in the case when $\Delta_{1}(x)$, $\Delta_{2}(x)$ each have a single zero in (0,1). From the resulting values, by means of equations (1), (2), (3), (1^X), (2^{Ξ}), (3^{Ξ}), the functions $\tilde{R}_{ij}(x)$, $\tilde{S}_{ij}(x)$, $\tilde{T}_{ij}(x)$ may be constructed, and it is found that the solutions in this case satisfy the requirements pertaining to a probability distribution with regard to positivity and boundedness by unity.

For some parameter values, however, it is found that $\triangle_{1}(x)$ has two zeros in (0,1), for at least one value of i. This may lead to the addition of further equations to (10), (10^{E}) , and it is found that by taking different pairs of these equations, together with the other eight linearly independent equations, differing solutions for $u_{1}, \ldots u_{10}$ are obtained which do not satisfy the basic requirements on these constants, such as positivity. We conclude that it is not possible to find a bounded solution to the stationarity equations in such a case. A similar conclusion would be expected in the case where more than two seros of either of $\triangle_{1}(x)$, $\triangle_{2}(x)$ lie in (0,1). There is also the possibility that either of $\triangle_{1}(x)$, i = 1, 2, does not have a zero in (0,1). In this case it appears impossible to derive a unique stationary distribution.

From this discussion it seems that a necessary and sufficient condition for ergodic behaviour of the system is that each of $\Delta_1(x)$, $\Delta_2(x)$ should have a single zero in (0, 1).

4.3.4. The distribution of delay.

As in section 3.3.4 we consider the distribution of delay to vehicles in lane 2, which may be deduced from the distribution of lengths of queues in lane 2 left by vehicles departing from that lane. Vehicles depart only at regeneration points, and vehicles in lane 2 only at points of the types $R_2(n)$, $S_2(n)$, $T_2(n)$, $R_2(-2)$, $R_2(-1)$. Points at which vehicles depart in lane 2, and at which a queue remains in this lane, are of the types $S_2(n)$, $T_2(n)$. Thus if $v_r = \text{prob} \left\{ a \text{ vehicle departing in lane 2 leaves a queue of length} \right\}$, r in lane 2

$$\Lambda_{2}(\mathbf{x}) = \sum_{\mathbf{r}=0}^{\infty} \mathbf{v}_{\mathbf{r}} \mathbf{x}^{\mathbf{r}}, \quad |\mathbf{x}| \leq 1,$$

and
$$\widetilde{V}_{i}(x) = \widetilde{S}_{i}(x) + \widetilde{T}_{i}(x)$$
,

then

$$v_{r} = \frac{1}{H_{2}} \begin{cases} 0 & n \\ p_{2} & \sum (\sigma_{2}(n) + \tau_{2}(n)) p_{2}^{k-1} \varepsilon (n-r) \varepsilon (r-n+k-1) \\ n=1 & k=1 \end{cases}$$

+
$$\sum_{n=1}^{\infty} \left(\frac{\sigma_2(n) + c_2(n)}{p_2} \right) \frac{\sigma_2(n-r)}{p_2(-r)}$$

+ $\left(\rho_2(-2) + \rho_2(-1) + \overline{R}_2(1) \right) \frac{\sigma_2(-r)}{p_2(-r)}$,

so that

$$\Lambda_{2}(\mathbf{x}) = \frac{1}{H_{2}} \left\{ \frac{P_{2} \tilde{V}_{2}(\mathbf{p}_{2}) - \mathbf{x} \tilde{V}_{2}(\mathbf{x})}{P_{2} - \mathbf{x}} + P_{2}(-2) + P_{2}(-1) + \tilde{H}_{2}(1) \right\} ,$$
where $H_{2} = \frac{1}{q_{2}} \left(\tilde{V}_{2}(1) - \tilde{V}_{2}(\mathbf{p}_{2}) \right) + \tilde{V}_{2}(\mathbf{p}_{2}) + \tilde{H}_{2}(1) + P_{2}(-2) + P_{2}(-1) \right)$

The Laplace transform of the delay distribution in lane 2 is $B_{2}^{\Xi}(s) = \Lambda_{2} \left(\frac{\lambda_{2-s}}{\lambda_{2}} \right) \text{ from which the first and second moments of the delay} \\ \text{ distribution may be derived as} \\ \mu_{1}^{*} = \frac{1}{H_{2}\lambda_{2}} \left\{ \frac{P_{2}}{q_{2}^{*}} \left[\tilde{V}_{2}(p_{2}) - \tilde{V}_{2}(1) \right] + \frac{1}{q_{2}} \tilde{V}_{2}^{*}(1) \right\}, \\ \text{ and } \mu_{2}^{*} = \frac{1}{\lambda_{2}^{2}q_{2}H_{2}} \left\{ \frac{2p_{2}}{q_{2}^{*}} \left[\tilde{V}_{2}(1) - \tilde{V}_{2}(p_{2}) \right] - \frac{2p_{2}}{q_{2}} \tilde{V}_{2}^{*}(1) + \tilde{V}_{2}^{*}(1) \right\}.$

4.3.5. The probability that a vehicle in lane 1 is not delayed.

This probability is of some interest in matching the model to data, chiefly because of its immediate interpretation in the practical situation. It is considered further in section 5.4.2.

In order that a vehicle in lane 1 should not be delayed it must cross the intersection at a regeneration point of type $R_1(-1)$, $R_1(-2)$ or $R_1(n)$. If it crosses at a point of type $R_1(n)$, then it has not been delayed. However it is possible that a vehicle of type II crossing at an $R_1(-1)$ or $R_1(-2)$ could be either the last vehicle in a queue in lane 1 or could (at an $R_1(-1)$) accept a gap but take a finite time to do so. (All type II vehicles faced with an opposing I are delayed).

The situations in which a vehicle experiences no delay are as follows. A vehicle which arrives in lane 1 at a situation other than these is necessarily delayed.

| Regeneration poin | nt at which vehicle | .Previous regeneration | | | | | |
|---------------------|---------------------|--|--|--|--|--|--|
| Crosses. | | point. | | | | | |
| Type of point. | Type of vehicle. | | | | | | |
| R ₁ (n) | any | any | | | | | |
| $R_{1}(-1)$ | I | any | | | | | |
| R ₁ (-2) | I | any | | | | | |
| R ₁ (-2) | II | $\begin{cases} R_1(-2) \\ R_2(-2) \\ T_1(k) \text{ with } k \text{ queueing type I's} \\ T_2(k) \text{ with } k \text{ queueing type I's} \end{cases}$ | | | | | |

These considerations lead to the following expression for the probability that a vehicle in lane 1 suffers no delay at the intersection. Prob. {no delay in lane 1}

$$= \frac{1}{H_{1}} \left\{ \tilde{R}_{1}(1) + P_{1}(-2) \left(p_{1} + \frac{\lambda_{1}}{\lambda} q_{1} \right) + p_{1} P_{1}(-1) + \frac{\lambda_{1}}{\lambda} q_{1} q_{2} P_{2}(-2) \right. \\ \left. + \frac{\lambda_{1}}{\lambda} q_{1} \left(\tilde{T}_{1}(p_{1}) + q_{2} T_{2}(p_{2}) \right) \right\},$$

where H_1 is defined in section 4.3.4.

4.3.6. Numerical results.

An ICT 1905 computer was programmed in Algol to calculate (a) any zeros of the denominators of $\tilde{R}_{1}(x)$; (b) the set of subsidiary constants u_{1}, \ldots, u_{10} ; (c) the mean and variance of the delay distribution and (d) the probability of no delay for each lane. These calculations were performed for the particular case in which each of the gap requirement time distributions possessed a density function of translated negative exponential form.

In the evaluation of (b), the quantities $\prod_{i,j} \prod_{i,j} required$ for equation (11) of section 4.3.3 were found by writing the left hand matrix of equation (6) of that section in the form $\left[\Lambda_{i,j}\right]$ where $\Lambda_{i,j} = \Lambda_{i,j}(x)$. $\Lambda_{i,j}(1)$, $\Lambda_{i,j}(1)$, $\prod_{i,j}(1)$, $\prod_{i,j}(1)$ may then be determined.

The calculation of (c) is simplified by writing

$$\Delta_{1}(x).\vec{v}_{2}(x) = \left[A_{1}(x)(\Gamma_{13}(x)-\Gamma_{12}(x))-B_{1}(x)(\Gamma_{23}(x)-\Gamma_{22}(x))+C_{1}(x)(\Gamma_{33}(x))-\Gamma_{32}(x))-\Gamma_{32}(x)\right]$$

$$\begin{split} & \text{in the form} \\ \vec{\nabla}_{2}(\mathbf{x}) = \left[\tilde{S}_{2}(\mathbf{p}_{2}) \left\{ \lambda_{1} \left[q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) - (\mathbf{x} - \mathbf{p}_{2}) (\mathbf{p}_{1} g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) + q_{1} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) \right) \right] \\ & + \lambda_{2} (\mathbf{x} - \mathbf{p}_{2} - q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) (\mathbf{x} - \mathbf{p}_{2} - q_{2} g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) - \frac{\lambda_{2} q_{2}}{\lambda} (\mathbf{x} - \mathbf{p}_{2}) (1 - f_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) \right] \\ & - \frac{\lambda_{2}^{2}}{\lambda} (\mathbf{x} - \mathbf{p}_{2}) (\mathbf{x} - \mathbf{p}_{2} - q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) (1 - f_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) (\mathbf{p}_{1}^{\mathbf{x}} (\mathbf{p}_{1}) + q_{2} \rho_{2} (-1) + \mathbf{p}_{1} \rho_{1} (-2)) \\ & + \tilde{T}_{2} (\mathbf{p}_{2}) \left\{ -\lambda_{1} \left[(\mathbf{x} - \mathbf{p}_{2}) (q_{1} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) + \mathbf{p}_{1} g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) - q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) \right] \\ & + (\mathbf{x} - \mathbf{p}_{2}) \lambda_{2} (\mathbf{x} - \mathbf{p}_{2} - q_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) - q_{2} \lambda_{2} h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) (\mathbf{x} - \mathbf{p}_{2} - q_{2} g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) \\ & - \frac{\lambda_{2} q_{2}^{2}}{p_{2}} h_{2}^{\mathbf{x}} (\lambda) (\mathbf{x} - \mathbf{p}_{2}) (h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) - g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) \right] \\ & - \frac{\lambda_{2} q_{2}^{2}}{p_{2}} h_{2}^{\mathbf{x}} (\lambda) (\mathbf{x} - \mathbf{p}_{2}) (h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) - g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) \right] \\ & + \rho_{1} (1) q_{1} q_{2} \lambda_{2} h_{2}^{\mathbf{x}} (\lambda) (\mathbf{x} - p_{2}) (g_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x}) - h_{2}^{\mathbf{x}} (\lambda - \lambda_{2} \mathbf{x})) \right] (1 - \mathbf{x}) / \Delta_{1} (\mathbf{x}), \end{split}$$

and cancelling (1-x) in numerator and denominator.

The results are listed in table 4.1.

Table 4.1.

Results of calculations for model III.

The distributions of gap-acceptance times are assumed to have translated exponential form. All parameters are assumed to be lane-symmetric.

| | Parameter values † | | | | | | Distn. of delay in lane i | | | | |
|----------------------------------|-----------------------------------|---|----------------|-----------------------|-----|-----------------------|---------------------------|-----------------------|---|---|--|
| LOM | Pi | λ _i | a _i | bi | °i | di | °i | ^k i | Mean | Variance | Pr. (no delay) |
| 1 2 3 4 5 6 7 | .5 | .2 .3 .5 .9 1.5 3.0 5.0 | .1 | 2 | 7 | 5 | .1 | 10 | .058 .071 .112 .338 3.283 | 1.17 .65 .44 1.15 42.15 | .684 .681 .674 .647 .551 |
| 8 9 10 11 | | .2 | | | | | 1 3 6 10 | | .291 .686 1.094 1.434 | 5.54 12.10 18.37 23.89 | .673 .661 .656 .654 |
| 12 | | | | | | | 1 | 20 | . 279 | 5.33 | .673 |
| 13 14 | .2 | •5 | | | | | .1 | 10 | .083 .057 | .14 1.26 | .782 .792 |
| 15 16 17 18 19 20 | .1 .3 .5 .7 .9 .95 | .3 | 3-5 | .7 | 3-5 | .7 | 3.5 | .7 | # # 49.843 3.317 1.633 | E T 7604.73 60.40 25.96 | # # • 507 • 679 • 802 |
| 21 22 23 24 | -5 | .1 | 3.5 | .6 .3 .1 .05 | 3.5 | .6 .3 .1 .05 | 3.5 | .6 .3 .1 .05 | 2.641 4.724 91.225 | 24,17 80.82 26021,44 | .633 .610 .515 |
| 25 26 27 | -5 | .3 | 3.5 | •3 •1 •05 | 3.5 | .3 .1 .05 | 3.5 | •3 •1 •05 | X X X | H H H | X F X |

These parameters are defined in chapter 4. Certain values are omitted for clarity - in such cases the relevant value is the nearest above in the same column.

x For these sets of parameter values there is not a unique zero of $\Delta_1(x)$ in (0,1).

Comments on table 4.1.

- (i) It appears that mean delay is critically related to λ_i , and is an increasing function of λ_i . For large λ_i , a unique zero of $\Delta_1(x)/(1-x)$ in 0 < x < 1 does not exist, and it may be inferred that the behaviour of the system is not ergodic. (Rows (1) - (7)).
- (11) Rows (13), (3), (14); and (15) (20) illustrate possible effects of variations of p_i on the delay distribution. For the situations of rows (13), (3), (14), the gap-acceptance distributions are such that the probability of a type II vehicle accepting a gap with which it is presented is quite high. Thus we expect the mean delay to be small for p_i near zero. On the other hand, in the situations of rows (15) - (20), the arrival rates and gap-requirement distributions are such that it is not probable that a turning vehicle accepts a random gap with which it is presented. Since all queueing type II vehicles make use of a gap-requirement distribution,

we might expect congestion to increase in this case as p_i tends to zero. This supposition is confirmed by the results of rows (15) - (20): it appears that the increase in congestion is so great that the behaviour of the model is not ergodic for the smaller values of p_i .

- (iii) The effect of changes in the gap-acceptance parameters b_i, d_i, k_i is shown in rows (21) - (24). Decreasing these parameters has the effect of decreasing the probability that a type II vehicle accepts a gap with which it is presented. As expected, the mean delay is found to increase substantially as b_i, d_i, k_i are decreased, to the extent that for small values of these parameters, ergodic behaviour of the system is not possible.
- (iv) The behaviour of the prob. {no delay} for variations in the parameters is consistent with that of the mean delay.
- (v) It would be interesting to investigate the circumstances in which the change from ergodic to non-ergodic behaviour of the system occurs. Although an analytic investigation of this does not seem feasible, it is suggested by the data in table 4.1 that broadly speaking the system is not ergodic when it is overloaded. It seems that, since ergodicity is dependent on the parameters through the functions $\Delta_i(x)$, it is not affected by the form of the functions $f_i(x)$. This situation might have been expected from intuitive considerations.

4.4. Remarks on Model III.

Model III should deal with some of the criticisms of model II made in section 3.4. In particular the 'follow-on' behaviour of vehicles has been made sufficiently realistic to introduce non-ergodic behaviour for certain parameter values.

It is possible to select various aspects of the model which may not at times be in agreement with observation. For instance, the release of a queue consisting entirely of type I vehicles does not occur instantaneously in practice and some delay may be caused by these vehicles to vehicles in the other lane. Some drivers may base their decisions not merely on the nearest opposing vehicle but also on the vehicle following it. In some situations courteous type I drivers may give way to opposing type II's. In general, however, the occurrence of such events is comparatively infrequent, and model III may well give a reasonable overall picture of behaviour at the intersection.

Other aspects of model III which could represent a more serious lack of conformity with the actual situation are that the postulated gap acceptance behaviour may be unrealistic, as discussed in section 3.1, that the arrival process may not be Poisson, and that the classification of vehicles into types I and II may not be valid. Some of these points are further discussed in section 5.4 in the light of the data.

4.5. 'Impatient' queueing type I vehicles : model III(a).

Observation on actual intersections suggests that model III could be made more generally applicable by allowing for the tendency of some type I vehicles, which in model III would join the end of a queue, to pass alongside the queue and make a crossing without delay. This would lead effectively to consideration of the intersection of a four lane major road with a minor road.

In an attempt to fit model III to an intersection at which the proportion of these 'impatient' type I's is small, observed queues could be slightly modified by the inclusion of hypothetical vehicles corresponding to these type I's, provided they did not interfere with turning traffic.

Another approach to this problem would be to medify model III to simulate this type of vehicle behaviour, again under the assumption that the 'impatient' type I vehicles do not interfere with turning traffic. (If this assumption is not applicable there is a possibility of two simultaneous queues.) The modification to medel III depends on its characteristic that, as soon as a type I vehicle has joined a queue, it ceases to exist, at least so far as any further influence it has on the behaviour of the system is concerned. Its presence is of course noted by the previous analysis of model III.

Two ways in which model III could be modified are the following: (i) The analysis of the model is as above up to section 4.3.4. At this point we suppose that a type I vehicle may leave a queue immediately upon its arrival with probability s₁ for lane i. The analysis of section 4.3.4 is modified to find the distribution of delay to vehicles which do not leave a queue upon arrival : from this the distribution of delay to all vehicles in a given lane is constructed.

(ii) Alternatively, an earlier modification to the model would arrange the input of type I vehicles to queues at a different rate to that obtaining in the free flow situation. This would lead to a modification of the transition probabilities.

Chapter 5.

Statistical Analysis of Observations.

5.1. Introduction.

In any situation in which an attempt is made to analyse mathematically a practical problem, the success of the analysis depends on the extent to which its results agree with actual observations. In the present case we have three models for an intersection, each of which is a considerable idealisation of the actual situation. It is considered unlikely that close agreement between observed and predicted values for models I and II would be obtained, but that model III may perhaps be flexible and realistic enough to justify the effort involved in assessing its results against observation. It seems highly desirable to assess the progress of the work so far, before launching on the construction of further models.

This chapter briefly describes the observations which were made, the statistical processes by which the characteristics of model III were compared with them, and any conclusions regarding the value of the model which seemed to be justified. There is a brief reference to model I in section 5.4.3, for purposes of comparison.

5.2. The observations.

In Appendix II an account is given of the method used to obtain the observations, and of certain difficulties which were encountered in their interpretation. Here it is enough to note that the observations, which refer to only one intersection, consist of a series of pictures of the situation at the intersection, taken at a constant interval of approximately one second. From the filmed record it is possible to observe queue-sizes at any instant and to measure delays experienced by vehicles. These measurements are subject to the various kinds of experimental error discussed in Appendix II.

5.3. Estimation problems.

5.3.1. Estimation of p_i, λ_i .

The estimation of these parameters is comparatively simple. Each refers to a characteristic of the model which has a well defined counterpart in the actual situation. Under the assumptions of a Poisson input of vehicles of randomly assorted type in each lane, appropriate estimates consist, respectively, of the sample proportion of type I vehicles, and the observed rate of vehicle arrivals, in each lane.

These estimates could be modified, by the method to be described in section 5.3.3, in order to secure a better fit of the model to the data. Unless the modified estimates were found not to differ significantly from the original estimates, their adoption would be inconsistent with the natural interpretation of the parameters. Consequently, subject to the adherence of the observed input processes to the assumptions of the model (section 5.4.1), the intuitive estimates were accepted as final. Their values are as follows.

| | | Pį | λ_1 (vehicles/second) |
|------|---|---------|-------------------------------|
| iane | 1 | . 63855 | . 19280 |
| Lane | 2 | .96988 | . 20441 |

5.3.2. Estimation of other parameters in model III.

For the description of model III, details of the functions h_i , g_i , f_i are necessary. In the model these functions describe both the driver's estimate of his gap-requirement time and the time required by him to complete his crossing. As it is very likely that these intervals will differ in the actual situation, the practical interpretation of the functions h_i , etc., is not obvious. It is apparent also that direct information concerning the estimates formed by drivers in these circumstances will not be easily available, and it is therefore likely that the interpretation given to the gaprequirement distributions will be influenced by the types of relevant observations possible in practice.

The following are quantities which are relevant to gap-acceptance behaviour and upon which measurements may be made.

(a) The sizes of gaps which are accepted or rejected by type II vehicles. (b) The intervals actually used by turning vehicles to complete their manoeuvros.

From the film it is found that measurements of the intervals (b) usually have values of less than three secs. In view of the practical difficulties in making these measurements, and of the frame separation of about one second, such data are not considered to contain useful information.

Measurements of (a) are available only for the situations which in the model would require the use of g_1 or f_1 . They are listed in table 5.1. We observe that gap sizes of interest are in the range 4-7 secs. for g_1 , 3-5 secs. for f_1 , and that the frame separation of approximately one second is not small enough to permit any but the crudest inferences about g_1 and f_1 to be made, even ignoring the considerable subjective judgement involved in constructing these data.

Table 5.1.

Gaps accepted and rejected by type II vehicles.

| (i) | Stationary type II | I in lane | 1 wi | th a | pproa | ohing | type | I in | lane | 2(g ₁). |
|------|----------------------|-----------|------|------|-------|-------|------|------|------|---------------------|
| Gap | size (secs.) | 2 | 3 | 4 | 5 | 6 | 7 | 8 | 9 | |
| Proj | portion of gaps acce | pted.0 | 0 | 0 | . 6 | -75 | 1 | 1 | 1 | |
| Numb | er of gaps observed | 1. 24 | 8 | 3 | 5 | 4 | 1 | 1 | 1 | |

| (11) | Noving | type | II in | land | 9 1 | with | appro | baching | type | I | in | lane | 2 | (f ₁). |
|-------|----------|--------|-------|-------|-----|------|-------|---------|------|---|----|------|---|--------------------|
| Gap s | 120. | | | | 1 | 2 | 3 | 4 | 5 | 6 | | 7 | | |
| Propo | rtion of | gaps | | pted. | .0 | 0 | 0 | . 33 | 1 | 1 | | 1 | | |
| Numbe | r of ga | obe ag | erved | • | 3 | 9 | 4 | 3 | 1 | 5 | | 2 | | |

The initial approximations to g_1, f_1 , are therefore based on table 5.1., and on the work of others in this field, who have suggested several possible forms for these functions. (See e.g. Cohen, Dearnaley and Hansel (1955) who proposed the function erf (log t-A)/B for the probability that a driver accepts a gap of size t.) Herman and Weiss (1961) suggest that, from data obtained in a controlled experiment, the form of the functions h_i, g_i, f_i may be approximated by a translated exponential distribution, and in view of the computational advantages of this distribution, it was decided to adopt it here. Specifically, it is assumed that

$$h_{i}(t) = 0, t = a_{i};$$

$$= b_{i}exp(-b_{i}(t-a_{i})), t > a_{i};$$

$$g_{i}(t) = 0, t = c_{i};$$

$$= d_{i}exp(-d_{i}(t-c_{i})), t > c_{i};$$

$$f_{i}(t) = 0, t = e_{i};$$

$$= k_{i}exp(-k_{i}(t-e_{i})), t > e_{i}.$$

The problem is now reduced to one of estimating the 12 parameters a_i, b_i, c_i, d_i, e_i, k_i. By comparing a plot of the logarithm of the proportion of gaps rejected against size of gap for the data of table 5.1 with the logarithm of the survivor function of the appropriate gap-requirement distribution we obtain the following estimates.

| Parameter | °1 | a ₁ | •1 | k.1 |
|-----------|--------|----------------|-----|-----|
| Estimate | 3.5 | .45 | 3.0 | .50 |

Although it would be possible to apply more sophisticated methods to the data of table 5.1, such as the maximum likelihood method, it is considered that, because of the unreliability of the data, little would be gained.

5.3.3. Improvements on the crude estimates of the gap-acceptance parameters.

Since, in the model, the parameters considered in the previous section, together with p_i , λ_i , determine the entire probabilistic character of model III, it is likely that inferences about these parameters could be based on aspects of the behaviour of an actual intersection which are less immediately concerned with gap-acceptance. One could, for instance, consider the construction of maximum likelihood estimates of the gap-acceptance parameters, but it is found that the entire history of the system is too complex for the construction of a likelihood function to be practicable.

In the lack of a likelihood function for the complete set of observations, we might consider basing estimates on a likelihood function for a limited set of observations, which might nevertheless contain sufficient information about the parameters to be useful. Here we make use of observations on the embedded Markov Chain, and observation of the system is confined to the sequence of regeneration points defined in §4.3.1. The resulting data consist only of the lengths and types of queues at these regeneration points and are available with reasonable accuracy. A further advantage of this procedure is that observations on variables in the model, whose interpretation in the actual situation is uncertain, are not required.

The likelihood function for such observations may be constructed as follows. Transition intervals are first classified according to the types of the regeneration points defining the interval. Thus we have, e.g.,

| Transition | classification number (θ) ($\theta = 1, \dots, 54$). | |
|------------------------------|---|--|
| $R_1(-2) \rightarrow R_1(n)$ | 1 | |
| $R_2(-2) \rightarrow R_2(n)$ | 2 | |
| • | | |
| | | |
| • | • | |
| | - · · · · · · · · · · · · · · · · · · · | |
| $R_1(n) \rightarrow R_1(n)$ | 9 | |
| • | - Plant and a second second | |
| | | |
| | | |

We denote by $n_{\theta}(i, j)$ the number of observed transitions of type θ , where i and j may be used to denote relevant queue lengths: for some transitions one or both are not required. $p_{\theta}(i, j)$ denotes the corresponding transition probability, and the logarithm of the likelihood (conditional on the types of the regeneration points beginning the sequences of observations - see Appendix II.) is taken
28

54

$$\Sigma$$
 Σ Σ $n_{\theta}(i, j) \log p_{\theta}(i, j).$
 $\theta=1$ j
where relevant

 $p_{\theta}(i, j)$ must be completely evaluated for the particular functions h_i, g_i, f_i assumed: for the assumptions of section 5.3.2, we have, for example,

$$p_{9}(m, n) = \lambda_{1}\lambda_{2}^{n-m} \left[\frac{p_{1}e^{d_{2}c_{2}}}{(d_{2}+\lambda)^{n-m+1}} \prod^{n} ((d_{2}+\lambda)c_{2}, n-m+1) + \frac{q_{1}e^{a_{2}b_{2}}}{(d_{2}+\lambda)^{n-m+1}} \prod^{n} ((b_{2}+\lambda)a_{2}, n-m+1) \right] \in (n-m),$$

where $\Gamma(x, n) = \int_{x}^{\infty} e^{-t} t^{n-1} dt/(n-1)!_{a}$

A procedure was developed to compute the value of the loglikelihood for a given set of parameter values and a given vector of observed transitions. This was incorporated into a standard program for maximising a function by the steepest-descent method (E.J.Wasscher, 1963).

The results of the calculations carried out are noted in table 5.2, and detailed comments are made in the accompanying notes. The following considerations are relevant here.

(1) It was considered desirable to make some attempt to investigate the multiplicity of maxima of the likelihood function, and to run the program from a variety of starting points in the parameter space. Since very long runs were required to reach a maximum with respect to the whole set of gap-acceptance parameters, the maximisation was usually carried out with respect to a smaller number of variable parameters.

(ii) It seemed a possibility that the likelihood function would contain comparatively little information about some of the parameters under investigation. It was to be expected, therefore, that its use might lead to spurious estimates of these parameters. Certain methods were used to try to prevent this.

More specifically, it was observed that there were relatively few type II vehicles which arrived in lane 2 during the period of observation; in fact there were 5 from a total of 176 vehicles. It would be reasonable to expect the information available about a parameter such as e_2 , specifying the probability that a II in lane 2 accepts a gap, to be slight, and that this might be reflected in the likelihood function by the absence of factors which are functions of e_2 . However if we consider the total of 255 transitions extracted from film, it is found that 42 are of types whose transitions probabilities are functions of e_2 . A study of the 42 transitions reveals that 37 are of the type $R_2(-1) \Rightarrow R_2(-1)$ with associated probability $\lambda_2 \left\{ p_2 + (q_2 k_2 e^{-\lambda e_2}/(\lambda + k_2)) \right\}/\lambda$, a decreasing function

of e_2 , and we accordingly expect this function to have a considerable influence on the value of the maximum likelihood estimate of e_2 .

It may seem surprising that e, should occur in all 42 times in the interpretation of data for which it is known that the total number of II's observed in lane 2 is 5. The explanation is that the transition probabilities must allow for the fact that the type of the vehicle in lane 2, which defines the end of the transition interval, is not specified by the description of the interval used, and may be type II. In fact for the data such a vehicle very rarely is of type XI. Thus it would appear very likely that estimates obtained from the maximisation of this likelihood function are fundamentally affected by the inadequacy of the way in which the system is described at some regeneration points. This disadvantage of the estimation process is of course magnified by the particular set of data which are available, and in which so large a proportion of transitions are among regeneration points where no queue is present.

(111) In view of these doubts concerning the validity of the estimation process based on this particular likelihood function, it was considered prudent to make some assessment of the realism of each of the sets of estimated parameter values. This took two

forms.

- (a) The mean and variance of the distribution of delay in each lane of model III, together with the probability of no delay, were computed using the programs of §4.4. The resulting values were compared with the corresponding sample statistics. (At this stage a visual comparison was used a more detailed comparison is discussed in §5.4.2.)
- (b) Although the data of table 5.1 do not provide sufficient information to estimate the parameters with acceptable precision, they do suggest approximate bounds for functions which might be considered as gap-acceptance distribution functions. From inspection of the data we might reasonably require such a function to satisfy the following conditions.
 - (1) Prob. a driver accepts a gap of 2 seconds ~ 0 (2) Prob. a driver accepts a gap of 9 seconds ~ 1
 - (3) $\int_0^x h_1(t) dt < \int_0^x f_1(t) dt$, 2 < x < 9.

Table 5.2.

Maximum Likelthood estimates of parameters

| Restrictions | Parameter valu | tes at maximum | 11 and 1 | | 1 | | | | Constant in | |
|--------------|--|---|----------|-------|-----------|-----------|-------|---------|-------------|------|
| uu | | | HA HOW | | ARTE ARTE | TA DO T.7 | 0 800 | 111TT T | TADOW DA | |
| meximisation | a, b, o, d, e, k, | *2 b2 c2 d2 e2 k2 | | Lail | Re 1 | | | Lai | 1e 2 | |
| (see notes) | | | usen. | YBZ. | pr no de | 1ay | mean | TBA | pr no d | elay |
| (a) | 3.2 4.5 1.0 ^m 0.1 ^m 1.0 ^m 0.1 ^m | 4.0 5.0 4.0 5.0 3.0 10.0 | | | | | | | | - |
| (P) | 3.5 .178 3.5 .179 3.0 .054 | 3.5 .45 3.5 .45 3.0 .50 | | | | | | | | |
| (°) | 1.152 0.185 .001 [#] .140 .001 [#] .071 | .062 .252 .001 [#] .987 1.341 .001 [#] | 3.09 | 20.9 | . 38 | | .41 | 15.8 | .94 | |
| (q) | .038 .191 .001 ^E .155 .001 ^E .058 | .038 .191 .001 [#] .155 .001 [#] .058 | 9.16 | 139.4 | .27 | | . 25 | 10.5 | 96. | |
| (•) | .001 ^m .156 .001 ^m .156 .001 ^m .058 | .001 [#] .156 .001 [#] .156 .001 [#] .058 | 9.10 | 1.721 | .27 | | .27 | 4.01 | .96 | |
| + | | | 5.05 | 44.9 | .42 | | .432 | | .94 | |
| | | | | | | | | | | |

"Value at boundary of parameter range.

MM Moments calculated only for those estimates of parameters which seemed reasonably lane-symmetrics.

Corresponding sample statistics.

Notes on table 5.2.

<u>Maximisation (a).</u> The parameters varied for these first runs were a_1 , b_1 , c_1 , d_1 , e_1 , k_1 , as it was expected that there would be little information relevant to the others in the likelihood function for the data. Several sets of starting values were chosen, a typical set being, respectively, 4, 5, 4, 5, 3, 10, lane symmetry being assumed. The results for these values are shown. It is necessary to impose arbitrary positive lower limits on the range of variation of the parameters (to avoid log 0), and it will be observed that several of the estimates listed have values at boundary points of the range of parameter variation. As it was expected that any realistic estimates of these parameters would possess a degree of lane-symmetry, determination of moments of the delay distributions for these values was not considered worthwhile.

<u>Maximisation (b).</u> The lower bounds for the parameters used in (a) were decreased. In view of consideration (ii) above, values of the parameters a_1 , c_1 , e_1 (the translation parameters), were fixed, and the maximisation was carried out with respect to b_1 , d_1 , k_1 , from starting values given by the graphical method of section 5.3.2, it being assumed initially that gap-acceptance parameter values are independent of lane, and that $a_1 = c_1$, $b_1 = d_1$. <u>Maximisation (c)</u>. In this all twelve parameters a_1 , ..., k_1 were varied. Starting values were chosen as for (b). <u>Maximisation (d).</u> Consideration (ii) above suggests it is unlikely that useful estimates of the parameters for gap-acceptance by lane 2 vehicles would be obtained from the likelihood function. The physical characteristics of the intersection were such that one would expect the gap-acceptance parameters to be nearly lane-symmetric. The program was arranged to maximise with respect to the variables a, b, c, d, e, k, where $a = a_1 = a_2$ etc.

<u>Maximisation (e)</u>. The scarcity of information in the likelihood function about a_i , b_i , suggests that as a first approximation we consider the behaviour of a II, stationary at the intersection, to be independent of the type of the approaching vehicle. Thus we maximise as in (d), subject to the further restriction that $a_i = c_i$, $b_i = d_i$, i = 1, 2.

If a provisional assessment of the results quoted in table 5.2 is made along the lines indicated in (iii) (b) above, it is found that none of the estimates of rows (a) to (e) may be regarded as acceptable. The question of the acceptability of these estimates according to other criteria is discussed in section 5.4.

In view of the known deficiencies of the likelihood function, it was decided to make ad hoc modifications to the maximum likelihood estimates, in order to bring them into conformity with requirements (iii) (b). Several sets of parameter values were produced in this

way, and the corresponding moments of the delay distribution were calculated : the results are shown in table 5.3.

Table 5.3.

Modified M.L. estimates of parameters

| | Est | imate | e of | para | amete | er ** | | Lar | 1e 1 | | Lar | ne 2 |
|-----|----------------|----------------|------|----------------|----------------|----------------|-------|-------|--------------|-------|------|---------------|
| bet | a _i | b _i | ci | ^d i | e _i | ^k i | mean | var | pr{no delay} | mean | var | pr {no delay} |
| 1 | 1.0 | . 20 | 0.0 | .20 | •5 | .05 | 6.52 | 70.9 | - 30 | .24 | 19.9 | .96 |
| 2 | 2.5 | . 30 | 2.5 | . 30 | 1.5 | .40 | 13.10 | 278.9 | .22 | . 187 | 2.4 | . 97 |
| 3 | 1.5 | . 30 | 1.5 | . 30 | 1.5 | .40 | 5.55 | 64.89 | . 38 | .15 | 1.44 | -96 |
| 4 | 0 | . 30 | 0 | . 30 | 0 | .40 | 1.80 | 11.5 | . 50 | .07 | .48 | .97 |
| 5 | 0 | .20 | D | .20 | 0 | . 30 | 3.60 | 40.2 | .41 | .12 | 1.16 | .96 |
| x | | | 3 | í. | | | 5.05 | 44.9 | .42 | .432 | | -94 |

(Values of p_i , λ_i as before)

* Corresponding sample statistics.

Lane symmetry assumed.

5.4. The assessment of goodness-of-fit.

In any attempt to assess the adequacy of model III as a model for the behaviour of actual traffic, certain basic questions must be considered. The model is based on a number of assumptions concerning the behaviour of individual vehicles - how realistic are these assumptions and do they reflect sufficiently those aspects of this behaviour which are most relevant to a study of congestion? From these basic assumptions certain deductions are made about the behaviour of traffic. To what extent are these deductions consistent with the behaviour of actual traffic?

Perfect agreement between model and reality is not expected. One of the purposes of building models, such as those of this thesis, is to provide a simple explanation of complex situations which is nevertheless sufficiently realistic to provide approximate agreement with observations on the actual system. We are now trying to assess whether model III gives a sufficiently realistic picture of an actual intersection to make further development of the model unnecessary.

The plausibility of some aspects of model III has been discussed in §4.5 and we now reconsider the predictions of the model in the light of the data.

5.4.1. The distribution of arrivals.

With the purpose of comparing observed arrivals of vehicles in





each lane with the postulated Poisson process, frequency distributions for the inter-arrival intervals (shown in Appendix III) were constructed. In this context, an arrival is defined as the passage of a vehicle past a fixed point chosen as far distant from the intersection as possible, so as to reduce to a minimum the effects of congestion on the arrival distribution. Intervals between arrivals are listed in Appendix III. The separation of these intervals into groups, corresponding to continuous periods of observation, is ignored, and the sequence of inter-arrival instants resulting for each lane is examined in order to assess its conformity with a Poisson arrival process.

The log survivor function for the inter-arrival intervals is plotted (fig.5.1) for arrivals in each lane. It will be observed that this is roughly linear as would be expected for a Poisson input. In order to assess the significance of the deviations from linearity the modified mean test of Lewis (1965) based on the statistic S' is applied to the intervals of each lane. In each case the value of S' is far from significant. (In lane 1, S' = 63.30, and the distribution of S' under the null hypothesis is normal with mean 62.5, variance 13.75; in lane 2, S' = 73.70 (mean 72.5, variance 12)). Although as Lewis points out the S' test should detect serially correlated alternatives, it was though useful to determine the correlation coefficient of lag one for the intervals of lanes 1 and 2. These were respectively .00645 and .00436, and are far from significant

5

if assessed by the criterion of Cox and Lewis (1966), p. 165.

It appears that on the data available the arrival process in each lane conforms with a Poisson process. A more detailed assessment of the adequacy of the Poisson model for arrivals seems inappropriate here partly because of the weakness of the techniques available for assessing goodness-of-fit of predictions resulting from the model, and the inaccuracies of the data, but also because it is to be hoped that the applicability of model III would not be too fundamentally affected by slight deviations of the input processes from complete randomness.

5.4.2. Realism of predictions from model III.

1 4 4 1 4 1

We consider next the comparison of derived aspects of model III with the data, in particular the comparison with sample values of certain predicted parameters of the delay distributions.

Means and variances of the delay distributions, both theoretical and empirical, are shown in tables 5.2, 5.3. The remarks of Appendix II concerning difficulties in measuring delay times are relevant here, and we note that an average bias of, say, 2 seconds in lane 1, arising from misinterpretation of 'delay', is not improbable. We would expect that biases for lange 1 and 2 would differ because of greater difficulty in assessing the positions of vehicles in lane 2.

Differences in the values of the sample and predicted means

The second second second Startings

might therefore be accounted for as due to a combination of four factors: laok of realism in the model, biased interpretation of 'delay', imprecision of parameter estimates in the fitted model, and sampling fluctuations of mean delay about its expected value. If the bias in observations of delay for a given lane is roughly constant, the variance of the delay distribution should be almost unaffected by it, and might provide a better statistic than the sample mean upon which to base a comparison of data with the model.

It is known that the estimate of the variance of the delay distribution calculated by regarding the observed sequence of delays as a random sample will be influenced to an unknown extent by the autocorrelations of the observations. Jowett (1955) has proposed a method of obtaining approximately unbiased estimates of the population variance and of the sampling variance of the mean from a section of a stationary time series, and it seems that this technique might be useful here. We denote the observed sequence of delays, in order, by x_i , i = 1, ..., n. The sth serial variation parameter, b_g , may be defined by

$$\delta_{g} = E\left\{\frac{1}{2}(x_{n} - x_{n+g})^{2}\right\} = \sigma^{2}(1 - \rho_{g}),$$

where σ^2 is the variance of the marginal delay distribution and ρ_B the serial correlation coefficient of lag s of the joint delay distribution in lane 1. Jowett suggests that it should often be possible to find an so such that, for some suitable small c, $\left|b_B - \sigma^2\right| < z$.

in which case we may write $E\left\{\frac{1}{2}(x_i - x_j)^2\right\} = \sigma^2 + \theta_8$ where $|\theta| < 1$, provided that $|i - j| \ge s_0$. His estimate of σ^2 is the average of all semi-squared differences which are provided by the sample, i.e.

$$\begin{split} \mathfrak{G}_{\mathbf{g}_{0}}^{\mathbf{2}\mathbf{X}} &= \left\{ \begin{array}{c} \Sigma & \frac{1}{2}(\mathbf{x}_{1} - \mathbf{x}_{j})^{2} \\ \mathbf{i}, \mathbf{j}: |\mathbf{i} - \mathbf{j}| \geq \mathbf{s}_{0} \end{array} \right\} / \left\{ \begin{array}{c} \Sigma & \mathbf{1} \\ \mathbf{i}, \mathbf{j}: |\mathbf{i} - \mathbf{j}| \geq \mathbf{s}_{0} \end{array} \right\} \\ &= \left\{ \begin{array}{c} \mathbf{n} \\ \mathbf{n} \\ \mathbf{\Sigma} \\ \mathbf{i} = \mathbf{1} \end{array} \right\} - \left\{ \begin{array}{c} \mathbf{S}_{0}^{-1} \\ \Sigma \\ \mathbf{n} \end{array} \right\} / (\mathbf{n} - \mathbf{s}_{0})(\mathbf{n} - \mathbf{s}_{0} + 1), \\ \mathbf{n} - \mathbf{s} \end{array} \right\} \\ \mathbf{n} - \mathbf{s} \end{split}$$

where

Ss 2 has a bias of at most g.

 $D_{B} = \sum_{i=1}^{2} (x_{i} - x_{i+3})^{2}$

He suggests that, in practice, as so is unknown, a value for so may be selected by inspection of sample estimates of b given by $d_{a} = \frac{1}{2}D/(n-s)$.

For the data under discussion a delicate assessment of the most appropriate value for s is not necessary, since, for purposes of comparison with the theoretical model, $\sigma_{s}^{2^{R}}$ is effectively constant, save for sampling fluctuations, for s > 6.

| 10 | 12 | | |
|-----|----|------|---|
| - 8 | | ~ | - |
| | | 6.00 | |
| | | | - |

| Tabl | A | 5.4. |
|------|---|------|
| | | |

| Analysia | of Semi-squared differen | ces of Delays. |
|----------------|--------------------------|----------------|
| ⁸ 0 | d.so | Geogram |
| 0 | 0.0 | 43.70 |
| 1 | 15.79 | 43.97 |
| 2 | 27.65 | 44.32 |
| 3 | 3 3.7 9 | 44.53 |
| 4 | 36.99 | 44.67 |
| 5 | 37.24 | 44.76 |
| 6 | 40.70 | 44.86 |
| 7 | 45.12 | 44.91 |
| 8 | 47.03 | 44.91 |
| 9 | 45-53 | 44.88 |
| 10 | 42.45 | 44.80 |

From Table 5.4, the estimate of 6^2 for lane 1 has a value of about 44.9.

Observations of delay experienced by vehicles in lane 2 are toofew to permit a sample estimate of the variance of the delays in that lane to be made.

A comparison of the variance for lane 1 derived in this way with the predicted variances of Tables 5.2 and 5.3 reveals that the entries of rows (c), 1, 3, 5 appear consistent with the sample value.

Jowett shows in addition that an approximately unbiased estimate of the sampling variance of the mean of a series of correlated observations is given by $\sigma_{B_0}^{2^*} - v$, where $v = \frac{1}{2n^2} \sum_{i,j=1}^{n} (x_i - x_j)^2$. Thus for lane 1 the estimated standard deviation of the mean observed delay is $\sqrt{44.9 - 43.7} = 1.10$. This assists in the comparison of the sample mean with the predicted means: those of rows (1), (3), (5), (Table 5.3) are within about 1.5 estimated standard deviations of the sample mean, whereas the mean of row (c) (Table 5.2) is distant almost two standard deviations.

Inspection of the sequence of observed delays to wehibles in lane 1 suggests the likelihood of considerable correlation between delays of successive vehicles. It would be interesting, therefore, to match the serial correlation coefficients of the model against those of the data, possibly by means of a suitable kind of harmonic analysis. Unfortunately when the attempt is made to evaluate the serial correlation coefficient of lag one in the model, difficulties are encountered which are of such a substantial nature that it would appear that the evaluation of serial correlation coefficients of lags greater than one, by similar methods at least, is not feasible.

We are thus reduced to consideration of the single lag 1 correlation coefficient, and the derivation of this is discussed in some detail in section 5.4.4. This coefficient was not evaluated numerically for the following reasons.

- (i) Quantities such as $\tilde{S}_{2}'(1)$, $\tilde{S}_{2}''(1)$ are required. In the evaluation of the mean and variance of the delay distribution only $\bar{S}_{2}''(1) + \tilde{T}_{2}''(1)$ and $\tilde{S}_{2}''(1) + \tilde{T}_{2}''(1)$ were required and the algebra and computation were arranged in order to take advantage of any simplifications which might ensue from considering the sum $\tilde{V}_{2}(x) = \tilde{S}_{2}(x) + \tilde{T}_{2}(x)$. The evaluation of $\tilde{S}_{2}(x)$, though certainly possible, would necessitate repetition of much of this work.
- (ii) The comparison of an observed with a predicted correlation coefficient would, it appears, be merely on an inspection basis in the lack of a suitable test of significance.

Despite these difficulties, it is the case that, as with the sampling variance of the delay distribution, the sample serial correlation coefficient of lag one should not be unduly influenced by the existence of an approximately constant bias in the observation on delay. It would appear to be possible to expand Jowett's work slightly to obtain an approximately unbiased estimate of the lag 1 serial correlation coefficient by a suitable averaging of the quantity

$$\sum_{i,j:} \frac{1}{2} (x_i - x_j) (x_{i+1} - x_{j+1})$$

i, j:
i-j |> s_0

since $\mathcal{X}_{s} = \mathbb{E}\left\{\frac{1}{2}(\mathbf{x}_{i} - \mathbf{x}_{i+s})(\mathbf{x}_{i+1} - \mathbf{x}_{i+s+1})\right\} = \sigma^{2}(\rho_{1} - \frac{\rho_{s-1} + \rho_{s+1}}{2})$. so would be chosen so that $|\mathcal{X}_{s} - \sigma^{2}\rho_{1}| \leq \varepsilon$ for $s \gg s_{0}$. The previous paragraphs have been concerned with attempts to assess how well the model predicts fairly general aspects of the distribution of delays in a given lane. However the question of proportionally how many vehicles pass through the intersection without delay is of considerable practical interest, and is another aspect of the data which might be compared with the model. Since only the first two moments of the marginal delay distributions have been considered, information derived from this comparison may be largely independent of previous results. The theoretical probabilities of no delay under the assumption of model III are determined in section 4.3.5, and calculated values listed in Tables 5.2 and 5.3.

In order to construct a sample value for the probability of no delay in lane 1, we associate with the n^{th} successive vehicle crossing the intersection after the commencement of the period of observation a dummy random variable z_i defined by

 $z_{i} = 1$ if this vehicle is not delayed,

= 0 if it is delayed.

The probability of no delay is estimated by \overline{z} for a large sample, and once again we may invoke the work of Jowett to provide an estimated standard error for this mean, since the z_i form a correlated sequence of observations. Table 5.5 gives for the estimated atandard error of this probability a value of about .06 (with $s_0 = 10$).

From tables 5.2 and 5.3, we may therefore regard the entries for

'probability of no delay' in each of rows (c), (3) and (5), which are each within one estimated standard deviation of the sample value, as indicating consistency of the model with the data.

| Analysis of | Semi-squared | differences | of zi. |
|----------------|--------------|-------------|--------|
| ⁸ 0 | dso | | J 2 ** |
| 0 | 0.0000 | | 0,2410 |
| 1 | 0.1080 | | 0.2424 |
| 2 | 0.1615 | | 0.2441 |
| 3 | 0.2063 | | 0.2451 |
| 4 | 0.2201 | | 0.2456 |
| 5 | 0.2405 | | 0,2460 |
| 6 | 0.2293 | | 0.2460 |
| 7 | 0.2468 | | 0.2462 |
| 8 | 0.2645 | | 0.2462 |
| 9 | 0.2792 | | 0.2460 |
| 10 | 0.2810 | | 0.2456 |
| 11 | 0.2632 | | 0.2451 |
| 12 | 0.2450 | | 0.2449 |

Table 5.5.

1 3

In the preceding paragraphs we have attempted to compare certain significant aspects of the observations with the model. In spite of difficulties in finding suitable estimates of the parameters specifying the model, and although the interpretation given to certain aspects of the model in the actual situation is probably not the most suitable, it has been possible to demonstrate a considerable degree of conformity of predicted with observed behaviour. There is every likelihood that a more direct method of estimation of parameters in the model, possibly based on a minimisation of a measure of distance of observed delay frequency distributions from their theoretical counterparts, might lead to an even closer fit of the model to the data.

5.4.3. A comparison of model III with model I.

A likelihood function similar to that constructed for model III was programmed for model I, and a sequence of regeneration points of the type defined in model I was extracted from the film. In order to make this sequence of points compatible with the assumptions of model I a slight adjustment of the data was required. The likelihood function was maximized with respect to the parameters r_i , i = 1, 2. The maximized log-likelihood for model I was found to be -312.7. The comparable figure for model III is -174.8.

If these values were regarded as calculated for the entire record of the system which is available, then (Bartlett 1967) they might be

considered to indicate the superiority of model III to model I. However the likelihoods refer to observed sets of regeneration points, which are not identical for both models, and thus the likelihoods are not directly comparable.

We may obviate this difficulty by observing that the set of regeneration points for model I (set I) is included in the set for model III (set 3), and thus if we consider set 1 under the assumptions of model III the value of the likelihood obtained in this way will be greater than or equal to that calculated for set 3 under model III.

It would appear that model III represents a substantial improvement on model I.

5.4.4. The correlation coefficient of successive delays to vehicles in lane 2 of model III.

Let A and B refer to consecutive vehicles in lane 2, selected at random. The vehicles are delayed for intervals of length t, t', and leave queues of lengths q, q', respectively. The joint distribution function of t, t' is $B_2(t,t')$: $g_2(q,q')$ is the probability of a particular pair of values q, q'.

Suppose first that q is non-sero, i.e. B arrives before A departs, say an interval τ after the arrival of A. Then in the interval t- τ immediately following the arrival of A, q-1 vehicles arrive in lane 2 and in the succeeding interval t'-t+ τ a further q'-q+1 vehicles arrive in lane 2.

Thus

$$g(q, q^{:}) = \int_{0}^{\infty} \int_{0}^{\infty} \int_{z=0}^{t} \lambda_{2}^{e^{-\lambda_{2}e^{-\lambda_{2}e^{-\lambda_{2}(t-\tau)}q-1}} e^{-\lambda_{2}(t-\tau)} \frac{\lambda_{2}(t^{i}-t+\tau)}{(q^{i}-q+1)!} q^{i}-q+1}{(q^{i}-q+1)!} e^{-\lambda_{2}(t^{i}-t+\tau)} \frac{\lambda_{2}(t^{i}-t+\tau)}{(q^{i}-q+1)!} e^{-\lambda_{2}(t^{i}-t+\tau)} d\tau dB_{2}(t,t^{i}), q^{i} \ge q-1, q\neq 0.$$

Similarly

$$g(0,q^{\dagger}) = \int_0^{\infty} \int_0^{\infty} e^{-\lambda_2 t} e^{-\lambda_2 t^{\dagger}} \frac{(\lambda_2 t^{\dagger})^{q^{\dagger}}}{q^{\dagger} t} dB_2(t,t^{\dagger})$$

Let us define $G_2(x,x') = \sum_{q=1}^{\infty} \sum_{q'=q-1}^{\infty} g_2(q,q') x^q x^{q'} + \sum_{q'=0}^{\infty} g_2(0,q') x^{q'},$

and
$$B_2^{\mathbb{Z}}(s,s') = \int_0^{\infty} \int_0^{\infty} e^{-(st+s't')} dB_2(t,t').$$

It follows that

$$G_{2}(\mathbf{x}, \mathbf{x}^{\circ}) = \lambda_{2} \mathbf{x} \int_{0}^{\infty} \int_{0}^{\infty} \int_{0}^{t} e^{-\lambda_{2}(t^{\circ}+t^{\circ}) \lambda_{2} \mathbf{x}^{\circ}(t-t^{\circ}) \lambda_{2} \mathbf{x}^{\circ}(t^{\circ}-t+t^{\circ})} d\mathbf{x} d\mathbf{B}_{2}(t, t^{\circ})} + \int_{0}^{\infty} \int_{0}^{\infty} e^{-\lambda_{2}(t^{\circ}+t-t^{\circ}\mathbf{x}^{\circ})} d\mathbf{B}_{2}(t, t^{\circ})} d\mathbf{B}_{2}(t, t^{\circ})$$

$$= \frac{\mathbf{x}}{(1+\mathbf{x}\mathbf{x}^{\circ}-\mathbf{x}^{\circ})} \left[\mathbf{B}_{2}^{\mathbf{z}}(\lambda_{2}\mathbf{x}^{\circ}(1-\mathbf{x}), \lambda_{2}(1-\mathbf{x}^{\circ})) - \mathbf{B}_{2}^{\mathbf{z}}(\lambda_{2}, \lambda_{2}(1-\mathbf{x}^{\circ})) \right] + \mathbf{B}_{2}^{\mathbf{z}}(\lambda_{2}, \lambda_{2}(1-\mathbf{x}^{\circ}))$$

This expression reduces immediately to the expression of 84.3.4 for the generating function of the marginal distributions of q, q° . If we write $y = \lambda_2 x^{\circ}(1-x)$, $y^{\circ} = \lambda_2(1-x^{\circ})$, and define further $K_2(y, y^{\circ}) \propto (\lambda_2 - y)(\lambda_2 - y^{\circ})G_2(1 - \frac{y}{\lambda_2 - y^{\circ}}, 1 - \frac{y^{\circ}}{\lambda_2})$, we have, when $y = y^{\circ} = 0$,

$$\frac{\partial}{\partial y \partial y^{*}} \mathbb{E}_{2} = \lambda_{2}^{2} \frac{\partial}{\partial y \partial y^{*}} \mathbb{B}_{2}^{\Xi}(y, y^{*}) - \lambda_{2} \frac{\partial}{\partial y} \mathbb{B}_{2}^{\Xi}(y, y^{*}) - \lambda_{2} \frac{\partial}{\partial y^{*}} \mathbb{B}_{2}^{\Xi}(y, y^{*}) + \mathbb{B}_{2}^{\Xi}(\lambda_{2}, y^{*}) = \lambda_{2} \frac{\partial}{\partial y^{*}} \mathbb{B}_{2}^{\Xi}(y, y^{*})$$

From this relation $E\{tt^{i}\}$ may be recovered in terms of $\frac{\partial^{2}}{\partial y \partial y^{i}} E_{2}(y, y^{i})$ and known quantities. (Since e.g.

$$\frac{\partial}{\partial y^{\dagger}} B_2^{\mathbf{x}}(\mathbf{y}, \mathbf{y}^{\dagger}) \Big|_{\mathbf{y}=\mathbf{y}^{\dagger}=0} = \mathbf{E}(\mathbf{t}^{\dagger}) \Big) .$$

We proceed to describe how $G_2(x, x^{\circ})$ may be determined. Vehicle A departs either at or immediately following a regeneration point which is called the "preceding regeneration point". Only the component of G_2 which arises from a preceding regeneration point of type $S_2(n)$ is evaluated: similar methods may be used for the other components of $G_{2^{\circ}}$. It is convenient to denote by k the <u>total</u> number of vehicles which leave at the $S_2(n)$ preceding the departure of A. H₂ is the normalising constant defined in §4.3.4.

$$= \frac{\frac{1}{H_2}}{\frac{1}{H_2}} P_2^{k-1} q_2 \cdot \frac{1}{k} \epsilon (n-q) \epsilon (q-n+k-2), 2 \leq k \leq n$$

= $\frac{\frac{1}{H_2}}{\frac{1}{H_2}} P_2^{n} \epsilon (n-q) \epsilon (q-1), \qquad k = n + 1.$

Case (ii): A is the last vehicle to leave at its proceeding regeneration point, and the next vehicle to depart in either lane is B.

B may now have as preceding regeneration point, with nonsero probability, either $S_2(n)$ (for certain m), or $R_2(-1)$. We have, e.g., Pr. {case (ii), 2nd regeneration point is of type $S_2(n)$, q = n + 1 - k, $q^{\circ} = m$ } = $\frac{\sigma_2(n)}{H_2} p_2^{k-1} q_2 \frac{g(n-n+k)}{(n-n+k)!} \int_0^{\infty} e^{-\lambda \alpha} (\lambda_2 \alpha)^{n-n+k} g_2(\alpha) d\alpha$, $1 \le k \le n$; = $\frac{\sigma_2(n)}{H_2} p_2^n q_2 \frac{\lambda_2^{m+1}}{\lambda m!} \int_0^{\infty} f_2(\alpha) e^{-\lambda \alpha} a^m d\alpha$, k = n + 1.

For the remaining cases((iii) - (v)) it is convenient to make use of probability generating functions from the outset. The following p.g.f's are useful and refer to the distribution of the number of arrivals in lane 2 in an interval defined by the initial and final conditions detailed. (I.C. and F.C.). In each situation I.C. refers to a regeneration point, with the exceptions of (b), (c), (3).

Situation (a).

I.C. : S₂(n) at which at least one vehicle does not depart. Gap to I in 1 rejected.

F.C. : Departure of this I in 1.

Pr. { r arrivals in lane 2 during interval of length t }

$$\int_{0}^{\infty} \lambda_{1} e^{-\lambda_{1}t} (1-C_{2}(t)) \frac{(\lambda_{2}t)^{r}}{r!} e^{-\lambda_{2}t} dt,$$

$$p_{\alpha}(z) = \lambda_{1} (1-S_{2}^{z}(\lambda-\lambda_{2}z))/(\lambda-\lambda_{2}z).$$

Situation (8).

I.C. : Quone exists in lane 2. Next vehicle in ? (of unspecified type) has gap rejected.

F.C. : Departure of this vehicle in 1.

$$p_{\beta}(s) = \lambda_1(1 - p_1 g_2^{\mathbf{x}}(\lambda - \lambda_2 s) - q_1 h_2^{\mathbf{x}}(\lambda - \lambda_2 s))/(\lambda - \lambda_2 s),$$

Situation (8).

I.C. : Queue in 2. Gap to next vehicle in lane 1 is accepted.

F.C. : Departure of leading vehicle in lane 2.

$$p_{\chi}(z) = p_{1}g_{2}^{\Xi}(\lambda-\lambda_{2}z) + q_{1}h_{2}^{\Xi}(\lambda-\lambda_{2}z),$$

Situation (8).

I.C. : No queues: arrival in lane 2. Nearest vehicle in 1 is a I, and gap to this vehicle is rejected.

F.C. : Departure of this I in 1.

(Omit possibility that first lane 2 arrival is a I: already covered in Case (ii) above.)

$$p_{b}(s) = \lambda_{1}\lambda_{2}q_{2}s(1 - f_{2}^{R}(\lambda-\lambda_{2}s))/\lambda(\lambda-\lambda_{2}s).$$

Situation (R).

I.C. : No queues: arrival in lane 2 of type II. Gap to next arrival in 1 (of type I) is rejected.

F.C. : Departure of this I in 1.

$$p_{\alpha}(z) = \lambda_{1} z (1 - f_{2}^{\alpha}(\lambda - \lambda_{2} z)) / (\lambda - \lambda_{2} z).$$

Situation (3).

I.C. : No queues: arrival in lane 2 of type II. Gap to nort arrival in 1 (of type I) is accepted.

F.C. : Departure of this II in 2.

 $p_{\chi}(s) = sf_2^{H}(\lambda-\lambda_2s).$

Situation (7).

I.C. : No queues in lane 2: arrival in lane 2 of type I.

F.C. : Departure of this I in 2.

 $\mathbf{p}_{c}(\mathbf{z}) = \mathbf{z}.$

Situation (0).

I.C. : No queues: arrival in lane 2 of type II. Next in 7 is of type II.

F.C. : Departure of this II in 2.

p_A(z) = z.

We now return to consideration of possible developments of the system from an $S_2(n)$.

Case (iii): A is last vehicle to leave at an S₂(n), at which not all queueing vehicles depart. The next vehicle to cross the intersection is in lane 1.

This corresponds to situation (a) followed by possibly several situations (β), terminating with a situation (2).

If we define

 $r_{\alpha} = no.$ of arrivals in lane 2 during situation (α) $r_{\beta_1} = no.$ of arrivals in lane 2 during ith situation (β) $r_{\chi} = no.$ of arrivals in lane 2 during situation (χ),

we have

$$q^{1} = n+1-k+r_{a}+r_{g}+\sum_{i=1}^{p} -1$$

where p = 0, 1, ... is the number of situations (β) which occur. Since r_{α} , r_{β} , r_{β} are independent,

$$p \cdot g \cdot f \cdot (q \cdot |q, S_2)(x^{\dagger}) = x^{\dagger} (q - 1) \frac{\lambda_1 (1 - g_2^{\Xi}(\lambda - \lambda_2 x^{\dagger}))}{(\lambda - \lambda_2 x^{\dagger})} \cdot \frac{p_1 g_2^{\Xi}(\lambda - \lambda_2 x^{\dagger}) + q_1 h_2^{\Xi}(\lambda - \lambda_2 x^{\dagger})}{\lambda_2 (1 - x^{\dagger}) h_1 (1 - g_2^{\Xi}(\lambda - \lambda_2 x^{\dagger}))}$$

and so the contribution to $G_2(\mathbf{x}, \mathbf{x}')$ is $\frac{\lambda_1 q_2 \mathbf{x}}{H_2(\mathbf{p}_2 - \mathbf{x}\mathbf{x}')} \left\{ \tilde{S}_2(\mathbf{p}_2) - \tilde{S}_2(\mathbf{x}\mathbf{x}') \right\} \frac{(1 - g_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}') \left\{ \mathbf{p}_1 g_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}') + q_1 h_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}') \right\}}{\left\{ \lambda_2(1 - \mathbf{x}') + \lambda_1(\mathbf{p}_1 g_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}') + q_1 h_2^{\mathbf{x}}(\lambda - \lambda_2 \mathbf{x}')) \right\}}$

Case (iv): A is the last vehicle to leave at an S₂(n) at which all queueing vehicles depart. The next vehicle to arrive in either lane is B, which is delayed for a non-zero interval of time.

This corresponds to situation (5) followed by possibly several situations (β) and situation (χ).

Case (v): A is the last vehicle to leave at an S₂(n) at which all queueing vehicles depart. The next vehicle to arrive is in lane 1 (and of type I). There may be several further arrivals in lane 1 prior to the next arrival in lane 2 (the arrival of B). Immediately following the arrival B one of situations (ε), (ζ), (η) or (θ) obtains. Situation (ε) would be followed by possibly several situations (β) and a situation (χ).

The contributions to $G_2(x, x')$ arising from cases (iv) and (v) total to

$$\frac{\lambda_{1}}{\lambda} = \frac{\tilde{s}_{2}(p_{2})}{H_{2}} \left\{ \frac{q_{2}(1-f_{2}^{\#}(\lambda-\lambda_{2}x^{*}))(\lambda_{2}+\lambda_{1}p_{1})(p_{1}s_{2}^{\#}(\lambda-\lambda_{2}x^{*})+q_{1}h_{2}^{\#}(\lambda-\lambda_{2}x^{*}))}{\left[\lambda_{2}(1-x^{*})+\lambda_{1}(p_{1}s_{2}^{\#}(\lambda-\lambda_{2}x^{*})+q_{1}h_{2}^{\#}(\lambda-\lambda_{2}x^{*}))\right]} + p_{1}q_{2}f_{2}^{\#}(\lambda-\lambda_{2}x^{*})+p_{2}+q_{2}q_{1}\right\}$$

The total contribution to $G_2(x, x^*)$ arising from a preceding regeneration point of type $S_2(n)$ consists of the sum of the contributions arising from cases (iii) to (v) (listed above) and the contributions from cases (i) and (ii), which are :

$$\frac{1}{H_2} \left[\frac{x p_2}{p_2 - x x'} \left(\tilde{s}_2(p_2) - \tilde{s}_2(x x') \right) + \frac{x q_2}{p_2} g_2^{\#}(\lambda - \lambda_2 x') \left\{ \frac{p_2 \tilde{s}_2(x x') - x x' \tilde{s}_2(p_2)}{x x' - p_2} + \tilde{s}_2(p_2) \right\} \\ + \tilde{s}_2(p_2) \frac{\lambda_2}{\lambda} \left(q_2 f_2^{\#}(\lambda - \lambda_2 x') + p_2 \right) \right] .$$

The components of $G_2(x, x^*)$ arising from preceding regeneration points of types $T_2(n)$, $R_2(n)$, $R_2(-1)$, $R_2(-2)$ may be discussed in a similar way. For the reasons given in §5.4.2, this calculation is not completed here.

As a check on this complicated analysis, one might use the wellknown properties of a bivariate probability generating function. On inspection of the component of G_2 above, after multiplication by H_2 , it is found that

- a) with x = x' = 1 this reduces exactly to those terms in H₂ which may be regarded as arising from S₂(n).
- b) with x' = 1, this quantity reduces to the corresponding part of the marginal p.g.f. of q. evaluated in §4.3.4.

Thus two partial checks on the calculations are available. When the attempt is made to obtain the relevant part of the marginal p.g.f. of x', the expression obtained is not immediately identifiable as in cases (a), (b). It seems likely, however, that, with appropriate use of the basic equations of the system, the entire function $G_2(x, x')$ would satisfy this further condition.

Chapter 6.

Conclusion

6.1. Discussion of chapters 2-5: problems outstanding and conclusions.

Chapters 2-5 appear to demonstrate that it is at least possible to construct plausible mathematical models for the problem of delay due to right-turning vehicles, and to make some progress with their analysis.

Further analysis of these models and in particular of model III might be directed towards answering the questions:

- (i) what is the distribution of delay caused to vehicles arriving in the minor road?
- (ii) is a time-dependent analysis of these models possible?
- (iii) could model III be supplemented by a model for pedestrian behaviour at the intersection?

It might for instance be possible to utilise the formulae of Tanner (1962) for the mean delay to vehicles in a minor road caused by major road traffic, if it were possible to derive the first two moments of the distribution of the 'busy periods' of the intersection.

With regard to the results of fitting model III to an actual intersection, a claim could be made that the model reflects in a meaningful way those aspects of the real situation arising from rightturning vehicles. Given sufficient resources, one could envisage a much more comprehensive statistical analysis of the model. Data from several intersections might be available, possibly consisting of sequences of observations at different times of day reflecting a range of parameter values in the model, and in particular for situations in which the proportion of type II vehicles in each lane is substantial. These data would be obtained on film with a frame speed sufficiently high to give the detail required for an empirical study of the gap-acceptance mechanisms in use. It would be interesting to select firstly intersections to which the assumptions of the model (particularly concerning randomness of arrivals) might provide a good approximation, and secondly, a group of intersections to which these assumptions would apply to a progressively lesser extent. In this way information as to the range of validity of the model might be obtained.

At the same time a variety of distributional forms for gaprequirement times could be considered, and the most suitable determined. It seems that the likelihood function used in chapter 5 is not adequate and that modifications to this based on a more complete description of the system should be made. For instance, either the number of vehicles departing at a regeneration point, or the duration of the transition interval might be used to augment the description used above.

6.2. Practical applications of Model III.

It is possible to envisage certain applications of model III, although the construction and the analysis of the models of this paper is intended to be not so much a discussion of practical problems as a demonstration of the feasibility of this type of approach to problems arising from actual road layouts.

An obvious application is to the design of a road system. The model could be used to give an indication of how much traffic a particular intersection could be expected to deal with, assumptions about gap-acceptance being based on experience with similar intersections elsewhere. The question of the optimal loading of an intersection would become important if some form of traffic routeing through a network were available, and the model might give information on this.

We note that it is frequently the case that the intersection of a four lane major road with a minor road may be regarded as the intersection of a two lane major road with a minor road, because of the reduction in useful width of the major road caused by parked vehicles. Model III or III(a) might be applicable to such situations.

At a particular intersection for which suitable data are to hand, model III could give information concerning (i) the choice of gaprequirement times by drivers, (ii) the capacity(corresponding to extreme values of parameters) of the intersection. If such data were available for a large number of intersections for which, in addition, accident statistics were available, it might be possible to relate either some aspect of the vector of parameters describing traffic at a particular intersection, or some measure of the loading of the intersection relative to its capacity, to aspects of the accident record of the intersection. Possibly a multivariate technique such as canonical correlations might be useful in this respect.

6.3. Further Problems.

A problem related to that considered in this thesis concerns the delays caused by right turning vehicles at the priority intersection of a four lane major road with a minor road.

It is possible to propose a set of rules of vehicular behaviour in a model of this situation which appear to make its analysis comparatively straightforward, by reducing the system to two independent subsystems. This may be achieved by the following assumptions (with the usual notation):

- Type I vehicles are confined to the outer lanes, and type II to the inner lanes, at the intersection.
- (ii) Type II vehicles base their decision to cross on a gap-requirement time applied only to the stream of opposing type I's, i.e. to the opposing outer lans.
- (iii) Vehicles arriving at the intersection are of randomly assorted type.

With this set of rules, each of the two subsystems consisting of type I vehicles together with the opposing type II vehicles, is independent of the other, at least so far as delays are concerned. The four lane intersection may be regarded as a combination of two independent 'T-junctions', models for which are discussed in Chapter 1. The model might be plausible for some situations, in particular when lane markings are present.

It is however a matter of common experience that turning traffic may be delayed by opposing queueing type I vehicles. It is not difficult to construct a model along similar lines which would incorporate this feature. It is not obvious in what way a system of regeneration points may be constructed to assist in its analysis. The difficulties are twofold: a) the possibility of queues simultaneously present in two lanes would presumably increase the number of types of regeneration point, with a coreasponding increase in the number of transitions to be considered; and b) it would appear necessary to incorporate into the description of the system at any instant when two queues existed a measure of the time due to elapse before the arrival of the next vehicle in a particular lene. Thus it appears that the analysis of such a system might be more appropriately based on that of Cohen and Stam (1963).

Appendix I.

Transition probabilities for model I.

| Transit | lon | Probability |
|----------------------|---------------------|--|
| R ₁ (-1), | R ₁ (-1) | $\lambda_1 \Sigma_1 / \lambda$ |
| R ₁ (-1), | R ₂ (-1) | $\lambda_2 p_1/\lambda$ |
| R ₁ (-1), | R ₁ (-2) | 0 |
| R ₁ (-1), | R ₂ (-2) | 2192/2 |
| R ₁ (-1), | R ₁ (n) | 0 |
| R ₁ (-1), | R ₂ (n) | $\frac{\lambda_2 q_1}{\lambda} \frac{(1-r_1) \left(\lambda_1\right)}{\lambda}^{n}$ |
| R (-2) | P (-1) | 0 |

| $R_{1}(-2),$ | R ₁ (-1) | 0 |
|----------------------|---------------------|---|
| R ₁ (-2), | R ₂ (-1) | 2291 2/2 |
| R ₁ (-2), | R ₁ (-2) | 2/2 |
| R ₁ (-2), | R ₂ (-2) | 2291/2 |
| R ₁ (-2), | $R_{j}(n)$ | $\frac{\lambda_1 p_1}{\lambda} \frac{(1-r_2) \left(\frac{\lambda_2}{\lambda}\right)^n}{\left(\frac{1-r_2}{\lambda}\right)^n}$ |
| $R_{1}(-2),$ | R ₂ (n) | 0 |
| | | |
- (ii)
- $\frac{\lambda_{1}p_{2}}{1} \left(p_{1}r_{2}r_{2}^{n-1} + q_{1}r_{1} \right)$ $R_{1}(n), R_{1}(-1)$ $R_1(n), R_2(-1)$ λ2p1r2f2^m/λ $\frac{\lambda_1 q_2}{\lambda} \left(q_1 + p_1 r_2 g_2^{n-1} \right)$ $R_{1}(n), R_{1}(-2)$ $R_{q}(n), R_{2}(-2)$ 2291/X $p_{1}\lambda_{1}(1-r_{2})\left\{\frac{1}{\lambda_{1}}\left(\frac{\lambda_{2}}{\lambda_{1}}\right)^{\text{Hom}} \in (\mathbf{R}-\mathbf{E})\right\}$ + $\frac{r_{2}q_{2}r_{2}}{\lambda_{1}+\lambda_{2}q_{2}(1-r_{2})}\left[1-\left(\frac{\lambda_{2}r_{2}}{\lambda_{1}}\right)^{\text{Hom}}\right] \in (\mathbf{R}-\mathbf{E}-1)$ $R_1(n), R_1(n)$ $+\left(\frac{\lambda_{2}}{\lambda}\right)^{n-m+1} \cdot \frac{r_{2}q_{2}}{\lambda_{1}+\lambda_{2}q_{2}(1-r_{2})} \left|1-\left(\frac{\lambda_{2}f_{2}}{\lambda}\right)^{m-1}\right|_{\mathcal{E}}(n-m)$ $+ \frac{\lambda_2^{n_q} 2^r 2^{\tilde{r}} 2^{m-1}}{\lambda^{m+1}}$ $\mathbf{R}_1(\mathbf{m}), \mathbf{R}_2(\mathbf{n})$ $\frac{\lambda_2 \mathbf{p}_2 \mathbf{q}_1}{\lambda} \begin{pmatrix} 1-\mathbf{r}_1 \end{pmatrix} \begin{pmatrix} \lambda_1 \\ \lambda \end{pmatrix}^{\mathbf{B}}$

| Transiti | on probabilities fo | r model II. | |
|----------------------|---------------------|---|---|
| $R_{1}(-1),$ | R ₁ (-1) | $\frac{\lambda_{1}}{2} \left(p_{1} + q_{1} f_{1}^{E}(\lambda) \right)$ |) |
| $R_{1}(-1),$ | R ₂ (-1) | x x2p1/x | |
| R ₁ (-1), | R ₁ (-2) | 0 | |
| $R_{q}(-1)_{p}$ | R ₂ (-2) | ×299/X | |
| $R_{1}(-1),$ | $R_q(n)$ | 0 | |

(i1i)

| $R_{1}(-1),$ | R ₂ (n) | $\frac{\lambda_2 \lambda_1^{n_q}}{\lambda(n-1)!} \int_0^\infty (1-F_1(\mathcal{I})) e^{-\lambda \mathcal{I}} \mathcal{I}^{n-1} d\mathcal{I}$ |
|----------------------|---------------------|--|
| R ₁ (-1). | S ₁ (n) | $\frac{\lambda_1^{n+1}q_1}{\lambda \cdot n!} \int_0^{\infty} f_1(\sigma) e^{-\lambda \sigma} \sigma^n d\sigma$ |
| R ₁ (-1), | s ₂ (n) | 0 |
| | | |
| R ₁ (-2) | R ₁ (-1) | 0 |
| $R_{4}(-2)$. | $R_{o}(-1)$ | $\lambda_{0} p_{s} f_{0}^{*} (\lambda) / \lambda$ |

| $R_{1}(-2),$ | R ₂ (-1) | $\lambda_2 p_1 f_2^{\mathbf{x}} (\lambda) / \lambda$ |
|----------------------|---------------------|--|
| $R_{1}(-2),$ | $R_{1}(-2)$ | λ_1/λ |
| $R_{1}(-2),$ | R ₂ (-2) | $\lambda_2 q_1 / \lambda$ |
| R ₁ (-2), | $R_1(n)$ | $\frac{\lambda_1 \lambda_2 p_1}{\lambda(n-1)!} \int_0^{\infty} (1-F_2(z)) e^{-\lambda z} z^{n-1} dz$ |
| R ₁ (-2), | $R_2(n)$ | 0 |
| $R_{1}(-2),$ | S ₁ (n) | 0 |
| R ₁ (-2), | S ₂ (n) | $\frac{\lambda_2^{n+1}p_1}{\lambda_n!}\int_0^\infty f_2(a) e^{-\lambda a} e^{n} da$ |

| R ₁ (n), | R ₁ (-1) | $\lambda_1 q_1 p_2 f_1^*(\lambda) / \lambda$ |
|---------------------|---------------------|--|
| R ₁ (n), | R ₂ (-1) | $p_1 f_2^{\pi}(\lambda) \in (1-m)$ |
| R ₁ (n), | R ₁ (-2) | λ19192/λ |
| R ₁ (n), | R ₂ (-2) | λ_{2q_1}/λ |
| R ₁ (m), | R ₁ (n) | $\frac{\lambda_1 \lambda_2^{n-m} p_1}{(n-m)!} \int_0^{\infty} e^{-\lambda t'} (1-F_2(\tau)) \tau^{n-m} d\tau \cdot \epsilon (n-m)$ |

(iv)

$$\begin{split} R_{1}(n), \ R_{2}(n) & \frac{\lambda_{2}\lambda_{1}^{n} P_{2} q_{1}}{\lambda(n-1)!} \int_{0}^{\infty} e^{-\lambda t'} (1-P_{1}(t))t^{n-1} dt \\ R_{1}(n), \ S_{1}(n) & \frac{\lambda_{1}^{n+1} q_{1} P_{2}}{\lambda n!} \int_{0}^{\infty} f_{1}(t) t^{n} e^{-\lambda t'} dt \\ R_{1}(n), \ S_{2}(n) & \frac{\lambda_{2}^{n-m+1} p_{1}}{(n-m+1)!} \int_{0}^{\infty} f_{2}(t)t^{n-m+1} e^{-\lambda t'} dt \\ S_{1}(n), \ R_{2}(-1) & \frac{\lambda_{1}}{\lambda} s_{1}^{n} (p_{1} + q_{1}f_{1}^{*}(\lambda)) \\ S_{1}(n), \ R_{2}(-1) & \lambda_{2} p_{1}s_{1}^{n}/\lambda \\ S_{1}(n), \ R_{1}(n) & 0 \\ S_{1}(n), \ R_{2}(n) & \frac{\lambda_{2} q_{1}}{\lambda} \int_{0}^{n-m} \frac{m-1}{t} (s_{1}\lambda_{1})^{t'} e^{(n-m+1)} \\ & \frac{\lambda_{1}^{n} s_{1}^{n}}{\lambda} \int_{0}^{\infty} (1-k_{1}) (\frac{\lambda_{1}}{\lambda})^{n-m} \int_{t=0}^{m-1} (s_{1}\lambda_{1})^{t'} e^{(n-m+1)} \\ & + \frac{\lambda_{1}^{n} s_{1}^{n}}{\lambda} \int_{0}^{\infty} (1-p_{1}(t)) e^{-\lambda t'} t^{n-1} dt \\ & + \frac{\lambda_{1}^{n} s_{1}^{m}}{\lambda n!} \int_{0}^{\infty} (1-p_{1}(t)) e^{-\lambda t'} t^{n-1} dt \\ & S_{1}(n), \ S_{1}(n) & \frac{\lambda_{1}^{n+1} q_{1} s_{1}^{m}}{\lambda n!} \int_{0}^{\infty} f_{1}(t) e^{-\lambda t'} t^{n-1} dt \\ & S_{1}(n), \ S_{2}(n) & 0 \\ \end{split}$$

Transition probabilities for model III.

| R ₁ (-1), | R ₁ (-1) | $\frac{\lambda_1}{\lambda} \left(p_1 + q_1 f_1^{\mathfrak{U}}(\lambda) \right)$ |
|----------------------|---------------------|--|
| $R_{1}(-1),$ | R ₂ (-1) | $\lambda_2 p_1 / \lambda$ |
| $R_{1}(-1),$ | R ₁ (-2) | 0 |
| R ₁ (-1), | R ₂ (-2) | $\lambda_{2}g_{1}/\lambda$ |
| $R_{1}(-1),$ | R ₁ (n) | 0 |
| R ₁ (-1), | R ₂ (n) | $\frac{\lambda_2^q t^{\lambda_1}}{\lambda(n-1)!} \int_0^{\infty} (1-F_1(\tau)) e^{-\lambda \tau} \tau^{n-1} d\tau$ |
| R ₁ (-1), | S ₁ (n) | $\frac{\lambda_1^{n+1}q_1}{\lambda n!} \int_0^{\infty} f_1(\tau) e^{-\lambda \tau} \tau^n d\tau$ |
| $R_{1}(-1),$ | s ₂ (n) | 0 |
| $R_{1}(-1),$ | T ₁ (n) | 0 |
| $R_{1}(-1),$ | $T_2(n)$ | 0 |
| | | |
| R ₁ (-2), | R ₁ (-1) | 0 |
| R ₁ (-2), | $R_{2}(-1)$ | $\lambda_2 p_1 f_2^{\pi}(\lambda)/\lambda$ |
| R ₁ (-2), | R ₁ (-2) | λ_1/λ |
| R ₁ (-2), | R ₂ (-2) | $\lambda_{2}q_{1}/\lambda$ |
| R ₁ (-2), | R ₁ (n) | $\frac{\lambda_1 \mathbf{P}_1 \lambda_2^{\mathbf{n}}}{\lambda(\mathbf{n}-1)!} \int_0^{\infty} (1-\mathbf{F}_2(\mathbf{Z})) e^{-\lambda \mathbf{Z}_1 \mathbf{n}-1} d\mathbf{Z}$ |

| R ₁ (-2), | R ₂ (n) | 0 |
|----------------------|---------------------|---|
| R ₁ (-2), | s ₁ (n) | 0 |
| R ₁ (-2), | s ₂ (n) | $\frac{\lambda_2^{n+1}p_1}{\lambda n!} \int_0^\infty f_2(\tau) e^{-\lambda \tau} r^n d\tau$ |
| R ₁ (-2), | T ₅ (n) | 0 |
| R ₁ (-2), | T ₂ (n) | 0 |
| | | |
| R ₁ (n), | R ₁ (-1) | 0 |
| R ₁ (n), | R ₂ (-1) | $p_1 g_2^{\mathbf{Z}}(\lambda) e(1-n)$ |
| $R_{\gamma}(n)$, | R ₁ (-2) | 0 |
| R ₁ (n), | R ₂ (-2) | $q_1 h_2^{\mathcal{F}}(\lambda) \in (1-n)$ |
| R ₁ (m), | R ₁ (n) | $\frac{\lambda_1 \lambda_2^{n-m}}{(n-m)!} \int_0^{\infty} \tau^{n-m} e^{-\lambda \tau} \left\{ 1 - p_1 G_2(\tau) - q_1 H_2(\tau) \right\} d\tau_e(n-m)$ |
| R ₁ (m), | R ₂ (n) | 0 |
| R ₁ (m), | S ₁ (n) | 0 |
| R ₁ (m), | S ₂ (n) | $\frac{p_1 \lambda_2^{n-m+1}}{(n-m+1)!} \int_0^\infty \tau^{n-m+1} e^{-\lambda \tau} g_2(\tau) d\tau \ e(n-m+1)$ |
| R ₁ (m), | T ₍₁) | 0 |
| R ₁ (m), | T ₂ (n) | $\frac{q_1 \lambda_2^{n-m+1}}{(n-m+1)!} \int_0^{\infty} x^{n-m+1} e^{-\lambda t} h_2(t) dt e(n-m+1)$ |

(vi)

(vii)

 $\frac{\lambda_1}{\lambda} p_1^n (p_1 + q_1 f_1^{\Xi}(\lambda)) + q_1 p_1^{n-1} g_1^{\Xi}(\lambda)$ S1(n), $R_{1}(-1)$ $\lambda_2 p_1^{n+1} / \lambda$ R₂(-1) S₁(n), R₁(-2) S1(n), 0 $\lambda_2 p_1^{n_q} q_1/\lambda$ $S_1(n), R_2(-2)$ S₁(m), R₁(n) $\lambda_{2^{q}} \left\{ \begin{array}{c} m-1 \\ \Sigma \\ r=0 \end{array}^{r} \int_{0}^{\infty} \left[1-G_{1}(z) \right] e^{-\lambda \tau} \frac{(\lambda_{1} z)^{n-m+r}}{(n-m+r)!} dz \right\}$ $S_1(m)$, $R_2(n)$ $e(n-m+r)+\frac{\lambda_1^{n}p_q}{\lambda(n-1)I}\int_0^{\infty}\left[1-F_1(\tau)\right]e^{-\lambda \tau}\tau^{n-1}d\tau$ $\frac{\lambda_1^{n+1}q_1p_1^m}{\lambda_n!}\int_0^{\infty} f_1(\gamma)e^{-\lambda\gamma}\gamma^n d\gamma + q_1\frac{m-1}{\sum}\frac{(n-m+r+1)}{(n-m+r+1)!}p_1^r$ S₁(m), S₁(n) $\int_{0}^{\infty} e^{-\lambda \alpha} (\lambda_{1} \tau)^{\alpha - m + r + 1} g_{1}(\tau) d\tau$ S₁(m), S2(n) 0 S₁(m), T₁(n) 0 S₁(m), T₂(n) 0

(viii)

 $T_1(n), R_1(-1)$ 0 $\lambda_2 p_1^{\pm 1} f_2^{\pm}(\lambda)/\lambda$ $T_1(n), R_2(-1)$ $p_1^{n-1}\left(\frac{\lambda_1 p_1}{\lambda} + q_1 h_1^{m}(\lambda)\right)$ $T_1(n), R_1(-2)$ $\lambda_2 p_1^{n_q} q_1 / \lambda$ $T_1(n), R_2(-2)$ $\frac{\lambda_1 \lambda_2 p_1^{m+1}}{\lambda(n-1)!} \int_0^\infty (1-F_2(\tau)) e^{-\lambda \tau} \tau^{n-1} d\tau$ T₁(m), R₁(n) $\lambda_{2}q_{1}\sum_{r=0}^{m-1}r\int_{0}^{\infty}\left[1-H_{1}(r)\right]e^{-\lambda \frac{r}{2}\left(\lambda_{1}r\right)^{n-m+r}}\frac{dr\cdot e(n-m+r)}{(n-m+r)!}dr\cdot e(n-m+r)$ T₁(m), R₂(n) $T_1(m)$, $S_1(n)$ $\frac{\lambda_2^{n+1}p_1^{m+1}}{\lambda_n!} \int_0^{\infty} e^{-\lambda \mathcal{I}} f_2(\tau) \tau^n d\tau$ T₁(m), S₂(n) $q_{1} \sum_{r=0}^{m-1} \frac{g(n-m+r+1)}{(n-m+r+1)!} p_{1}^{r} \int_{0}^{\infty} e^{-\lambda t} (\lambda_{1} \tau)^{n-m+r+1} b_{1}(\tau) d\tau$ T₁(m), T₁(n) T₁(m), T₂(n)

Appendix II. Practical Difficulties.

1. The Intersection.

Great difficulty was encountered in finding an intersection at which conditions were approximately those assumed in the model. The reasons for this were (i) there are not many intersections in the Glasgow area at which the major road contains only two lance, with literally no opportunity for vehicles to pass to the left of a queue, (ii) almost all intersections of interest were controlled, or were subject to interfering factors such as pedestrian crossings, (iii) the traffic intensity or the proportion of turning vehicles in at least one lane was usually far too small. In the event, only one suitable intersection was found, and even it was not entirely satisfactory - some vehicles did by-pass the queue, and the proportion of type II vehicles in one lane was unexpectedly small.

2. The observations.

Pictures were taken with a time-lapse cine camera at a constant interval of about one second for about 75 minutes at a peak period, resulting in approximately 4500 frames.

Upon inspection, it was observed that for intervals of considerable length the intersection was quite free of congestion. This was found to be a result of the occasional blocking of each of lanes 1,2 some miles upstream of the intersection. It was decided, therefore, to ignore these parts of the record, and considerable care was exercised in selecting for consideration intervals (beginning and ending with free flow) during which the flow of traffic to the intersection in both lanes appeared to be unobstructed. It was hoped, by this means, to bring the arrival processes into reasonable conformity with the postulated Poisson process, and this was in fact achieved.

3. Interpretation of the record.

(1) Queues.

The interpretation of a 'queue' in the model is not quite so well defined in the actual situation as might be supposed. Drivers tend to avoid periods of stationary queueing by slowing their approach to an intersection if congestion is observed ahead. The film shows clearly the brake-lights of vehicles in lane 1, and this information may be used when necessary to estimate the queue length at a given instant. Moving vehicles may be considered to be queueing - the criterion is not one of speed, but whether a vehicle would or would not have joined the end of a stationary queue had it continued to move with the speed at which it initially approached the intersection.

Some difficulty was experienced in distinguishing individual vehicles at the intersection during periods of heavy congestion, but it was usually possible to deduce queue lengths to within reasonable accuracy.

(ii) Delays.

Several interpretations of delay in the actual situation are

available. The one considered most meaningful and adopted here is as follows.

The delay experienced by a given vehicle is the difference between the times taken by this vehicle and by a vehicle of similar type, unimpeded by congestion and travelling initially at the same speed, to complete a journey through the intersection. This journey is from a point sufficiently far upstream to be clear of any congestion, to a point clear of the intersection in the exit lang.

The practical difficulties in estimating the delay of a webicle from the film are considerable. The number of frames taken by the vehicle to complete its journey was counted, and the nearest unobstructed vehicle was used to provide a comparison. It was found that the speeds of unobstructed vehicles through the intersection varied considerably.

Thus observations on delay are subject to several sources of error. Likelihood of bias arising from an improper interpretation of delay in the actual situation, the limitations imposed by frame speed, and the considerable subjective element in measuring delays, are some of the components contributing to discrepancies of observed delays from their predicted values.

(11i)

Appendix III. Observations.

(i) Inter-arrival intervals.

Measurements made to the nearest frame-separation (approximately 1 second). Commas define continuous periods of observation. (Data to be read horizontally.)

Lane 1.

| 9 | 3 | 4 | 5, | 8 | 7 | 1 | 8 | 6 | 8, | 2 | 6 | 39 | 1 | 2 | 6 |
|----|----|----|----|----|-----|----|-----|------|-----|-----|-----|----|----|----|---|
| 22 | 9 | 2 | 2 | 14 | 2 | 6 | 4 | 6 10 | D, | 2 1 | 8 | 36 | 4, | 2 | 3 |
| 16 | 5 | 1 | 4 | 1 | 5 | 2, | 2 | 2 | 1 | 2 | 4 | 2 | 15 | 2 | 2 |
| 3 | 1 | 5 | 4 | 1 | 1 | 2 | 1 | 3 | 7 | 1 3 | ; 1 | 2 | S | 3 | 3 |
| 3 | 2 | 1 | 3 | 1 | 2 | 1 | 9 | 6 | 10, | 12 | e. | Ą. | 3 | 4 | 7 |
| 1 | 10 | 16 | 6, | Ċ | 15, | 3 | 5 | 4 | 4, | 2 | 2 | 1 | 1 | 3 | 5 |
| 7 | 3 | 2 | 2, | 2 | 4 | 1 | 3 | 2 | 2 | 16, | 4 | 2 | Ą | 12 | |
| 7 | 6 | 2 | 2, | 3 | 3 | 4 | 5 | 5, | 2 | 1 | 3 | 1 | 2 | 4 | |
| 2 | 1 | 3 | 4 | 1 | 3 | 2 | 19, | 2 | 5 | 2 | 3 | 6 | 7 | 11 | 2 |
| 1 | 2 | 1, | 2 | 2 | 3 | 18 | 4 | 4 | 2 | 3, | 5 | 1 | 3 | 1 | |
| 12 | 1 | 1 | 8 | 2 | 1 | 1 | 4 | | | | | | | | |

Lane 2.

| 1 | 11 | 1 | 1 | 10 | 15 | 2 | 1 | 2, | | 1 | 1 | 1 | 2 | 9 | 1 |
|---|----|---|----|----|----|---|----|----|---|---|---|---|---|---|---|
| 2 | 3 | 3 | 18 | 2, | 1 | 2 | 3 | 6 | 1 | 1 | 2 | 2 | 2 | | |
| 6 | 25 | 1 | 1 | 1 | 4 | 1 | 19 | 2 | 4 | 3 | 2 | 3 | | | |
| 4 | 7 | 1 | 4 | 2 | 2, | 2 | 2 | Ą | 1 | 3 | 3 | Ą | 3 | | |

7, 2, 2, 5. 13, 6 12, 4 4 8 6, 2 3 2, 4 2 2 1, 1 1 2 3, 2 1 3 1 13, 9. 6 9.

(ii) Delays to vehicles.

I

i.

Lane 1.

0 0. 12 5 4, 0 0 0 22 0 0, 0. 0 0 0. 0. 0 0 0 6 5 0, 2, 0 7, 0 11, 8, 2 0, 17 4 0. 7, 7 2 2 0 0 0 5 2, 0. 2. 18 17 15 15 20 12 11

Lane 2. (omitting zero delays). 5 2, 11 7 8 8 2, 11 9 8 5 (ii)

Appendix IV. References.

BARTLETT, M.S. (1967), Inference and stochastic processes, J.R.Statist.Soc.A, <u>130</u>, 457-474.

- COHEN, J., DEARNALEY, E.J., HANSEL, C.E.M., (1955), The risk taken in crossing a read, Operational Research Quarterly, <u>6</u>, 120-128.
- COHEN, J.W., DE LANGE, S.J., (1965), Numerical results for "Queueing for gaps in a traffic stream", Mathematical Institute, Technological University, Delft.
- COHEN, J.W., STAM, A.J., (1963), Queueing for gaps in a traffic stream, Mathematical Institute, Technological University, Delft.
- COX, D.R., LEWIS, P.A.W., (1966), The Statistical Analysis of Series of Events, Methuen, London.
- GAVER, D.P., Jr., (1965), Methods for describing time-dependent waits at traffic merges, Third International Symposium on the Theory of Traffic Flow, New York.
- HAIGHT, F.A., (1963), Mathematical Theories of Traffic Flow, Academic Press, New York.
- HAWKES, A.G., (1965), Queueing for gaps in traffic, Biometrika, <u>52</u>, 79-85.
- HAWKES, A.G., (1966), Delay at traffic intersections, J.R.Statist. Soc.B. 28, 202-212.
- HERMAN, R., WEISS, G.H., (1961), Comments on the highway crossing problem, Operations Research, 2, 828-840.
- JORDAN, C., (1950), Calculus of Finite Differences, Chelsea Publishing Co., New York.
- JOWETT, G.H., (1955), The comparison of means of sets of observations from sections of independent stochastic series, J.R.Statist.Soc.B, <u>17</u>, 208-227.

- KENDALL, D.G., (1951), Some problems in the theory of queues, J.R. Statist. Soc. B, 13, 151-173. KENDALL, D.G., (1953), Stochastic processes occurring in the theory of queues and their analysis by the method of the imbedded Markov Chain, Ann. Math. Stat. 24. 338-354. LEWIS, P.A.W., (1965), Some results on tests for Poisson processes, Biometriks, 52, 67-78. MAYNE, A.J., (1954), Some further results in the theory of pedestrians and road traffic, Biometrika, 41, 375-389. MILLER, A.J., (1962). Road traffic flow considered as a stochastic process, Proc. Camb. Phil. Soc., 58, 512-325. NEWELL, G.F., (1955). Mathematical models for freely-flowing highway traffic, Operations Research, 3, 176-186. NEWELL, G.F., (1959). The effect of left turns on the capacity of a traffic intersection, Quart. Appl. Math., 17, 67-76. OLIVER, R.M., BISBEE, E.F., Queueing for gaps in high flow traffic streams, Operations Research, 10, 105-115. REID, D.H., (1967), A mathematical model for delays caused by rightturning vehicles at an uncontrolled intersection, Journal of Applied Probability, 4, 180-191. TANNER, J.C., (1953). A problem of interference between two queues, Biometrika. 40. 58-69. TANNER, J.C., (1961). Dolays on a two-lane road, J.R. Statist. Soc., B. 23. 38-63. TANNER, J.C., (1962). A theoretical analysis of delays at an uncontrolled intersection, Biemstrika, 49, 163-170. WASSCHER, E.J., (1963), Algorithm 203: Steep 1, Communications of the Association for Computing Machinery, 6, 517-519. WEISS, G.H., (1965). A survey of some recent research in road traffic, Proc. of Symp. on Congestion Theory, University
- WEISS, G.H., MARADUDIN, A.A., (1962), Some problems in traffic delay, Operations Research, 10, 74-104.

of North Carolina Press, 255-280.